Mathematical Modeling and Analysis of Flexible Production Lines

by

Young Jae Jang

Submitted to the Department of Mechanical Engineering in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Mechanical Engineering

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2007

© Massachusetts Institute of Technology 2007. All rights reserved.

Author

Department of Mechanical Engineering February 8, 2007

Certified by..... Stanley B. Gershwin Senior Research Scientist, Department of Mechanical Engineering Thesis Supervisor

Accepted by Lallit Anand Chairman, Department Committee on Graduate Students

Mathematical Modeling and Analysis of Flexible Production Lines

by

Young Jae Jang

Submitted to the Department of Mechanical Engineering on February 8, 2007, in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Mechanical Engineering

Abstract

We present a model and analysis of a production line that processes different part types on unreliable machines which operate according to a priority rule. The production line consists of machines separated by storage areas in which parts flow in a fixed sequence. A machine operates on the highest priority part whenever possible, and only operates on lower priority parts when unable to produce the higher priority parts. Part priorities are static and are a function only of part type. The purpose of this thesis is to present mathematical formulations and algorithms for estimating production rates and average inventory levels for each part type in a flexible production line. The qualitative behavior of the multiple-part-type line under different supply and demand scenarios is described.

Thesis Supervisor: Stanley B. Gershwin Title: Senior Research Scientist, Department of Mechanical Engineering

Acknowledgments

I believe that I am one of fortunate people whose professional and personal lives have been benefited by outstanding people. I would like to express my gratitude to all those who gave me the possibility to complete this thesis. My colleagues in the Manufacturing Systems Engineering Community at MIT including Irvin Schick, Baris Tan, and Chiwon Kim, supported me in my research work.

I want to thank them all for their help, support, interest, and valuable hints. I want to thank Shinsung ENG, Ltd., Co. for providing me with valuable information and data about the current semiconductor and LCD industries. Mr. Jae Myung Yoo and Dr. Gi-Han Choi in Shinsung ENG shared their knowledge with me in the current issues in industry. Particularly, I am bound to the President Joo-Hun Kim in Shinsung ENG for supporting me throughout my graduate years in various ways.

Being a student of Dr. Stanly B. Gershwin, my research thesis advisor, is one of the luckiest things in my life. Stan taught, trained, and directed me toward the world of manufacturing systems. However, our relationship was not limited to the academic interest. We shared our view of nature, life, philosophy, and pretty much everything we could think of. I have been proud of being his student. I also would like to thank his wife, Fran Gershwin, for generously inviting me their summer house in New Hampshire every summer as well as letting be a welcomed guest in her house whenever I needed a temporary accommodation during my graduate years.

In shaping and molding my life and personality, there is no one who deserves more credit and thanks than my parents. Their love has sustained me and nourished me. They sacrificed a great deal for me, tolerated my shortcomings. My dad taught me how to be a strong person while my Mom showed me how to be a loving and sharing person.

I want to thank my wife, Yujin Lee. She has been more than my wife. Her love has been my inspiration. She has been patiently trusted me all the time. I could not have been completed my study successfully without her inspiration and love. Finally, thank you, God.

Contents

1	Introduction		11
	1.1	Motivation	12
	1.2	General Modeling Assumptions	13
	1.3	Literature Review	15
	1.4	Thesis Outline	15
2	Rev	view of Production Lines Processing a Single Part Type	17
	2.1	Single-Part Type Production Line	17
	2.2	Two-Machine One-Buffer Building Block	18
	2.3	Decomposition	20
3	Three-Part-Type Processing Line		
	3.1	Notation	22
	3.2	Assumptions and Approximations	23
	3.3	Part Type Priority Policy	26
	3.4	Production Rate	26
	3.5	Basic Idea of Decomposition	28
4	Two	o-Machine Lines with Idleness Failures	31
	4.1	Idleness Failure and Failure-Mode Change	31
		4.1.1 Idleness Failure	31
		4.1.2 Failure Mode Change	32
	4.2	Two-Machine-Line Notation and Parameters	32

	4.3	Transition Equations	35
		4.3.1 Transient states \ldots	35
		4.3.2 Non-transient states 3	35
		4.3.3 State Classification	36
	4.4	Performance Measures	37
		4.4.1 Efficiency	37
		4.4.2 Average buffer level, probability of blockage, and probability of	
		starvation	41
	4.5	Solution Algorithm	41
5	Dec	omposition Analysis for Type 1 4	14
	5.1	State and Parameter Definitions	44
	5.2	Equalities	46
	5.3	Resumption of Flow	49
	5.4	Failure Mode Change	52
	5.5	Interruption of Flow	54
	5.6	Idleness Failure of Type 1	57
	5.7	Boundary Conditions	58
	5.8	Summary of the Decomposition Equations for Type 1	59
		5.8.1 Resumption of flow	59
		5.8.2 Failure mode change	60
		5.8.3 Interruption of flow	61
		5.8.4 Idleness failure $\ldots \ldots \ldots$	61
6	Dec	omposition Analysis for Type 2 6	32
	6.1	State definitions	52
	6.2	Equalities	64
	6.3	Resumption of Flow	66
	6.4	Failure mode change	73
	6.5	Interruption of Flow	77
	6.6	Idleness failure	34

	6.7	Bound	lary Conditions	86
	6.8	Summ	hary of the Decomposition Equations for Type 2	86
		6.8.1	Resumption of flow	87
		6.8.2	Failure mode change	88
		6.8.3	Interruption of flow	89
		6.8.4	Idleness failure	90
7	Dec	compos	sition Analysis for Type 3	91
	7.1	State	definitions	91
	7.2	Equal	ities	92
	7.3	Resun	nption of flow	95
	7.4	Failur	e Mode Changes	98
	7.5	Interr	uption of Flow	100
	7.6	Idlene	ss Failure	104
	7.7	Summ	hary of the Decomposition Equations for Type 3	105
		7.7.1	Resumption of flow	105
		7.7.2	Failure mode changes	106
		7.7.3	Interruption of flow	107
		7.7.4	Idleness failure	108
8	Ger	neraliza	ation of Decomposition	109
	8.1	Idea o	f Generalization	109
	8.2	Gener	al Z Part Type Decomposition Equations $\ldots \ldots \ldots \ldots$	110
		8.2.1	Highest priority part type	110
		8.2.2	Intermediate priority part type	113
		8.2.3	Failure mode change	114
		8.2.4	Interruption of flow	115
		8.2.5	Idleness failure	116
		8.2.6	Lowest priority part type	117

9	Algorithm and Numerical Behavior		
	9.1	Algorithm	121
	9.2	Algorithm Behavior	125
	9.3	Randomly Generated Cases	125
	9.4	Computation of Approximation Error	126
	9.5	Numerical Results for Two-Part-Type Lines	127
	9.6	Numerical Results for Three-Part Type Lines	129
10 System Behavior 13			134
	10.1	Changing Demand	134
	10.2	Changing Processing Machine Capacity and Buffer Size	140
11 Extension: Re-entrant Line 14			144
	11.1	Re-entrant Production Line	144
	11.2	Decomposition Equations	145
	11.3	Numerical Results	148
		11.3.1 Case1: Varying p_4 and r_4 (e_4 constant)	148
		11.3.2 Case 2: Varying p_4 with constant $r_4 \ldots \ldots \ldots \ldots \ldots$	149
12	Con	clusion	154
\mathbf{A}	Two	-Machine One-Buffer Line Transition Equations	156
	A.1	Internal States	156
	A.2	Upper Boundary States	158
	A.3	Lower Boundary States	162

List of Figures

2.1.1 A production line processing a single-part type	17
2.2.1 Two-machine one-buffer building block	18
2.3.1 A decomposed production line processing a single-part type \ldots .	20
3.1.1 A two-part type production line	23
3.5.1 The decomposition of a line into building blocks $\ldots \ldots \ldots \ldots$	30
4.1.1 Example of Markov states with three failure modes	33
4.5.1 Example of a sparsity pattern $J = 3$ and $L = 3$	42
5.5.1 Markov Chain of $M^u(i, 1)$ and $M^d(i, 1)$	56
9.5.1 The errors in the decomposition approximation for Type 1 production	
rates	128
9.5.2 The errors in the decomposition approximation for Type 2 production	
rates	129
9.6.1 The errors in the decomposition approximation for Type 1 production	
	131
9.6.2 The errors in the decomposition approximation for Type 2 production	191
9.6.3 The errors in the decomposition approximation for Type 2 production	191
rates	132
10.1. Throughput vs Demand for Type 1	136
10.1.2 Throughput vs Demand for Type 2	138

10.1.3 Throughput vs Demand for Type 3	139
10.2. If hroughput vs Capacity of M_1	141
10.2.2 Throughput vs Size of	143
11.1. \mathbb{R} e-entrant production line model $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	144
11.3. Simple re-entrance production line model	148
11.3. \mathcal{P} roduction rate vs. p_4 (e_4 fixed)	150
11.3. Percent of Error vs. p_4	150
11.3.4 Average buffer level vs. p_4	151
11.3. Production rate vs. p_4 (r_4 fixed)	152
11.3. Percent of Error vs. p_4	153
11.3.7Average buffer level vs. p_4	153

List of Tables

9.5.1 Machine parameters for the randomly generated two-part type process-	
ing lines	128
9.5.2 Absolute percent errors for production rates and average buffer levels	
for the two-part type processing machines	130
9.6.1 Machine parameters for the randomly generated three-part type pro-	
cessing lines	130
9.6.2 Absolute percent errors for production rates and average buffer levels	
for the three-part type processing machines. \ldots \ldots \ldots \ldots \ldots	132
10.1 Machine and buffer parameters for Figure 10.1.1	135
	107
10.1.2 Machine and buffer parameters for Figure 10.1.2	137
10.1.3 Machine and buffer parameters for Figure 10.1.3	137
10.2. Machine and buffer parameters for Figure 10.2.1	140
10.2.2 Machine and buffer parameters for Figure 10.2.2	142
11.3. Failure and repair parameters. $(N_1 = N_2 = N_3 = N_4 = 20)$	149
11.3. Failure and repair parameters. $(N_1 = N_2 = N_3 = N_4 = 20)$	152

Chapter 1

Introduction

This thesis presents a model and analysis of a synchronous tandem production line that produces different part types on unreliable machines. Inventory is stored between machines in finite buffers. It is assumed that machines in the processing line are flexible in that they can operate on different part types, and there are no set-up penalties incurred when machines switch production from one part type to another. The machines operate according to a static priority rule, working on the highest priority part whenever possible, and on lower priority parts only when unable to process those with higher priorities due to either blockage or starvation. In this thesis, we investigate the flow behavior of a production system for a synchronous tandem flow line consisting of machines separated by storage areas in which parts flow in a fixed sequence. Such a tandem flow line can be observed in automotive fabrication, electronic components assembly and consumer products manufacturing industries.

The purpose of this research is to develop mathematical formulations and algorithms to analyze a manufacturing system processing more than one part type. That is, the goal of this thesis is to answer the question, "What average throughput rate and *work-in-progress* (WIP) for each part type can be expected from a given flexible production line design?" An accurate and logical answer to this question will help in understanding the best way to allocate buffer capacity in order to achieve the maximum throughput rate for each part type for a given process. Analyzing the system behavior and evaluating the performance of manufacturing systems is challenging due to the complexity and randomness of the systems. Systematic approaches to the problem include real world experimentation, simulation and analytical methods. Real-world experimentation is not feasible when analysis is required to design a new production system — a production line in which experiments need to be performed does not exist. Even if a production line is available for actual experimentation, doing so is often too expensive and too time-consuming. Simulation is widely used in industry to predict the behavior of production lines. It can be useful for the detailed evaluation of a single final design. However, simulation requires a significant amount of computation and modeling time to generate analytically meaningful results. Analytical models are by far the fastest method for evaluating system performance, but very hard to construct. The analytical approach is usually quicker, and cheaper than simulation, but subject to errors associated with the abstracting of reality. This thesis focuses on developing an analytical model to evaluate real production lines by accurately accounting for key system phenomena.

1.1 Motivation

Since the idea of *Flexible Manufacturing Systems* (FMS) — in which installed machines and operations provide the production line with the ability to build different kinds of products on the same production facility — was proposed in the 1950s, various industries have implemented it in actual production lines. In the automotive industry, for instance, with the development of the robot technology and advanced automated material handling systems, machines are able to switch operations from one product type to another flexibly. This system flexibility has been adopted as a way to produce several different types of products on the same production line. For instance, an article in the *New York Times*[11] describes the current trends of FMS in the automotive industry:

"Detroit automakers are striving to embrace this practice, which is a hallmark of Japanese manufacturers. It allows companies to quickly change the mix of vehicles that their plants make to reflect demand. In 2004, Ford said, only 38 percent of its plants were flexible. By 2008, the company expects that 82 percent of its plants will be flexible, an increase from its previous goal of 75 percent."

Producing a diversity of products in a single production line has become a common practice in high-tech industries, such as semiconductor memory chips and liquidcrystal-display (LCD) panel productions. For example, the author observed that an LCD panel production line typically produces several different sizes of panels. A flexible and rapid response to the customers needs of different product types have recently become the primary determinant of competitiveness. Considering the enormous capital investment of production lines (well in excess of 2 billion dollars), it is crucial to be able to understand and predict the overall behavior of the FMS production line. However, little analysis of flexible production has been done. Due to the lack of tools or methods, line designers in industry cannot consider the dynamic behavior of the production lines as well as methods to predict their systems behavior and estimate their performance are needed.

1.2 General Modeling Assumptions

A manufacturing system (also called transfer line, production line and within the queueing literature, tandem queueing system with finite buffers consists of parts (material), machines (work stages), and finite storage areas or buffers. In modeling the production lines in this research, we applied the concept of independence of events, which says that events in the future are only contingent on the present state of the system and are otherwise independent of each other and of past events [8]. In the production line model, this means that failure times and repair times on different machines are independent of one another. It also implies that the time between fail-

ures of a given machine are independent of previous failure times, repair times and the amount of material in the buffer, assuming the machine is not blocked or starved. This independence assumption allows the system to be modeled as a *Markov process*. Markovian models are frequently employed in analyzing production systems because of their tractability.

The cycle time is the time required for a single operation on an isolated machine. Cycle times are considered *fixed and deterministic* when they do not vary from one part to the next on a specific process. *Stochastic* cycle times vary randomly from part to part. Since production lines are usually designed to produce similar or identical mature products in large quantities, they usually perform their tasks with a low level of variability when they are operational. Therefore, in our model it is assumed that there is no variation in the processing time.

A machine failure occurs when a machine in the production line ceases to run due to a malfunction. There are two important categories of failures. Operation dependent failure, or ODFs, occur as a function of the amount of material processed since the last time a machine was repaired. Time dependent failures, or TDFs, occur as a function of the time a machine runs since its last failure [8]. There is a significant impact on flow line model estimate based on whether one assumes TDFs or ODFs. However, casual observation have shown that TDFs are not very prevalent. It is assumed that the machines in the proposed model operate with ODFs.

Quality issues, rework and yield, customer satisfaction, environmental impact, machine maintenance polices, and other management issues are not addressed in this thesis.

1.3 Literature Review

Buzacott [5] modeled a two-machine one-finite-buffer production system as a Markov process. The model assumed that machines are unreliable and subject to fail under operations dependent failures (ODFs). Also the model assumed that operations times of the two machines were identical and deterministic. Failure and repair times were geometrically distributed. As a justification for using the ODF assumption, Buzacott and Hanifin [4] provided data from Chrysler Corporation that showed that the majority of failures are ODFs. Production lines longer than two machines are not in general analytically tractable. However, that most factories have production lines with many machines. One method which has proved practically robust for production line with finite buffers and unreliable machines is *decomposition* which was introduced by Gershwin [7]. This model evaluates the throughput and average buffer levels of the line with tandem configurations. Later this model was extended to assembly/disassembly network by Gershwin [1]. Under the same assumption, a decomposition method for lines with a loop configuration was introduced by researchers [10]. However, all the models described so far were limited to a single part type. Nemec [12] formulated a deterministic single failure multi-part type transfer line processing two-part type. However, this formulation worked only for small two-part type lines, and there is no clear way of generalizing his equations for longer lines and more than two-part types. Tolio [13] proposed a way of analyzing two-part type lines with multiple failure modes. However, the Markov model of the two-machine line, the basic building block of the decomposition, for this model was complex. Moreover, in Tolio's model, the number of equations to solve increase exponentially as the production line becomes longer.

1.4 Thesis Outline

This thesis is organized as follows. Chapter 2 reviews the analysis of a single-part type production line using decomposition. Since the modeling and analysis of the multiple-part type production line follows the assumptions made for the single-part type line, it is critical to review this system first. Chapter 3 introduces a Markov model of a processing line with multiple-part types. The decomposition of the long line into smaller, tractable two-machine lines is also discussed. Chapter 4 presents the analysis of the Markov chain for two-machine lines. As a first step in modeling a production line with more than one part type, we restrict ourselves to the case where there are three part types. Type 1, Type 2, and Type 3 decompositions are introduced in Chapter 5, Chapter 6, and Chapter 7. Then in Chapter 8, we propose a way to expand the three-part-type line to general line processing more then three part types. An algorithm to solve the decomposition is presented in Chapter 9, as are numerical results concerning the accuracy of the decomposition. The system behavior of the three-part type processing line under different supply and demand scenarios is also investigated in Chapter 10. In Chapter 11, we present a decomposition for a re-entrant production line. In developing equations for the building blocks for the reentrant production line, we modify the decomposition model that has been created for the multiple-part-type line. Semiconductor device and liquid crystal display (LCD) fabrication processes are characterized as a re-entrant process, in which a similar sequence of processing step is repeated several times. Chapter 12 summarizes this work and proposes future research.

Chapter 2

Review of Production Lines Processing a Single Part Type

2.1 Single-Part Type Production Line

This section reviews the analysis of a production line processing a single part type particularly for deterministic processing model introduced by Gershwin [7]. The basic model of a production line consists of *machines* and *buffers*. Figure 2.1.1 shows a diagram of a production line processing a single part type. In the figure, machines and buffers are depicted by squares and circles, respectively. A machine, denoted by the letter M, is a processing unit where parts are processed. Between the machines, there are buffers, denoted by the letter B, where *in-process inventory* or *in-process* WIP are stored. In the model, it is assumed that the buffer space is limited. A part flows though the machines and buffers linearly in that it begins at the first machine and then is sent to the next downstream buffer immediately after the completion of the process in the first machine. Then the part is loaded to the second machine and so on. In the model, it is assumed that a part always enters to the line through the



Figure 2.1.1: A production line processing a single-part type

first machine and travels all the way down to the last machine. There is no re-entrant flow in the line and parts are never created no destroyed. Therefore, once a part enters into the machine M_1 , it moves on to B_1, M_2, B_2, M_3 and so on to M_K and then it exits the line.

We assume that the production times of each machine are deterministic and identical and this time is unity. However, machines are subject to failures and the failure quantities and repair quantities are not necessarily the same. The model also assumes that the machines have operation dependent failures and therefore, they can only fail while working on a part. It is further assumed that the machines have geometrically distributed up and down times. That is, in any given time step, each machine, M_i has a fixed probability p_i of failing if it is working on a part and has a fixed probability r_i of being repaired, if it is down.

The size of buffer B_i is denoted by N_i and the level of buffer B_i at time t is denoted by $n_i(t)$. We assume that the buffer is finite $0 \le n_i(t) \le N_i$ for all time t. Note that the buffer sizes are not required to be identical. In this configuration of the line, a machine can be idle due to *blockage* (i.e. $n_i = N_i$) or *starvation* (i.e. $n_i = 0$). Note that a machine not processing a part due to failure is different from machine being idle due to starvation or blockage. If a machine is starved or blocked, it is forced to stop processing although it is up and operational.

2.2 Two-Machine One-Buffer Building Block



Figure 2.2.1: Two-machine one-buffer building block

In the production line analysis, typical questions that need to be answered are "What is the throughput rate?" and "Which machine is the bottleneck?", and "How

much inventory is stored in the line?" Answering these questions is not trivial and providing the exact answers for the general case may not be possible. However, the exact analysis for a small system is possible. Let us consider a system consisting of two processing machines and one buffer as shown in Figure 2.2.1. In the line, a finite buffer of size N is placed between the machines. The first machine, denoted by M^{u} , is called the *upstream machine*, and the second machine, denoted by M^d , is called the *downstream machine*. All the assumptions described for the long production line also apply to this line. That is, the machines operate at the same rate and they can be down only when they are operating on a part. The machine M^u can fail with probability p^u while it is processing a part and it can be repaired with probability of r^u when it is down. Likewise, M^d has machine parameters of p^d and r^d . Let α^u denote the state of the machine M^u . Let us define α^u to be 1 if M^u is up and α^u to be 0 if M^u is down. Similarly, α^d represents the state of machine M^d . The size of the buffer is N and therefore, the level of buffer n satisfies $0 \le n \le N$. Then the state of the two-machine one-buffer line can be completely characterized as $s = (n, \alpha^u, \alpha^d)$. Let E^u be the probability that M^u makes a part in a given time step and E^d be the probability that M^d makes a part in a given time step. These quantities are determined by r^u, p^u, r^d, p^d , and N. Again, it is assumed that no part is created nor discarded in the middle of the line, that is, once a part enters M^u , the part always leaves from M^d . Therefore, $E^u = E^d = E$, where E is called the *production rate* or throughput rate of the line. That is, E is the probability that the line produces a part in any given time step. Since the operation time is 1, E is also the production rate or throughput rate.

This two-machine one-buffer system can be modeled as a discrete-time, discretestate Markov process and exact values of the production rate and average buffer level can be calculated. Also other interesting performance measures such as the probability of blockage of M^u , denoted by P_b , and the probability of starvation of M^d , denoted by P_s , can be evaluated.



Figure 2.3.1: A decomposed production line processing a single-part type

2.3 Decomposition

Unfortunately, production lines longer than two machines are not analytically tractable. Since the most factories have production lines with many machines, a method that can analyze a longer production line is needed. Developing a method that evaluates exact solutions for longer lines may not be possible, therefore a method that approximates the solution is useful. One method, which has proved be robust, accurately approximating performance measures for longer production lines with unreliable machines and finite buffers is *decomposition*. This method decomposes the long production line into tractable two-machine one-buffer production lines, called the *building blocks*. Each individual building block is analyzed and then the building blocks are patched back together in a way that the performance measures of the longer line are approximated by those of the building blocks.

The idea of the decomposition can be better explained with the decomposed line shown in Figure 2.3.1. Suppose that there is an observer in each buffer in the line. Each observer watches the flow of parts entering and leaving the buffer that she is in. Although the observer watches the flow of the actually line, she is informed and believes that the buffer she is watching is not part of the long production line, but it is the buffer of a small line consisting of two machines and one buffer.

For any buffer in the line, the inflow may stop for two reasons: the machine directly

upstream of the buffer may have failed, or some machine further upstream may have failed, starving the line downstream of it. Likewise, the outflow may stop for two reasons: the machine directly downstream of the buffer may have failed, or a machine further downstream may have failed, blocking the line upstream of it. However, the observer is unable to determine the reason that inflow or out flow has ceased. The parameters of the machines of the building block are found by the decomposition so that judging by the behavior of the inflow and outflow, the observer in the buffer is unable to tell whether of not she is in a buffer of the real line or building block. The decomposition is based on equations that relate the parameters of the building blocks to those of the neighboring building blocks and their performance. The derivations for the equations for the building blocks and detailed analysis of the decomposition method is described in Gershwin [8].

Chapter 3

Three-Part-Type Processing Line

3.1 Notation

Figure 3.1.1 represents a production line processing three different part types. The line consists of two kinds of components: machines M_i , denoted by the squares; and finite-capacity storage buffers $B_{i,j}$ for work in process inventory, denoted by the circles. Let us define K to be the number of machines that are processing two different part types in the line. At the beginning and end of the line, there are supply machines $M_{0,1}$, $M_{0,2}$, and $M_{0,3}$, and demand machines, $M_{K+1,1}$, $M_{K+1,2}$, and $M_{K+1,3}$.

Machines $M_{0,j}$ and $M_{K+1,j}$ process only Type j parts. Each machine, other than the supply and demand machines, processes all part types. We assume that there is no set-up time incurred when the machines switch production from one part type to another. When M_i completes work on a part, it sends the part to a buffer downstream of the machine. Each part type has a distinct buffer after each machine. For instance, a Type 1 part processed at M_i would be sent to $B_{i,1}$ and a Type 2 part processed at the same machine would be sent to $B_{i,2}$.



Figure 3.1.1: A two-part type production line

3.2 Assumptions and Approximations

It is assumed that all the machines in the line, including supply and demand machines, are unreliable. Let $\alpha(t)$ denote the state of a machine at time t. If $\alpha(t) = 1$, the machine is said to be up or working. If $\alpha(t) = 0$, the machine is said to be down or failed. We let $\alpha_{0,j}(t)$ denote the state of supply machine $M_{0,j}$ at the end of time t. For the demand machine, $M_{K+1,j}$, we let the corresponding state variables be $\alpha_{K+1,j}(t)$. For processing machine M_i , the state variable representing the state of the machine at the end of time t is written $\alpha_i(t)$. We make the assumption that all the machines in the line, including the supply and demand machines, have homogeneous processing times. That is, the lengths of time that parts spend in machines are fixed, known in advance, and the same for all the machines. For convenience, the processing times are assumed to be scaled to unity. Furthermore, we assume that the yield of all machines is 100%. That is, we do not model the scrapping or rework of parts.

We assume that all buffers, including the supply and demand buffers, have finite size. The size of buffer $B_{i,j}$ is denoted $N_{i,j}$, where *i* indicates the production stage, and *j*, represents the part type. We let buffers $B_{0,j}$ denote the supply buffers for Type *j*. Likewise, buffers $B_{K,j}$ denote the demand buffers for Type *j*. We denote the level of $B_{i,j}$ at the end of time *t* by $n_{i,j}(t)$. Therefore, $0 \le n_{i,j}(t) \le N_{i,j}$, for all (i, j), and for all $t \ge 0$. A machine is said to be *starved* for a given part type if the upstream buffer corresponding to that part type is empty. It is *blocked* for a given part type if the corresponding downstream buffer is full. We make the assumptions that the supply machines are never starved and the demand machines are never blocked. As mentioned earlier, all machines in the line are assumed to be unreliable. We further assume that machines cannot fail if they are idle. This is called *operation dependent failures*. It means that the supply machines cannot fail if they are blocked and the demand machines cannot fail if they are starved. A processing machine cannot fail if it is either starved or blocked for the all the part types at the same time unit.

All machines are assumed to have geometrically distributed up and down times. We assume that the probability that M_i fails is the same, regardless of the part type the processing machine is working on. We let r_i represent the probability that M_i is up at time t + 1, given it was down at time t. Likewise, p_i represents the probability that M_i is down at time t + 1, given it was up and not blocked or starved at time t. For the supply machines, we let $r_{0,j}$ represent the probability that $M_{0,j}$ is up at time t + 1, given it was down at time t. Also, $p_{0,j}$ represents the probability that $M_{0,j}$ is down at time t + 1, given it was up and not blocked at time t. For the demand machines $M_{K+1,j}$, the corresponding parameters are written $r_{K+1,j}$ and $p_{K+1,j}$. For M_i , the machine parameters defines the state transitions as follows:

$$r_{i} = Pr [\alpha_{i}(t+1) = 1 | \alpha_{i}(t) = 0]$$

$$p_{i} = Pr [\alpha_{i}(t+1) = 0 |$$

$$\{\alpha_{i}(t) = 1 \cap n_{i-1,1}(t) > 0 \cap n_{i,1}(t) < N_{i,1}\} \cup$$

$$\{\alpha_{i}(t) = 1 \cap n_{i-1,2}(t) > 0 \cap n_{i,2}(t) < N_{i,2}\} \cup$$

$$\{\alpha_{i}(t) = 1 \cap n_{i-1,3}(t) > 0 \cap n_{i,3}(t) < N_{i,3}\}]$$
for $i = 1, ..., K$

$$(3.2.1)$$

Likewise, for the supply and demand machines, the machine parameters are defined as:

$$\begin{aligned} r_{0,j} &= \Pr\left[\alpha_{0,j}(t+1) = 1 | \alpha_{0,j}(t) = 0\right] \\ p_{0,j} &= \Pr\left[\alpha_{0,j}(t+1) = 0 | \alpha_{0,j}(t) = 1 \cap n_{0,j}(t) < N_j\right] \\ & \text{for } j = 1, 2, 3 \end{aligned}$$

$$r_{K+1,j} = Pr \left[\alpha_{K+1,j}(t+1) = 1 | \alpha_{K+1,j}(t) = 0 \right]$$

$$p_{K+1,j} = Pr \left[\alpha_{K+1,j}(t+1) = 0 | \alpha_{K+1,j}(t) = 1 \cap n_{K,j}(t) > 0 \right]$$

for $j = 1, 2, 3$

We define e_i to be *isolated production rate* of machine M_i . Then e_i is the production rate of M_i if the machine were never impeded by blockages or starvations. It is given by

$$e_i = \frac{r_i}{r_i + p_i} \tag{3.2.2}$$

It is the same s the isolated efficiency because the operation time is 1. In order to derive the decomposition, we need to make a crucial approximation. We will assume that the probability that a machine M_i is simultaneously starved and blocked for a given part type is negligible. That is, we assume that

$$Pr[n_{i,j}(t) = 0 \cap n_{i+1,j}(t) = N_{2,j}] \approx 0$$

$$i = 1 \dots K$$

$$j = 1, 2, 3$$
(3.2.3)

We justify this approximation with the following argument. In order for the machine to be both starved and blocked for the same part type simultaneously, it is necessary that at some point, the machine had exactly one part in the upstream buffer, and exactly one space in the downstream buffer. At the same time, the upstream machine must be unable to process parts to place in its downstream buffer, and the downstream machine must be unable to process parts, depleting the stores of its upstream buffer. Since the probability of a machine failing is usually small — on the order of 0.01 in a time step — the probability that all three of these occurrences happen at the same time is likely to be quite low. In fact, testing this hypothesis using discrete event simulation has shown that the approximation holds in many systems with moderate sized buffers. Also, this approximation works successfully with the single-part-type decomposition method of Gershwin [7].

3.3 Part Type Priority Policy

Since each machine in the production line must choose which part to work on when it has a choice, we are required to state a policy by which that choice is made. Our assumption is that each machine will work on Type 1 parts whenever the machine is up, the upstream buffer for Type 1 parts is not empty, and the downstream buffer for Type 1 parts is not full. Each machine will only work on Type 2 parts if it is up, and either blocked or starved for Type 1 parts, and not blocked or starved for Type 2 parts. If a machine is blocked or starved for Type 1 and Type 2, it seeks to work for Type 3. In other words, when the machine has a choice, it always chooses the highest priority part type possible.

3.4 Production Rate

Let us denote the production rate of Type j parts at M_i by $E_{i,1}$. This is the fraction of time that M_i is working on Type 1 parts. We know that M_i will make a Type 1 part at the end of time step t + 1 if M_i is not starved for Type 1 parts at time t, M_i is not blocked for Type 1 parts at time t, and M_i is up at the end of time step t + 1. This probability is expressed as follows:

$$E_{i,1} = \Pr\left[\alpha_i(t+1) = 1 \cap n_{i-1,1}(t) > 0 \cap n_{i,1}(t) < N_{i,1}\right]$$
(3.4.1)

Let the quantity $E_{i,2}$ denote the production rate of Type 2 parts. This is the fraction of time that M_i is working on Type 2 parts. From our assumptions, we know that M_i will make a Type 2 part at time t + 1, if M_i is either blocked or starved for Type 1 at time t; M_i is not starved or blocked for Type 2; and M_i is up at the end of time t + 1. This is:

$$E_{i,2} = Pr[\alpha_i(t+1) = 1 \cap (3.4.2)$$

$$(n_{i-1,1}(t) = 0 \cup n_{i,1}(t) = N_{i,1}) \cap (n_{i-1,2}(t) > 0 \cap n_{i,2}(t) < N_{i,2})]$$

Similarly, for the efficiency for Type 3,

$$E_{i,2} = Pr[\alpha_i(t+1) = 1 \cap (3.4.3)$$

$$(n_{i-1,1}(t) = 0 \cup n_{i,1}(t) = N_{i,1}) \cap (n_{i-1,2}(t) = 0 \cup n_{i,2}(t) = N_{i,2}) \cap (n_{i-1,3}(t) > 0 \cap n_{i,3}(t) < N_{i,3})]$$

In steady state, because of conservation of flow, we require that each machine in the line makes the same number of parts — Type 1, Type 2, and Type 3 combined. If we denote the throughput of the demand machine for Type j parts by $E_{K+1,j}$, and the supply machine of Type j parts by $E_{0,j}$, then we must have

$$E_{0,j} = E_{1,j} = E_{2,j} = \dots = E_{i,j} = E_{K+1,j},$$

for $j = 1, 2, 3$

3.5 Basic Idea of Decomposition

We intend to break down the larger system into analytically tractable two-machine lines, which are called *building blocks*. The building blocks capture the local behavior of the long line, as seen by an observer in a buffer, by choosing appropriate parameters of machines This decomposition procedure is represented in Figure 3.5.1. As discussed earlier, the idea is to fool an observer in a buffer in the long, multi-part type processing line into thinking he is in a two-machine line. In the figure, the inflow and outflow behavior of material that an observer in buffer $B_{i,1}$ could see is modeled by the two-machine, one-part line L(i, 1).

Close observation of the dynamics of the long line, however, shows the necessity for a new two-machine line model. The reason is as follows. The building block used in the single-part-type line decomposition is not adequate here. Suppose that we take the point of view of an observer in $B_{i,1}$. We misinform this observer: we lead him to believe that he is watching the flow in the only buffer in a two-machine, one-buffer, one-part type system. Let us assume that the observer sees that the outflow from his buffer has ceased, but the inflow has not. Eventually, unless the outflow resumes or the inflow ceases, $B_{i,1}$ will fill up. According to our scheduling rule, M_i will immediately begin making Type 2 parts, if it is able to. Suppose it does, and that M_i fails while making a Type 2 part. Now suppose that while M_i is down, the outflow from $B_{i,1}$ begins again. Then the sequence of events that the observer will see are that the outflow ceased, the buffer filled up, but when the outflow began again and the buffer was not full, the inflow did not begin. As far as the observer in the buffer is concerned, the machine upstream of him failed while it was blocked.

There is subtlety here. While this apparent idleness failure is behavior that an observer in a buffer sees, the real machines do not fail when they are idle. It only appears to the observer that the machine has failed during an idle period because the observer believes that he is in a two-machine, one-part type line. Therefore, while in the previous model in [7] it is assumed that the pseudo-machines in the two-machine sub-lines had operation dependent failures, we must relax that assumption for the two-machine sub-lines in the multiple-part type case. Thus, a new two-machine line model is in order. We present a discrete-time, discrete-state Markov model of precisely such a line in Chapter 4.



Figure 3.5.1: The decomposition of a line into building blocks

Chapter 4

Two-Machine Lines with Idleness Failures

4.1 Idleness Failure and Failure-Mode Change

As discussed in Chapter 3, in order to decompose the Markov chain model of the two-part-type processing line, we need a new two-machine line building block. The two-machine-line presented here is similar to the deterministic processing time with multiple-failure model described by Tolio [13]. However, the model presented here differs in that the machines are allowed to fail while they are idle.

4.1.1 Idleness Failure

We use the concept of multiple failure modes in constructing the building block. As in the Tolio decomposition, the machines of the building block can fail in *local failure modes* and *remote failure modes*. A local failure mode is a failure of a machine adjacent to the buffer in the real line. A remote failure mode is a failure introduced to account for the effect of a failure of a machine not adjacent to the buffer. In this model, the machines in the building block are no longer restricted to failing only if they are not blocked and not starved. Since a machine in the two-machine line can fail while it is idle — starved or blocked — we call the line *a building block with idleness failure*.

4.1.2 Failure Mode Change

In order to explain the failure mode change, let us consider the following events.

- 1. $M_{i'}$ is failed.
- 2. M_i is blocked. i < i'
- 3. M_i begins to process Type 2 parts.
- 4. M_i gets failed.

Suppose that M_i is blocked for Type 1 due to a failure of a machine in the downstream, for example, $M_{i'}$, i < i'. In this case, the downstream machine, $M^d(i-1,1)$, is failed in a remote failure mode. Now, M_i begins working on Type 2 parts. However, while M_i is working on Type 2, it gets failed. At this moment, $M^d(i-1,1)$ will switch from the original failure mode to a new failure mode. We call this shifting mode change *a failure mode change*. There are two important observations about failure-mode changes. The first is that a failure mode can only change to a mode corresponding to a machine which is closer to the observer. The reason for this is that the initiating failure corresponds to a real failure of some machines, which has propagated by means of starvation or blockage to the observer's location.

4.2 Two-Machine-Line Notation and Parameters

The building blocks are illustrated in Figure 4.1.1. The Markov chain transition graphs of each machine are shown. As is our convention, the machines are denoted by squares, and the buffer by circles. We denote the upstream machine by M^u , and the downstream machine by M^d . We denote the size of the intermediate buffer by N, and the current level of the intermediate buffer by n. It follows that $0 \le n \le N$. We define the state of the two-machine line to be $s = (n, \alpha^u, \alpha^d)$. α^u is Δ_i if M^u is down at mode i, and Υ^u if M^u is up. α^d is Δ_j^d if M^d is down at mode j, and Υ^d if M^d is





Figure 4.1.1: Example of Markov states with three failure modes

up.

Material flows into the upstream machine from an infinite supply, is processed by the machine, and when processing is complete, the material is placed into the buffer until it is processed by the downstream machine. Upon finishing being processed by the downstream machine, the part leaves the line. We assume that there is always room for the downstream machine to unload a part it has just completed processing. We make the assumption that there is only one class of parts produced by the line, and that the production time at each of the machines is identical, and equal to one.

The machines are unreliable and can fail in multiple failure modes. We assume that the machines can fail while they are either operating on a part or idle, but we do not assume that the probabilities of failure are identical. In particular, we assume that the probability that M^u fails into mode *i* while it is working on a part, given it is not blocked, is p_i^u , and the probability that it fails into mode *j* while it is blocked is q_j^u . Note that we do *not* assume that there is a new failure mode, but only that there is a new way of reaching the failure mode. We define the quantities p_i^d and q_j^d for M^d similarly. Finally, we denote the probabilities that M^u and M^d are repaired while they are down at failure mode j by r_j^u and r_j^d , respectively.

The quantity $z_{i,i'}^u$ is the probability of the upstream machine changing from down mode *i* to down mode *i'*. The expression $z_{j,j'}^d$ represents the probability that the downstream machine changes from down mode *j* to down mode *j'*. Defining $\alpha^{\dagger}(t)$ as the state (up state or down state) of a machine \dagger at time *t* (where \dagger is either *u* or *d* for upstream or downstream), then we can define *r*, *p*, *q*, and *z* as

$$\begin{split} r_{j,i}^{\dagger} &= \Pr[\alpha^{\dagger}(t+1) = \Upsilon_{i}^{\dagger} \mid \alpha^{\dagger}(t) = \Delta_{j}^{\dagger}] \\ p_{i,j}^{\dagger} &= \Pr[\alpha^{\dagger}(t+1) = \Delta_{j}^{\dagger} \mid \alpha^{\dagger}(t) = \Upsilon_{i}^{\dagger} \text{ and } n(t) < N] \\ q_{i,j}^{\dagger} &= \Pr[\alpha^{\dagger}(t+1) = \Delta_{j}^{\dagger} \mid \alpha^{\dagger}(t) = \Upsilon_{i}^{\dagger} \text{ and } n(t) = N] \\ z_{j,j'}^{\dagger} &= \Pr[\alpha^{\dagger}(t+1) = \Delta_{j'}^{\dagger} \mid \alpha^{\dagger}(t) = \Delta_{j}^{\dagger}] \\ \text{ for } \dagger = u \text{ and } d \end{split}$$

We also define P^u and P^d such that

$$P^u = \sum_{j=1}^J p_j^u$$
 and $P^d = \sum_{l=1}^L p_l^d$

where J and L are the numbers of failure modes for the upstream machine and downstream machine respectively. The set of parameters p_j^u and p_l^d must be such that $P^u < 1$ and $P^d < 1$. We define

$$Q^u = \sum_{j=1}^J q_j^u$$
 and $Q^d = \sum_{l=1}^L q_l^d$

and again, the set of parameters q_j^u and q_l^d must be such that $Q^u < 1$ and $Q^d < 1$.

4.3 Transition Equations

4.3.1 Transient states

In the steady-state, the probability of a two-machine line being on a transient state is zero. Since idleness failure is considered, there are fewer transient states than in Gershwin and Tolio's model. The following are transient states:

- $P(0, \Upsilon^u, \Delta_l^d)$ and $P(0, \Upsilon^u, \Upsilon^d)$: Since the upstream machine is up, it puts a part at the beginning of a time step and therefore the buffer will have the part at the end of the time step. There is no transition to this state.
- $P(N, \Delta_j^u, \Upsilon^d)$ and $P(N, \Upsilon^u, \Upsilon^d)$: Since the downstream machine is up it takes out a part at the beginning of a time step and therefore, buffer will have a part always less then N at the end of the time step. There is no transition to these states.

4.3.2 Non-transient states

A internal state is the state where buffer level satisfies $2 \le n \le N-2$. The system can get from $P(n, \Delta_j^u, \Upsilon^d)$ to $P(n, \Delta^u, \Delta^d)$ if the upstream machine is repaired from failure mode j but the downstream machine remains down. The probability of this is $r_j^u(1-P^d)$. It can move from $P(n, \Upsilon^u, \Delta_l^d)$ to $P(n, \Delta^u, \Delta^d)$ if the upstream machine is not repaired but the downstream machine is repaired from mode l. Therefore the transition probability is $(1-Pu)r_l^d$. The transition probability from $P(n, \Delta_j^u, \Delta_l^d)$ to $P(n, \Upsilon^u, \Upsilon^d)$ is $r_j^u r_l^d$ since the upstream and downstream machines are repaired from the down mode j and l, respectively. Likewise, the transition probability from $P(n, \Upsilon_j^u, \Upsilon_l^d)$ to $P(n, \Upsilon^u, \Upsilon^d)$ is $(1-P^u)(1-P^d)$ since the upstream and downstream machines remain up. No other transitions from non-transient states are possible. Consequently, the first equation is:

$$P(n, \Upsilon^{u}, \Upsilon^{d}) = \sum_{j=1}^{J} \sum_{l=1}^{L} P(n, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} r_{l}^{d} + \sum_{j=1}^{J} P(n, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} (1 - P^{d}) + \sum_{l=1}^{L} P(n, \Upsilon^{u} \Delta_{l}^{d}) (1 - P^{u}) r_{l}^{d} + P(n, \Upsilon^{u} \Upsilon^{d}) (1 - P^{u}) (1 - P^{d})$$
(4.3.1)

The other equations are similar and they are represented in the Appendix. The lower boundary states $(n \leq 1)$ and upper boundary $(n \geq N - 1)$ are also shown in the Appendix.

4.3.3 State Classification

For our convenience and later use, we define the following quantities. For the upstream machine,

- $W^u \quad P(\alpha^u = \Upsilon^u, n < N)$
- $X_j^u \quad P(\alpha^u = \Delta_j^u, n < N)$, (*remote failure $j = 1 \dots J 1$, real failure j = J).
- $D_b^u \quad P(\alpha^u = \Delta_J^u, n = N, \alpha^d = \Delta_l^d)$, (*only real failure Δ_J^d).
- $P_b \quad P(\alpha^u = \Upsilon^u, n = N, \alpha^d = \Delta^d_l)$

These states are mutually exclusive and collectively exhaustive. Therefore, we can state

$$W^{u} + \sum_{j=1}^{J} X_{j}^{u} + \sum_{l=1}^{L} Db_{l}^{u} + \sum_{l=1}^{L} Pb_{l} = 1$$
(4.3.2)

For the downstream machine,

• $W^d \quad P[\alpha^d = \Upsilon^d, n > 0q]$
- $X_l^d \quad P[\alpha^d = \Delta_l^d, n > 0],$ (*remote failure l = 2...L, real failure l = 1).
- $D_s^d \quad P[\alpha^u = \Delta_j^u, \alpha^d = \Delta_1^d, n = 0],$ (*only real failure Δ_1^d).
- $P_s \quad P[\alpha^u = \Delta^u_j, \alpha^d = \Upsilon^d, n = 0]$

These states are mutually exclusive and collectively exhaustive. Therefore, we can states such that

$$W^{d} + \sum_{l=1}^{L} X_{l}^{d} + \sum_{j=1}^{J} Ds_{l}^{d} + \sum_{j=1}^{J} Ps_{j} = 1$$
(4.3.3)

4.4 Performance Measures

The main performance measures of interest in the two-machine line are the *throughput*, average buffer level, probability of starvation, and probability of blockage.

4.4.1 Efficiency

The throughput of the upstream machine is the probability that the upstream machine is working in time step t + 1, and not blocked at time t. This is,

$$E^u = \Pr[\alpha^u(t+1) = \Upsilon^u \cap n(t) < N]$$

$$(4.4.1)$$

Similarly, the throughput of the downstream machine is the probability that the downstream machine is working at time t + 1 and not starved at time t. This is,

$$E^{d} = Pr[\alpha^{d}(t+1) = \Upsilon^{u} \cap n(t) > 0]$$
(4.4.2)

Observe the equation in (4.4.1) that this expression has both time step t + 1 and time step t in it. We proceed by conditioning on events occurring time step t to write (4.4.1) in terms of events occurring entirely in time step t. By doing so, we will be able to express the production rate of the upstream machine entirely in terms of the state probabilities, which are defined only for one time step. Before we derive a single-time-step expression for the efficiency, notice that for every machine failure, there is also a repair. That is, for both the up-and downstream machines in the building block, the probability that a repair occurs at any time (the repair frequency) equals the probability that a failure occurs at any time(the failure frequency), consequently. For the upstream machine, that is expressed as

$$r_{j}^{u}(Pr[\alpha^{u} = \Delta_{j}^{u} \cap n < N] + Pr[\alpha^{u} = \Delta_{j}^{u} \cap n = N])$$

$$= p_{j}^{u}Pr[\alpha^{u} = \Upsilon_{j}^{u} \cap n < N] + q_{j}^{u}Pr[\alpha^{u} = \Upsilon_{j}^{u} \cap n = N]$$

$$(4.4.3)$$

Likewise, for the downstream machine,

$$r_l^d (Pr[\alpha^d = \Delta_l^d \cap n > 0] + Pr[\alpha^d = \Delta_l^d \cap n = 0])$$

$$= p_l^d Pr[\alpha^d = \Upsilon_l^d \cap n > 0] + q_l^d Pr[\alpha^d = \Upsilon_l^d \cap n = 0]$$

$$(4.4.4)$$

We can use (4.4.3) and (4.4.4) to derive expressions for the efficiencies for upstream and downstream machines defined in (4.4.1) and (4.4.2):

$$\begin{split} E^u &= \Pr[\alpha^u(t+1) = \Upsilon^u \cap n(t) < N] \\ &= \Pr[\alpha^u(t+1) = \Upsilon^u | \alpha^u(t) = \Upsilon^u \cap n(t) < N] \\ &\times \Pr[\alpha^u(t) = \Upsilon^u \cap n(t) < N] \\ &+ \sum_{j=1}^J (\Pr[\alpha^u(t+1) = \Upsilon^u | \alpha^u(t) = \Delta_j^u \cap n(t) < N] \\ &\times \Pr[\alpha^u(t) = \Delta_j^u \cap n(t) < N]) \\ &= (1 - P^u) \Pr[\alpha^u(t) = \Upsilon^u \cap n(t) < N] \\ &+ \sum_{j=1}^J r_j^u \Pr[\alpha^u(t) = \Delta_j^u \cap n(t) < N] \end{split}$$

If we apply the fact that the repair frequency equals failure frequency (4.4.3), then E^u is

$$E^{u} = Pr[\{\alpha^{u}(t) = \Upsilon^{u}\} \cap \{n(t) < N\}]$$

+ $Q^{u}Pr[\{\alpha^{u}(t) = \Upsilon^{u}\} \cap \{n(t) = N\}]$
- $\sum_{j=1}^{J} r_{j}^{u}Pr[\{\alpha^{u}(t) = \Delta_{j}^{u}\} \cap \{n(t) = N\}]$

Since

$$Pr[\{\alpha^{u}(t) = \Upsilon^{u}\} \cap \{n(t) = N\}] = \sum_{l=1}^{L} Pr(N, \Upsilon^{u}, \Delta_{l}^{d})$$
$$Pr[\{\alpha^{u}(t) = \Delta_{j}^{u}\} \cap \{n(t) = N\}] = \sum_{l=1}^{L} Pr(N, \Delta_{j}^{u}, \Delta_{l}^{d})$$

 E^u can be

$$E^{u} = \sum_{n=0}^{N-1} Pr(n, \Upsilon^{u}, \Upsilon^{d}) + \sum_{n=0}^{N-1} \sum_{l=1}^{L} Pr(n, \Upsilon^{u}, \Delta_{j}^{d})$$

$$+ Q^{u} \sum_{l=1}^{L} Pr(N, \Upsilon^{u}, \Delta_{l}^{d}) - \sum_{j=1}^{J} r_{j}^{u} \sum_{l=1}^{L} Pr(N, \Delta_{j}^{u}, \Delta_{l}^{d})$$

$$(4.4.5)$$

Note that if we were not considering idleness failure, Q^u would be zero and the last term of (4.4.5) also would be zero, since there will be no transition to the state $(N, \Delta_j^u, \Delta_l^d)$. Therefore, only the first two terms of (4.4.5) would be left. In this case, the efficiency is the probability that the upstream machine is up and buffer is not full. This expression is the same as the one described by Gershwin [8] in which the model does not consider idleness failure. Similarly for the downstream machine:

$$\begin{split} E^{d} &= Pr[\{\alpha^{d}(t) = \Upsilon^{d}\} \cap \{n(t) > 0\}] \\ &+ Q^{d}Pr[\{\alpha^{d}(t) = \Upsilon^{d}\} \cap \{n(t) = 0\}] \\ &- \sum_{1=1}^{L} r_{l}^{d}Pr[\{\alpha^{d}(t) = \Delta_{j}^{d}\} \cap \{n(t) = 0\}] \end{split}$$

Since,

$$Pr[\{\alpha^{d}(t) = \Upsilon^{d}\} \cap \{n(t) = 0\}] = \sum_{j=1}^{J} Pr(0, \Delta_{j}^{u}, \Upsilon^{d})$$
$$Pr[\{\alpha^{d}(t) = \Delta_{j}^{d}\} \cap \{n(t) = 0\}] = \sum_{j=1}^{J} Pr(0, \Delta_{j}^{u}, \Delta_{l}^{d})$$

, we can rewrite E^d as

$$E^{d} = \sum_{n=1}^{N} Pr(n, \Upsilon^{u}, \Upsilon^{d}) + \sum_{n=1}^{N} \sum_{j=1}^{J} Pr(n, \Delta_{j}^{u}, \Upsilon^{d})$$

$$+ Q^{d} \sum_{j=1}^{J} Pr(0, \Delta_{j}^{u}, \Upsilon^{d}) - \sum_{l=1}^{L} r_{l}^{d} \sum_{j=1}^{J} Pr(0, \Delta_{j}^{u}, \Delta_{l}^{d})$$
(4.4.6)

Note that in the expressions (4.4.6), if we were not consider the idleness failure, only the first two terms would be left. In this case, the efficiency is the probability that the downstream machine is up and there is at least one part in the buffer.

4.4.2 Average buffer level, probability of blockage, and probability of starvation

The average buffer level is denoted \bar{n} , and is given by

$$\bar{n} = \sum_{n=0}^{N} nP(n, \Upsilon^{u}, \Upsilon^{d}) + \sum_{j=1}^{J} \sum_{n=0}^{N} nP(n, \Delta_{j}^{u}, \Upsilon^{d})$$

$$+ \sum_{l=1}^{L} \sum_{n=0}^{N} nP(n, \Upsilon^{u}, \Delta_{l}^{d}) + \sum_{j=1}^{J} \sum_{l=1}^{L} \sum_{n=0}^{N} nP(n, \Delta_{j}^{u}, \Delta_{l}^{d})$$
(4.4.7)

The probability that the downstream machine is starved, denoted P_s , is the probability that the machine is up, and there is no part in the buffer. This is given by

$$P_s = P(\alpha^u = \Upsilon^u, \alpha^d = \Delta_l^d, n = N) \tag{4.4.8}$$

The probability that the upstream is blocked, denoted P_b , is the probability that the machine is up, and that the intermediate buffer is full. This is expressed as

$$P_b = P(\alpha^u = \Delta_j^u, \alpha^d = \Upsilon^d, n = 0)$$
(4.4.9)

4.5 Solution Algorithm

In this section, we propose a numerical solution algorithm to solve the transition equations expressed in Section 4.3. We denote the number of states by η and it is

$$\eta = (N+1)(J+1)(L+1)$$

, where J and L are the total number of the down states for ${\cal M}^u$ and ${\cal M}^d,$ respectively.

The steady state transition equations of the system has a form

$$p = Ap \tag{4.5.1}$$



Figure 4.5.1: Example of a sparsity pattern J = 3 and L = 3

$$v^T p = 1 \tag{4.5.2}$$

where

- p is an unknown n-vector.
- A is an $\eta \times \eta$ matrix. The rank if A I is $\eta 1$.
- v is an η -vector, each of whose elements is 1.

Note that the matrix, A, comes from the transition equations in the Appendix.

We know that p is the steady-state probability distribution of a Markov process and A is the transition matrix. The basic method of solving the linear equation is iterating the equation

$$p(k+1) = Ap(k)$$
(4.5.3)

where p(0) is chosen to satisfy (4.5.2), until p(k) converges. Practical convergence is defined, in this case, according to a criterion like

$$\delta_i(k+1) = |p_i(k+1) - p_i(k)| < \epsilon \quad \text{for all } i \tag{4.5.4}$$

for some suitable ϵ .

Note that most elements in the matrix A are zero. Figure 4.5.1 shows an example of matrix A. In this matrix, a nonzero block is indicated with 1 and zero element is left blank. As the figure shows, the majority of elements are zero. We can take the advantage of the sparsity when we numerically solve the equation.

Chapter 5

Decomposition Analysis for Type 1

In this chapter, we introduce the derivation of the decomposition equations for Type 1. The building block that imitates the flow in $B_{i,j}$ denoted by L(i,j). For example, L(1,2) represents the first building block imitating the behavior of Type 2. The upstream and downstream machines in L(i,j) are denoted by $M^u(i,j)$ and $M^d(i,j)$. We consider the perspective of an observer in a buffer for Type 1 part, $B_{i,1}$. Then we seek to capture the upstream and downstream behavior, as seen by the observer.

5.1 State and Parameter Definitions

The downstream machine, $M^d(i, 1)$ (i = 0, ..., K - 1), represents the Type 1 flow behavior as it leaves $B_{i,1}$. The downstream machine $M^d(i, 1)$ is up when M_{i+1} is up and is not blocked for a Type 1 part. That is,

$$\Upsilon^d(i,1) = \{\alpha_{i+1} = 1 \cap n_{i+1,1} < N_{i+1,1}\}$$

That is, the observer in $B_{i,1}$ will see a part moving out of the buffer, when M_{i+1} is up is not blocked for Type 1. There are three cases in which the observer does not see a part moving out of the buffer and therefore believes that $M^d(i, 1)$ is down. These cases are:

• $\Delta_1^d(i, 1) = \{M_{i+1} \text{ is down}\}$

- $\Delta_1^d(i,1) = \{M_{i+1} \text{ is up but blocked due to a failure of a machine downstream}\}$
- $\Delta_1^d(i, 1) = \{M_{i+1} \text{ is down and also blocked due to a failure of a machine down stream}\}$

Note that outflow from the buffer may stop for two reasons. First, the machine directly downstream of the buffer may have failed. This case is called the *local failure*. For the the other case, a machine further downstream have failed, blocking the upstream of it. The first down state, $\Delta_1^d(i, 1)$ represents the failure of the machine immediate downstream of the buffer the observer is watching. The second down state, $\Delta_2^d(i, 1)$, indicates a failure in a machine further downstream. This case is called the *remote failure*. Therefore, the first down state, $\Delta_1^d(i, 1)$, is caused by the local failure, while the second down state, $\Delta_2^d(i, 1)$, is caused by the remote failure.

The first and second down states can be seen in a building block for the single-parttype line decomposition. The last down state, $\Delta_3^d(i, 1)$, exists only when machines are processing multiple part types. This down state is a mixture of local and remote failures. This failure occurs when the following sequence of failures occurs: Suppose M_{i+2} is down. If that failure persists long enough, it will make $B_{i+1,1}$ full, causing the blockage of M_{i+1} . Now, M_{i+1} is blocked for Type 1 parts, and $M^d(i, 1)$ will be down in the second down state described above. While M_{i+1} is blocked for Type 1, it may work on a Type 2 part. Now suppose that M_{i+1} fails while it is working on a Type 2 part. At this moment, M_{i+2} is down and $B_{i+1,1}$ is full and M_{i+1} is also down. The observer in $B_{i,1}$ sees that its downstream machine is not only down but also blocked for Type 1 parts. In order to get into this down state, the remote failure must occur first before the local failure takes place. This is because the blockage cannot occur when the machine is down already. Therefore, from the observer's view point, the third down state, $\Delta_3^d(i, 1)$, can be reached only from the second down state, $\Delta_2^d(i, 1)$.

The states of $M_d(i, 1)$ defined in terms of line L are

$$\Upsilon^{d}(i,1) = \{\alpha_{i+1} = 1 \cap n_{i+1,1} < N_{i+1,1}\}$$

$$\Delta_{1}^{d}(i,1) = \{\alpha_{i+1} = 0 \cap n_{i+1,1} < N_{i+1,1}\}$$

$$\Delta_{2}^{d}(i,1) = \{\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}\}$$

$$\Delta_{3}^{d}(i,1) = \{\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1}\}$$

$$i = 0 \dots K - 1$$
(5.1.1)

Similarly, the state definitions for $M^u(i,1)$ are

$$\Upsilon^{u}(i,1) = \{\alpha_{i} = 1 \cap n_{i-1,1} > 0\}$$

$$\Delta_{1}^{u}(i,1) = \{\alpha_{i} = 0 \cap n_{i-1,1} > 0\}$$

$$\Delta_{2}^{u}(i,1) = \{\alpha_{i} = 1 \cap n_{i-1,1} = 0\}$$

$$\Delta_{3}^{u}(i,1) = \{\alpha_{i} = 0 \cap n_{i-1,1} = 0\}$$

$$i = 1 \dots K$$
(5.1.2)

5.2 Equalities

For convenience, we define the following probabilities for a building block:

$$W^{u}(i,1) = Pr[\Upsilon^{u}(i,1) \cap n_{i,1} > 0]$$

$$W^{d}(i,1) = Pr[\Upsilon^{d}(i,1) \cap n_{i,1} > 0]$$

$$X^{u}_{m}(i,1) = Pr[\Delta^{u}_{m}(i,1) \cap n_{i,1} < N_{i,1}], m = 1,2,3 \qquad (5.2.1)$$

$$X^{d}_{m}(i,1) = Pr[\Delta^{d}_{n}(i,1) \cap n_{i,1} > 0], n = 1,2,3$$

$$P_{b}(i,1) = Pr[\Upsilon^{u}(i,1) \cap n_{i,1} = N_{i,1}]$$

$$P_{s}(i,1) = Pr[\Upsilon^{d}(i,1) \cap n_{i,1} = 0]$$

$$D_{b}(i,1) = Pr[\Delta^{u}_{1}(i,1) \cap n_{i,1} = N_{i,1}]$$

$$D_{s}(i,1) = Pr[\Delta^{d}_{1}(i,1) \cap n_{i,1} = 0]$$

$$i = 0 \dots K$$

The quantity, $W^u(i, 1)$, is the probability that $M^u(i, 1)$ is up and not starved. The quantity $X_m^u(i, 1)$ is the probability that $M^u(i, 1)$ is in the down state $\Delta_m^u(i, 1), m =$ 1,2,3. $P_b(i, 1)$ is the probability that $M^u(i, 1)$ is up but idle because of blockage. $D_b(i, 1)$ is the probability that $M^u(i, 1)$ is down and also blocked for a part. Likewise, the quantity, $W^d(i, 1)$, is the probability that $M^d(i, 1)$ is up and not blocked. The quantity $X_m^d(i, 1), m = 1, 2, 3$ is the probability that $M^d(i, 1)$ is down at the down state, $\Delta_m^d(i, 1)$. $P_s(i, 1)$ is the probability that $M^d(i, 1)$ is up but idle because of starvation. $D_s(i, 1)$ is the probability that M_{i+1} is down and also starved for a part.

 $W^{d}(i, 1)$ is the probability that that M_{i+1} is up and neither starved nor blocked for Type 1. If we relate this quantity in the building block with events the real line, then

$$W^{d}(i,1) = Pr[\Upsilon^{d}(i,1) \cap n_{i,1} > 0]$$

= $Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}]$

Notice that this expression can be written as

$$Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}]$$

= $Pr[\Upsilon^u(i+1,1) \cap n_{i+1,1} < N_{i+1,1}]$
= $W^u(i+1,1)$

Therefore,

$$W^{d}(i,1) = W^{u}(i+1,1)$$
(5.2.2)

This equality is a statement of the conservation of flow.

Next, $X_1^d(i, 1)$ is the probability that the downstream machine of L(i, 1) is down in mode 1 and not starved. From definitions (5.1.1) and (5.2.1), it is

$$\begin{aligned} X_1^d(i,1) &= Pr[\Delta_1^d(i,1) \cap n_{i,1} > 0] \\ &= Pr[\alpha_{i+1} = 0 \cap n_{i+1,1} < N_{i+1,1} \cap n_{i,1} > 0] \\ &= Pr[\Delta^u(i+1,1) \cap n_{i+1,1} < N_{i+1,1}] \\ &= X_1^u(i+1,1) \end{aligned}$$

Therefore,

$$X_1^d(i,1) = X_1^u(i+1,1)$$
(5.2.3)

Next, $X_2^d(i, 1)$ is the probability that the downstream machine of L(i, 1) is down in mode 2 and not starved. From (5.1.1), it is

$$\begin{aligned} X_2^d(i,1) &= Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} = N_{i+1,1}] \\ &= Pr[\Upsilon^u(i+1,1) \cap n_{2,1} = N_{i+1,1}] \\ &= P_b(i+1,1) \end{aligned}$$

A similar equality can be derived for $X_2^u(i+1,1)$. Therefore,

$$X_2^d(i,1) = P_b(i+1,1) (5.2.4)$$

$$X_2^u(i+1,1) = P_s(i,1) (5.2.5)$$

Last,

$$X_3^d(i,1) = Pr[\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} = N_{i+1,1}]$$
$$= Pr[\Delta_1^u(i+1,1) \cap n_{i+1,1} = N_{i+1,1}]$$
$$= D_b(i+1,1)$$

Again, $X_3^u(i+1,1)$ can be derived in the similar way. Therefore,

$$X_3^d(i,1) = D_b(i+1,1)$$
(5.2.6)

$$X_3^u(i+1,1) = D_s(i,1) (5.2.7)$$

5.3 Resumption of Flow

Resumption of flow is a transition from a down state to an up state. Before we formulate the resumption of flow equations, we first define the following quantity for our convenience. Let $R^d(i, 1)$, the resumption of flow probability of $M^d(i, 1)$ denote the probability that $M^d(i, 1)$ is up at t + 1 given that it was down at t. This quantity

is actually the weighted sum of the transition probabilities from all the down states to the up state. Therefore,

$$R^{d}(i,1) = \frac{\sum_{n=1}^{3} r_{n}^{d}(i,1) X_{n}^{d}(i,1)}{\sum_{n=1}^{3} X_{n}^{d}(i,1)}$$

likewise,

$$R^{u}(i,1) = \frac{\sum_{m=1}^{3} r_{m}^{u}(i,1) X_{m}^{u}(i,1)}{\sum_{m=1}^{3} X_{m}^{d}(i,1)}$$

Now, let us formulate each resumption of flow equation. First, $r_1^d(i, 1)$, the transition probability from the first down state to the up state is

$$r_{1}^{d}(i,1) = Pr[\Upsilon^{d}(i,1) \text{ at } t+1 |\Delta_{1}^{d}(i,1) \text{ at } t]$$

$$= Pr[\alpha_{i+1}(t+1) = 1 \cap$$

$$n_{i+1,1}(t+1) < N_{i+1,1} |\alpha_{i+1}(t) = 0 \cap n_{i+1,1}(t) < N_{i+1,1}]$$

$$= r_{i+1}$$
(5.3.1)

That ism, since $\Delta_1^d(i, 1)$ represents a local failure, its repair probability is equal to the repair probability of the real machine M_{i+1} . Next, the repair probability of the remote down state is

$$r_{2}^{d}(i,1) = Pr[\Upsilon^{d}(i,1) \text{ at } t+1 |\Delta_{2}^{d}(i,1) \text{ at } t]$$

$$= Pr[\alpha_{i+1}(t+1) = 1 \cap n_{i+1,1}(t+1) < N_{i+1,1}|$$

$$\alpha_{i+1}(t) = 1 \cap n_{i+1,1}(t) = N_{i+1,1}]$$

$$= \left(1 - q_{1}^{u}(i+1,1)\right) R^{d}(i+1,1)$$
(5.3.2)

This is because the remote failure occurred due to the blockage of $B_{i+1,1}$ which

was caused by the failure of M_{i+2} or some further downstream machine. Therefore, $M^d(i+1,1)$ must be repaired in order for flow of $M^d(i,1)$ to resume. There is also one more condition required for the resumption of the flow. Notice that at time t, M_{i+1} is blocked and therefore, it could work on a Type 2 or lower priority part. The resumption of the flow happens at the next time step only when M_{i+1} does not go down while it is working on Type 2. The probability that M_{i+1} goes down while it is blocked and processing a Type $j, j \geq 2$ part is $q_1^u(i, 1)$ by the definition. Therefore, the transition probability from $\Delta_2^d(i, 1)$ to $\Upsilon^d(i, 1)$ is $(1-q_1^u(i+1, 1))R^d(i+1, 1)$.

For the third repair probability, $r_3^d(i, 1)$ is the transition probability from $\Delta_3^d(i, 1)$ to $\Upsilon^d(i, 1)$. We argue that this is

$$r_{3}^{d}(i,1) = Pr[\Upsilon^{d}(i,1) \text{ at } t+1 |\Delta_{3}^{d}(i,1) \text{ at } t]$$

$$= Pr[\alpha_{i+1}(t+1) = 1 \cap n_{i+1,1}(t+1) < N_{i+1,1}|$$

$$\alpha_{i+1}(t) = 0 \cap n_{i+1,1}(t) = N_{i+1,1}]$$

$$= r_{i+1}R^{d}(i+1,1)$$
(5.3.3)

As shown in the equation, M_{i+2} or some further downstream machine is down at t, causing $B_{i+1,1}$ to be full, and also M_2 is down at time t. Therefore, the both machines must be repaired for the flow of a Type 1 part to resume.

Similarly, the resumption of flow equations for $M^u(i+1,1)$ are:

$$r_1^u(i+1,1) = r_{i+1} (5.3.4)$$

$$r_2^u(i+1,1) = (1-q^d(i,1))R^u(i,1)$$
 (5.3.5)

$$r_3^u(i+1,1) = R^u(i,1)r_{i+1} (5.3.6)$$

5.4 Failure Mode Change

Failure mode changes are transitions taking place between down states. First, we consider transitions from $\Delta_1^d(i, 1)$ to other down states. This down state is the failure of M_{i+1} . Since the local machine is down and is not blocked, any state change further downstream this will not change the state of $M^d(i, 1)$. The only event that will change the state of $\Delta_1^d(i, 1)$ is the resumption of flow. Therefore, there is no transition from $\Delta_1^d(i, 1)$ to the rest of the down states. That is,

$$z_{1,2}^d(i,1) = 0 (5.4.1)$$

$$z_{1,3}^d(i,1) = 0 (5.4.2)$$

Next, we consider $\Delta_2^d(i, 1)$. In this down state, M_{i+1} is up but is blocked for a Type 1 part. Both other down states of $M^d(i, 1)$ can be reached from this one. First, consider the case in which M_{i+1} goes down, while processing Type 2 or lower priority part. In this case, M_{i+1} will be down and blocked at the same time. Therefore $M^d(i, 1)$ will move to $\Delta_3^d(i, 1)$. Since M_{i+1} goes down while it is blocked and $M^d(i+1, 1)$ is not repaired, the transition probability from $\Delta_2^d(i, 1)$ to $\Delta_3^d(i, 1)$ is $q_1^u(i+1, 1) \left(1 - R^d(i+1, 1)\right)$. The probability of this failure mode change is

$$z_{2,3}^{d}(i,1) = Pr[\Delta_{3}^{d}(i,1) \text{ at } t+1 \mid \Delta_{2}^{d}(i,1) \text{ at } t \mid]$$

$$= q_{1}^{u}(i+1,1) \left(1 - R^{d}(i+1,1)\right)$$
(5.4.3)

The second failure mode change from $\Delta_2^d(i, 1)$ is the case that M_{i+1} goes down while it is blocked, but $M^d(i+1, 1)$ gets repaired. In this case, M_{i+1} will be no longer blocked, but will move to the local failure mode. Therefore, with the transition probability of $q^u(i+1, 1)R^d(i+1, 1)$, $M^d(i, 1)$ will move from $\Delta_2^d(i, 1)$ to $\Delta_1^d(i, 1)$. That is,

$$z_{2,1}^d(i,1) = Pr[\Delta_1^d(i,1) \text{ at } t+1 \mid \Delta_2^d(i,1) \text{ at } t \mid]$$

$$= q_1^u(i+1,1)R^d(i+1,1)$$
(5.4.4)

The last down state to be considered is $\Delta_3^d(i, 1)$. In this state, $M^d(i, 1)$ is down because of the local failure of M_{i+1} and the blockage caused by the failure of M_{i+2} or some further downstream machine. The failure mode change to $\Delta_2^d(i, 1)$ happens when M_{i+1} gets repaired, but $M^d(i+1, 1)$ remains down. That is,

$$z_{3,2}^{d}(i,1) = Pr[\Delta_{2}^{d}(i,1) \text{ at } t+1 \mid \Delta_{3}^{d}(i,1) \text{ at } t]$$

$$= r_{i+1} \Big(1 - R^{d}(i+1,1) \Big)$$
(5.4.5)

On the other hand, when $M^d(i, 1)$ is in $\Delta_3^d(i, 1)$, if $M^d(i+1, 1)$ gets fixed while M_{i+1} remains down, M_{i+1} will be no longer be blocked for a Type 1 part. Instead it will be in a local failure mode. Therefore,

$$z_{3,1}^{d}(i,1) = Pr[\Delta_{1}^{d}(i,1) \text{ at } t+1 \mid \Delta_{3}^{d}(i,1) \text{ at } t]$$

$$= (1-r_{i+1})R^{d}(i+1,1)$$
(5.4.6)

Following similar approaches, we also can derive the failure mode change probabilities for $M^u(i+1,1)$:

$$z_{1,2}^u(i+1,1) = 0 (5.4.7)$$

 $z_{1,3}^u(i+1,1) = 0 (5.4.8)$

$$z_{2,1}^{u}(i+1,1) = q_1^{d}(i,1)R^{u}(i,1)$$
(5.4.9)

$$z_{2,3}^{u}(i+1,1) = \left(1 - R^{u}(i,1)\right)q_{1}^{d}(i,1)$$
(5.4.10)

$$z_{2,1}^{u}(i+1,1) = q_1^d(i,1)R^u(i,1)$$
(5.4.11)

$$z_{3,1}^{u}(i+1,1) = R^{u}(i,1)\left(1-r_{i+1}\right)$$
(5.4.12)

$$z_{3,2}^{u}(i+1,1) = (1-R^{u}(i,1))r_{i+1}$$
 (5.4.13)

5.5 Interruption of Flow

The interruption of flows of out $B_{i,j}$ are failures of $M^d(i, 1)$. The quantity $p_1^d(i, 1)$ is the probability of from $\Upsilon^d(i, 1)$ to $\Delta_1^d(i, 1)$, which represents the failure of M_{i+1} when $B_{i,1}$ is not empty. This transition probability is

$$p_{1}^{d}(i,1) = Pr[\Delta_{1}^{d}(i,1) \text{ at } t+1 | \qquad (5.5.1)$$

$$\Upsilon^{d}(i,1) \cap n_{i,1} > 0 \text{ at } t]$$

$$= Pr[\alpha_{2}(t+1) = 0 \cap n_{i+1,1}(t+1) < N_{i+1,1}|$$

$$\alpha_{i+1}(t) = 1 \cap n_{i,1}(t) > 0 \cap n_{i+1,1}(t) < N_{i+1,1}]$$

$$= p_{i+1}$$

Next, we derive the interruption of flow equation for $p_2^d(i, 1)$, the transition probability from the up state to the state in which $M^d(i+1, 1)$ is down and $B_{i+1,1}$ is full. This transition probability is

$$p_2^d(i,1) = Pr[\Delta_2^d(i,1) \text{ at } t+1 | \Upsilon^d(i,1) \cap n_{i,1} > 0 \text{ at } t]$$

We first start the derivation of this quantity by applying the fact that the prob-

ability of going out of a state is equal to the probability of going into that state. In this case,

$$X_2^d(i,1) \qquad \left(\left(1 - q^u(i+1,1) \right) R^d(i+1,1) + q^u(i+1,1) R^d(i+1,1) + q^u(i+1,1) \left(1 - R^d(i+1,1) \right) \right)$$
$$= W^d(i,1) p_2^d(i,1) + X_3^d(i+1,1) r_{i+1} (1 - R^d(i+1,1))$$

If we simplify this equation, then

$$X_2^d(1,1) \left(R^d(i+1,1) + q^u(i+1,1) \left(1 - R^d(i+1,1) \right) \right)$$

= $W^d(i,1) p_2^d(i,1) + X_3^d(i,1) r_{i+1} \left(1 - R^d(i+1,1) \right)$

That is,

$$p_{2}^{d}(i,1) = \frac{1}{W^{d}(i,1)} \times$$

$$\left[X_{2}^{d}(i,1) \left(R^{d}(i+1,1) + q^{u}(i+1,1) \left(1 - R^{d}(i+1,1) \right) \right) - X_{3}^{d}(i,1)r_{i+1}(1 - R^{d}(i+1,1)) \right]$$
(5.5.2)

Next, we consider the interruption of flow equation for the third down state, $\Delta_3^d(i, 1)$. This is the transition that goes to the state in which M_{i+1} is down and blocked for a Type 1 part. Note that when M_{i+1} is up and not blocked for a Type 1 part from $\Upsilon^d(i, 1)$, it is impossible for M_{i+1} to be down and blocked at the next time step because once the machine is down, it will not move a part into the downstream buffer. Therefore,



Figure 5.5.1: Markov Chain of $M^u(i,1)$ and $M^d(i,1)$

$$p_{3}^{d}(i,1) = Pr[\Delta_{3}^{d}(i,1) \text{ at } t+1 |$$

$$\Upsilon^{d}(i,1) \cap n_{i,1} > 0 \text{ at } t] = 0$$
(5.5.3)

Similarly, the interruption of flow equations for $M^u(i+1,1)$ are

$$p_1^u(i+1,1) = p_{i+1}$$

$$p_2^u(i+1,1) = \frac{1}{W^u(i+1,1)} \times \left[X_2^d(i+1,1) \left(R^u(i,1) + \left(1 - R^u(i,1)\right) q^d(i,1) \right) - X_2^d(i+1,1) \left(1 - R^u(i,1)\right) r_{i+1} \right]$$

$$p_3^u(i+1,1) = 0$$
(5.5.4)

As a summary, Figure 5.5.1 shows all the possible transitions of $M^d(i, 1)$ and $M^u(i+1, 1)$.

5.6 Idleness Failure of Type 1

Now we need to derive expressions for the idleness failure of the building block. The idleness failure is the transition from an up state to a down state while it is idle. Note that there are three down states. However, the idleness failure occurs only to the Mode 1 down state, $\Delta_1^d(i, 1)$. Suppose that $M^d(i, 1)$ is up and starved. That is, M_{i+1} is up and $B_{i,1}$ is empty. The assumption in (3.2.3) implies that M_{i+1} cannot get blocked at this moment. Therefore, $M^d(i, 1)$ cannot go to the second or third down state when it is starved. As a result,

$$q_2^d(i,1) = 0$$

 $q_3^d(i,1) = 0$

The probability $q_1^d(i, 1)$ represents the probability that $M^d(i, 1)$ fails to the first down state at t + 1 given that it was up and starved at t. That is,

$$q_1^d(i,1) = Pr[\Delta_1^d(i,1) \text{ at } t+1 | \Upsilon^d(i,1) \cap n_{i,1} = 0 \text{ at } t]$$

= $Pr[\alpha_{i+1} = 0 \text{ at } t+1]$
 $\alpha(t) = 1 \cap n_{i,1}(t) = 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t]$

Let us consider the probability that M_{i+1} fails while it is either blocked or starved for Type 1. This quantity is related to the parameters of Type 2 building blocks since M_{i+1} fails while it is working Type 2. When M_{i+1} is either starved or blocked for a Type 1 part, and it is also starved for a Type 2 part, $M^d(i, 2)$ can be one of the following three conditions:

- $M^d(i,2)$ is up and working $(\Upsilon^d(i,2) \cap n_{i,2} > 0)$.
- $M^{d}(i, 2)$ is starved $(\Upsilon^{d}(i, 2) \cap n_{i,2} = 0).$
- $M^{d}(i,2)$ is down in Mode 2 $(\Delta_{2}^{d}(i,2) \cap n_{i,2} > 0).$

However, among these events, M_{i+1} can fail only when it is working on a Type 2 part. Therefore, $q_1^d(i, 1)$ is a fraction of the failure probability, p_{i+1} and we state the idleness failure such that

$$q_1^d(i,1) = p_{i+1} \frac{W_d(i,2)}{W_d(i,2) + X_2(i,2) + P_s(i,2)}$$
(5.6.1)

Similarly,

$$q_1^u(i+1,1) = p_{i+1} \frac{W^u(i+1,2)}{W^u(i+1) + P_b(i+1,1) + X_2(i+1,1)}$$
(5.6.2)

5.7 Boundary Conditions

Let us consider the last building block for Type 1, L(K, 1). The only reason that the observer in the buffer $B_{K,1}$ does not find a part moving out of the buffer is that the machine, $M_{K+1,1}$, is down. Since $M_{K+1,1}$ is the last machine and only machine downstream from the buffer, no remote failure is possible for machine $M^d(K, 1)$. Therefore the downstream machine has only one down state. Also $M_{K+1,1}$ processes only Type 1 parts, therefore no idleness failure occurs in $M^d(K, 1)$. The parameters for $M^d(K, 1)$ are

$$r^{d}(K,1) = r_{K+1,1}$$

$$p^{d}(K,1) = p_{K+1,1}$$

$$q^{d}(K,1) = 0$$
(5.7.1)

Similarly, the parameters for $M^u(1,1)$ are

$$r^{d}(K,0) = r_{0,1}$$

 $p^{d}(0,1) = p_{0,1}$ (5.7.2)
 $q^{d}(0,1) = 0$

5.8 Summary of the Decomposition Equations for Type 1

For $i = 0 \dots K$

5.8.1 Resumption of flow

$$r^d(i,1) = r_{i+1}$$
 (5.8.1)

$$r_2^d(i,1) = (1 - q^u(i+1,1))R^d(i+1,1))$$
 (5.8.2)

$$r_3^d(i,1) = r_{i+1}R^d(i+1,1)$$
 (5.8.3)

$$r_1^u(i+1,1) = r_{i+1} (5.8.4)$$

$$r_2^u(i+1,1) = (1-q^d(i,1))R^u(i,1)$$
 (5.8.5)

$$r_3^u(i+1,1) = R^u(i,1)r_{i+1} (5.8.6)$$

5.8.2 Failure mode change

$$z_{1,2}^d(i,1) = 0 (5.8.7)$$

$$z_{1,3}^d(i,1) = 0 (5.8.8)$$

$$z_{2,1}^d(i,1) = q_1^u(i+1,1)R^d(i+1,1)$$
 (5.8.9)

$$z_{2,3}^{d}(i,1) = q_{1}^{u}(i+1,1)((1-R^{d}(i+1,1)))$$

$$z_{2,3}^{d}(i,1) = (1-r) P^{d}(i+1,1)$$
(5.8.10)
(5.8.11)

$$z_{3,1}^d(i,1) = \left(1 - r_{i+1}\right) R^d(i+1,1)$$
(5.8.11)

$$z_{3,2}^{d}(i,1) = r_{i+1} \left(1 - R^{d}(i+1,1) \right)$$
(5.8.12)

$$z_{1,2}^u(i+1,1) = 0 (5.8.13)$$

$$z_{1,3}^u(i+1,1) = 0 (5.8.14)$$

$$z_{2,1}^{u}(i+1,1) = q_{1}^{d}(i,1)R^{u}(i,1)$$

$$(5.8.15)$$

$$z_{2,1}^{u}(i+1,1) = (1 - D^{u}(i,1))z^{d}(i,1)$$

$$(5.8.16)$$

$$z_{2,3}^{u}(i+1,1) = (1-R^{u}(i,1))q_{1}^{d}(i,1)$$
 (5.8.16)

$$z_{2,1}^{u}(i+1,1) = q_{1}^{d}(i,1)R^{u}(i,1)$$
(5.8.17)

$$z_{3,1}^{u}(i+1,1) = R^{u}(i,1)\left(1-r_{i+1}\right)$$
(5.8.18)

$$z_{3,2}^{u}(i+1,1) = \left(1 - R^{u}(i,1)\right)r_{i+1}$$
(5.8.19)

5.8.3 Interruption of flow

$$p_1^d(i,1) = p_{i+1} \tag{5.8.20}$$

$$p_{2}^{d}(i,1) = \frac{1}{W^{d}(i,1)} \times (5.8.21)$$

$$\left[X_{2}^{d}(i,1) \left(R^{d}(i+1,1) + q^{u}(i+1,1) \left(1 - R^{d}(i+1,1) \right) \right) - X_{3}^{d}(i,1)r_{i+1}(1 - R^{d}(i+1,1)) \right]$$

$$p_3^d(i,1) = 0 (5.8.22)$$

$$p_1^u(i+1,1) = p_{i+1}$$
 (5.8.23)

$$p_{2}^{u}(i+1,1) = \frac{1}{W^{u}(i+1,1)} \times (5.8.24)$$

$$\begin{bmatrix} X_{2}^{d}(i+1,1) \left(R^{u}(i,1) + \left(1 - R^{u}(i,1)\right) q^{d}(i,1) \right) \\ -X_{2}^{d}(i+1,1) \left(1 - R^{u}(i,1)\right) r_{i+1} \end{bmatrix}$$

$$r_{2}^{u}(i+1,1) = 0 \qquad (5.8.25)$$

 $p_3^u(i+1,1) = 0$

(5.8.25)

5.8.4 Idleness failure

$$q_1^d(i,1) = p_{i+1} \frac{W_d(i,2)}{W_d(i,2) + X_2(i,2) + P_s(i,2)}$$
(5.8.26)

$$q_2^d(i,1) = 0 (5.8.27)$$

$$q^{u}(i+1,1) = p_{i+1} \frac{W^{-}(i+1,2)}{W^{u}(i+1,2) + P_{b}(i+1,2) + X_{2}(i+1,2)}$$
(5.8.29)

$$q_2^u(i+1,1) = 0 (5.8.30)$$

$$q_3^u(i+1,1) = 0 (5.8.31)$$

Chapter 6

Decomposition Analysis for Type 2

The existence of Type 2 parts is made apparent to Type 1 parts only through the existence of idleness failures. However, the existence of Type 1 parts has a major influence on the production of Type 2 parts, and therefore the derivation for the Type 2 part decomposition is much more complicated. In particular, we now have to account for the possibility that an observer in a Type 2 buffer will see flow in and out of his buffer cease because the machines switched from making Type 2 to Type 1 parts.

6.1 State definitions

A downstream machine $M^d(i, 2)$, i = 1, ..., k - 1, represents all the Type 2 flow behavior from downstream $B_{i,2}$. Note that the machine $M^d(i, 2)$ is up when M_{i+1} is up and is either blocked or starved for Type 1 but not blocked for Type 2. That is,

$$\Upsilon^{d}(i,2) = \{\alpha_{i+1} = 1 \cap (n_{i,2} = 0 \cup n_{i+1,1} = N_{i+1,1})$$

$$\cap n_{i+1,2} < N_{i+1,2}\}$$
(6.1.1)

The observer does not see a part moving out of the buffer $B_{i,2}$ when M_{i+1} is down; or M_{i+1} is up but processing Type 1; or M_{i+1} is up and it is either starved or blocked for Type 1, and it is blocked for Type 2. As we defined in Chapter 5, the first case is the local failure. The second case is a failure that cannot be observed in a Type 1 building block. We call this failure as the *type failure*. For our convenience, we combine the two failures — the local failure and the type failure – and create a new failure called the *local-type failure*. The down state caused by the local-type failure is stated such that,

$$\{\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cup (n_{i,1} > 0 \cup n_{i+1} < N_{i+1,1})\}$$
(6.1.2)

Note that the complement of the local-type down state is,

$$\sim \{\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cup (n_{i,1} > 0 \cup n_{i+1} < N_{i+1,1})\}\$$
$$= \{\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1})\}$$

With the local-type failure, we defined the following down states for $M^{d}(i, 2)$:

- $M^d(i,2)$ is local-type down and not blocked for Type 2.
- $M^d(i, 2)$ is up and blocked for Type 2.
- $M^d(i, 2)$ is local-type down and blocked for Type 2.

These down sates are:

$$\Delta_{1}^{d}(i,2) = \{\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\}$$

$$\cap n_{i+1,2} < N_{i+1,2}\}$$

$$\Delta_{2}^{d}(i,2) = \{\alpha_{i+1} = 1 \cap \{n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}\}$$

$$\cap n_{i+1,2} = N_{i+1,2}\}$$

$$\Delta_{3}^{d}(i,2) = \{\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\}$$

$$\cap n_{i+1,2} = N_{i+1,2}\}$$

$$(6.1.3)$$

It is observed that the down state definitions for Type 2 are similar to those of the Type 1 described in (5.1.1).

6.2 Equalities

For our convenience, we define the following building block probabilities:

$$W^{u}(i,2) = Pr[\Upsilon^{u}(i,2) \cap n_{i,2} < N_{i,2}]$$

$$W^{d}(i,2) = Pr[\Upsilon^{d}(i,2) \cap n_{i,2} > 0]$$

$$X^{u}_{m}(i,2) = Pr[\Delta^{u}_{m}(i,2) \cap n_{i,2} < N_{i,2}], \quad m = 1,2,3 \quad (6.2.1)$$

$$X^{d}_{n}(i,2) = Pr[\Delta^{d}_{n}(i,2) \cap n_{i,2} > 0], \quad n = 1,2,3$$

$$Pb(i,2) = Pr[\Upsilon^{u}(i,2) \cap n_{i,2} = N_{i,2}]$$

$$Ps(i,2) = Pr[\Upsilon^{d}(i,2) \cap n_{i,1} = 0]$$

$$Db(i,2) = Pr[\Delta_{1}(i,2) \cap n_{i,1} = N_{i,2}]$$

$$Ds(i,2) = Pr[\Delta_{1}(i,2) \cap n_{i,1} = 0]$$

$$i = 1...k - 1$$

In(6.1.1), the following equality can be stated.

$$\Upsilon^d(i,2) = \{ (\Delta_2^d(i,1) \cup \Delta_2^u(i+1,1)) \cap n_{i+1,2} < N_{i+1,2} \}$$

Then

$$W^{d}(i,1) = Pr[\Upsilon^{d}(i,2) \cap n_{i,2} < N_{i,2}]$$

= $Pr[(\Delta_{2}^{d}(i,1) \cup \Delta_{2}^{u}(i+1,1)) \cap n_{i+1,2} < N_{i+1,2} \cap n_{i,2} > 0]$
= $W^{u}(i+1,2)$

Next, $X_1^d(i, 1)$ is the probability that the downstream machine of L(i, 1) is down in mode 1 and not starved. From the definition (6.1.3) and (7.2.1), it is

$$\begin{aligned} X_1^d(i,2) &= Pr[\Delta_1^d(i,2) \cap n_{i,2} > 0] \\ &= Pr[\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1})) \\ &\cap (n_{i+1,2} < N_{i+1,2} \cap n_{i,2} > 0)] \\ &= Pr[\Delta^u(i+1,2) \cap n_{i+1,2} < N_{i+1,2}] \\ &= X_1^u(i+1,2) \end{aligned}$$

Therefore,

$$X_1^d(i,2) = X_1^u(i+1,2) \tag{6.2.2}$$

Next, $X_2^d(i, 2)$ is the probability that the downstream machine of L(i, 2) is down in mode 2 and not starved. From the definition (6.1.3), it is

$$X_{2}^{d}(i,2) = Pr[\{\alpha_{i+1} = 1 \cap \{n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}\} \\ \cap n_{i+1,2} = N_{i+1,2} \cap n_{i,2} > 0] \\ = Pr[\Upsilon^{u}(i+1,2) \cap n_{i+1,2} = N_{i+1,2}] \\ = P_{b}(i+1,2)$$

A similar equality can be derived for $X_2^u(i+1,1)$. Therefore,

$$X_2^d(i,2) = P_b(i+1,2) (6.2.3)$$

$$X_2^u(i+1,2) = P_s(i,2) (6.2.4)$$

Last,

$$\begin{aligned} X_3^d(i,1) &= Pr[\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\} \\ &\cap n_{i,2} > 0 \cap n_{i+1,2} = N_{i+1,2}] \\ &= Pr[\Delta_3^u(i+1,2) \cap n_{i+1,2} = N_{i+1,2}] \\ &= D_b(i+1,2) \end{aligned}$$

Again, $X_3^u(i+1,1)$ can be derived in the similar way. Therefore,

$$X_3^d(i,2) = D_b(i+1,2) (6.2.5)$$

$$X_3^u(i+1,2) = D_s(i,2) (6.2.6)$$

6.3 Resumption of Flow

Since there are three down states for $M^d(i, 2)$, three separate repair probabilities need to be derived. First, we define the following quantities for the derivations of the resumption of flow equations.

$$R^{d}(i,2) = \frac{\sum_{n=1}^{3} r_{n}^{d}(i,2) X_{n}^{d}(i,2)}{\sum_{n=1}^{3} X_{n}^{d}(i,2)}$$

likewise,

$$R^{u}(i,2) = \frac{\sum_{m=1}^{3} r_{m}^{u}(i,2) X_{m}^{u}(i,2)}{\sum_{m=1}^{3} X_{m}^{d}(i,2)}$$

The definition of the repair probability from the down Mode 1 is,

$$r_l^d(i,2) = Pr[\Upsilon^d(i,2) \text{ at } t+1|\Delta_l^d(i,1) \text{ at } t]$$

 $l = 1,2,3$

The transition probability from $\Delta_1^d(i,2)$ to $\Upsilon^d(i,2)$ denoted by $r_1^d(i,2)$ is derived such that

$$r_1^d(i,2) = Pr[\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} < N_{i+1,2} \text{ at } t+1|$$
$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1})$$
$$\cap n_{i+1,2} < N_{i+1,2} \text{ at } t]$$

Note that in this conditional probability, the event of $n_{i+1,2} < N_{i+1,2}$ is not affected during the transition. That is,

$$Pr[n_{i+1,2} = N_{i+1,2} \text{ at } t+1 | \alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1})$$
$$\cap n_{i+1,2} < N_{i+1,2} \text{ at } t] = 0$$

Therefore it can be stated that

$$r_1^d(i,2) = Pr[\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \text{ at } t+1$$
$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \text{ at } t]$$

In other words, a buffer $n_{i+1,2}$ cannot get full if M_{i+1} was either down or working for a Type 1 part in the previous time step. Also we assume that a machine is not likely be starved and blocked simultaneously, and therefore it can be stated that

$$\begin{aligned} r_1^d(i,2) &= Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t+1| \\ \alpha_{i+1} &= 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \text{ at } t] \\ &+ Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t+1| \\ \alpha_{i+1} &= 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \text{ at } t] \end{aligned}$$

The term $\{\alpha_{i+1} = 0\}$ is decomposed into the following mutually exclusive terms:

- { $\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}$ }
- $\{\alpha_{i+1} = 0 \cap n_{i,1} = 0\}$
- { $\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1}$ }

Using the law of probability,

$$\begin{split} r_1^d(i,2) = \\ 1/\bigg(\Pr[\{\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}\} \cup \{\alpha_{i+1} = 0 \cap n_{i,1} = 0\} \\ \cup\{\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1}\} \cup \{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}\}\bigg) \times \\ \bigg(\Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i,1} = 0 \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i,1} = 0] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i,1} = 0 \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i,1} = 0] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1|\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t] \times \\ \Pr[\alpha_{i+1} = 0 \cap n_{i,1} = 0] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1|\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t] \\ \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ + \Pr[\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1}] \\ \\ (n_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1}] \\ (n_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}] \\ (n_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1}] \\ (n_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1}] \\ (n_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1}] \\ (n_{i+1} = 1 \cap n_{i,1} > 0$$

Based on the assumption in (3.2.3) and the priority rules, the quantity of the following conditional probabilities are equal to zero.

- $Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1 | \alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t]$
- $Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1 | \alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t]$
- $Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1 | \alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t]$
- $Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t + 1 | \alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t]$

The rest of the terms are evaluated such that

- $Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t + 1 | \alpha_{i+1} = 0 \cap n_{i,1} = 0 \text{ at } t] = r_{i+1}(1 R^u(i, 1))$ Note that this is a conditional probability that a machine M_{i+1} being down and starved for Type 1 is fixed but it is still starved for Type 1. Therefore, this quantity is the product of r_{i+1} and the probability that any failure causing the starvation is not fixed, which is $1 - R^u(i, 1)$.
- $Pr[\alpha_{i+1} = 1 \cap n_{i,1} = 0 \text{ at } t+1 | \alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t] = p_2^u(i,1)$ This is the quantity that $M_{i+1,1}$ gets starved for Type 1. We define this to be $p_2^u(i,1)$ and derivation of this quantity is introduce in Section 6.5
- $Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t+1 | \alpha_{i+1} = 0 \cap n_{i,1} = 0 \text{ at } t] = r_{i+1}(1 R^d(i+1,1))$

This is a case that machine M_{i+1} being down and blocked for Type 1 is fixed but it is still blocked for Type 1. Therefore, this quantity is the product of r_{i+1} and the probability that any failure causing the blockage is not fixed, which is $1 - R^d(i+1, 1)$.

• $Pr[\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1} \text{ at } t+1 | \alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1} \text{ at } t] = p_2^d(i,1)$

This is the quantity that $M_{i+1,1}$ gets blocked for Type 1, which is $p_2^d(i,1)$

From the notation described in Chapter 5, the first repair probability is therefore,

$$r_{1}^{d}(i,2) = 1/\left(W^{d}(i,1) + D_{s}(i,1) + D_{b}(i+1,1) + X_{1}^{d}(i,1)\right) \times \left(W^{d}(i,1)(p_{2}^{u}(i+1,1) + p_{2}^{d}(i,1)) + D_{s}(i,1)(1 - R^{u}(i,1))r_{i+1} + D_{b}(i+1,1)r_{i+1}(1 - R^{d}(i+1,1))\right)$$

$$(6.3.1)$$

The next down repair probability, $r^d(i,2)$ is expressed as

$$r_{2}^{d}(i,2) = Pr[\Upsilon^{d}(i,2) \text{ at } t+1|\Delta_{2}^{d}(i,2) \text{ at } t]$$

$$= Pr[\alpha_{i+1,1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} < N_{i+1,2} \text{ at } t+1|$$

$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$
(6.3.2)

This transition indicates that M_{i+1} , which was once blocked for Type 2, starts working on Type 2. Machine M_{i+1} was in the state that it was up, and either blocked or starved for Type 1, and blocked for Type 2. This transition probability is approximated such that,

$$r_{2}^{d}(i,2) \approx Pr[\alpha_{i+1} = 1 \cap (n_{i+1,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \text{ at } t+1|$$
(6.3.3)
$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$
$$\times Pr[n_{i+1,2} < N_{i+2,2} \text{ at } t+1|$$
$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

In the first conditional probability in (6.3.3) indicates that M_{i+1} remains up during the transition. That is, M_{i+1} should not get failed while it is blocked for Type 2. This transient probability is equivalent to

$$Pr[\alpha_{i+1} = 1 \cap (n_{i+1,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \text{ at } t+1|$$
$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$
$$= (1 - q^u(i+1,2))$$

The next conditional probability in (6.3.3) is the transition that M_{i+1} gets not blocked for Type 2. Also note that the state of machine M_{i+1} and the Type 1 buffers, $n_{i,1}$ and n_{i+1} , are independent of the state of the Type 2 buffer $n_{i+1,2}$. Therefore, the transition occurs when $M^d(i+1)$ gets repaired:

$$Pr[n_{i+1,2} < N_{i+2,2} \text{ at } t+1]$$

$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

$$= Pr[n_{i+1,2} < N_{i+2,2} \text{ at } t+1|n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

$$= R^d(i+1,2)$$

Then $r_2^d(i,2)$ is approximated such that

$$r_2^d(i,2) = (1 - q^u(i+1,2))R^d(i+1,2)$$
(6.3.4)

The third repair probability , $r_3^d(i, 1)$, is the transition probability from $\Delta_3^d(i, 1)$ to $\Upsilon^d(i, 1)$. That is,

$$r_{3}^{d}(i,2) = Pr[\Upsilon^{d}(i,2) \text{ at } t+1|\Delta_{3}^{d}(i,2) \text{ at } t]$$

$$= Pr[\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} < N_{i+1,2} \text{ at } t+1|$$

$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1})$$

$$\cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$
(6.3.5)

This transition probability is approximated such that

$$r^{d}(i,2) \approx Pr[\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \text{ at } t+1|$$
(6.3.6)

$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1})$$

$$\cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

$$\times Pr[n_{i+1,2} = N_{i+1,2}) \text{ at } t+1|$$

$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1})$$

$$\cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

Note that in (6.3.6), the machine state and buffer states for Type 1, $n_{i,1}$ and $n_{i+1,1}$, are independent of the the Type 2 buffer state $n_{i+1,2}$ and therefore,

$$r^{d}(i,2) \approx Pr[\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \text{ at } t+1|$$
(6.3.7)
$$\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \text{ at } t]$$
$$\times Pr[n_{i+1,2} < N_{i+1,2}|n_{i+1,2} = N_{i+1,2}]$$

Observe that the first conditional probability is equal to $r_1^d(i, 2)$ and the second term is $R^d(i+1, 2)$;

$$r_3^d(i,2) = r_1^d(i,2)R^d(i+1,2)$$
(6.3.8)
As a summary, the resumption of flow equations for $M^d(i, 1)$ are:

$$r_{1}^{d}(i,2) = 1/\left(W^{d}(i,1) + D_{s}(i,1) + D_{b}(i+1,1) + X_{1}^{d}(i,1)\right) \times \left(W^{d}(i,1)(p_{2}^{u}(i+1,1) + p_{2}^{d}(i,1)) + D_{s}(i,1)(1 - R^{u}(i,1))r_{i+1} + D_{b}(i+1,1)r_{i+1}(1 - R^{d}(i+1,1))\right)$$

$$r_{2}^{d}(i,2) = (1 - q^{u}(i+1,2))R^{d}(i+1,2)$$

$$(6.3.10)$$

$$r_3^d(i,2) = r_1^d(i,2)R^d(i+1,2)$$
 (6.3.11)

Similarly, the resumption of flow equations for $M^{u}(i, 1)$ are:

$$r_{1}^{u}(i,2) = 1/\left(W^{d}(i,1) + D_{s}(i,1) + D_{b}(i+1,1) + X_{1}^{d}(i,1)\right) \times \left(W^{d}(i,1)(p_{2}^{u}(i+1,1) + p_{2}^{d}(i,1)) + D_{s}(i,1)(1 - R^{u}(i,1))r_{i+1} + D_{s}(i,1)(1 - R^{u}(i,1))r_{i+1} + D_{b}(i+1,1)r_{i+1}(1 - R^{d}(i+1,1))\right)$$

$$r_{2}^{u}(i,2) = (1 - q^{u}(i+1,2))R^{d}(i+1,2)$$

$$r_{3}^{u}(i,2) = r_{1}^{d}(i,2)R^{d}(i+1,2)$$

$$(6.3.14)$$

6.4 Failure mode change

Failure mode changes are transitions between down states. First, the transition from $\Delta_1^d(i, 2)$ to other down states is considered. This down state represents the state that either M_{i+1} is down or M_{i+1} is working on Type 1. From the observer in $B_{i,2}$, any state change further downstream of M_{i+1} , except the resumption of flow, will not change the state of $M^d(i, 2)$. Therefore, there in no transition from $\Delta_1^d(i, 1)$ to any

other down state. Therefore,

$$z_{1,2}^d(i,2) = 0 (6.4.1)$$

$$z_{1,2}^d(i,2) = 0 (6.4.2)$$

Next, once $M^d(i, 2)$ is in the down state $\Delta_2^d(i, 2) - M_{i+1}$ is up and is able to work for Type 2 due to starvation or blockage of Type 1 but idle due to blockage for Type 2 — it can make a transition into two other down states. First, let us consider the transition to $\Delta_3^d(i, 2)$

$$z_{2,3}^{d}(i,1) = Pr[\Delta_{3}^{d}(i,2) \text{ at } t+1|\Delta_{2}^{d}(i,2) \text{ at } t]$$

$$= Pr[\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1})$$

$$\cap n_{i+1,2} = N_{i+1,2} \text{ at } t+1|$$

$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$
(6.4.3)

Then

$$\begin{aligned} z_{2,3}^d(i,1) &= & Pr[\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \text{ at } t+1| \\ &\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t] \\ &\times Pr[n_{i+1,2} = N_{i+1,2} \text{ at } t+1| \\ &\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t] \end{aligned}$$

Since the state $n_{i+1,2} = N_{i+1,2}$ and the state of $M_{i+1}, n_{i,1}$, and $n_{i+1,1}$ are independent, it can be approximated as

$$z_{2,3}^d(i,1) \approx Pr[\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \text{ at } t+1|$$

$$\alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

$$\times Pr[n_{i+1,2} = N_{i+1,2} \text{ at } t+1|n_{i+1,2} = N_{i+1,2} \text{ at } t]$$

The first term is $q^u(i+1,2)$ while the second term is $(1 - R^d(i+1,2))$. Therefore,

$$z_{2,3}^d = q^u(i+1,2)(1 - R^d(i+1,2))$$
(6.4.4)

The next failure mode change in the second down state, $z_{3,2}^d(i,2)$ is derived with the same procedure.

$$\begin{aligned} z_{2,1}^{d}(i,1) &= Pr[\Delta_{1}^{d}(i,2) \text{ at } t+1|\Delta_{2}^{d}(i,2) \text{ at } t] \\ &= Pr[\alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1}) \\ &\cap n_{i+1,2}< N_{i+1,2} \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= Pr[\alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1}) \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &\times Pr[n_{i+1,2}< N_{i+1,2} \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &\approx Pr[\alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1}) \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1}) \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1}) \text{ at } t+1| \\ &\alpha_{i+1}=1\cap(n_{i,1}=0\cup n_{i+1,1}=N_{i+1,1}) \text{ at } t] \\ &\times Pr[n_{i+1,2}< N_{i+1,2} \text{ at } t+1|n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= q^{u}(i+1,2)R^{d}(i+1,2) \end{aligned}$$

Likewise, the failure mode changes from the last down state $\Delta^d_3(i,2)$ are

$$\begin{aligned} z_{3,2}^{d}(i,1) &= Pr[\Delta_{2}^{d}(i,2) \text{ at } t+1|\Delta_{3}^{d}(i,2) \text{ at } t] \end{aligned} \tag{6.4.6} \\ &= Pr[\alpha_{i+1}=1 \cap (n_{i,1}=0 \cup n_{i+1,1}=N_{i+1,1}) \cap n_{i+1,2}=N_{i+1,2} \text{ at } t+1| \\ \alpha_{i+1}=0 \cup (\alpha_{i+1}=1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= Pr[\alpha_{i+1}=1 \cap (n_{i,1}=0 \cup n_{i+1,1}=N_{i+1,1}) \text{ at } t+1| \\ \alpha_{i+1}=0 \cup (\alpha_{i+1}=1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &\times Pr[n_{i+1,2}=N_{i+1,2} \text{ at } t+1| \\ \alpha_{i+1}=0 \cup (\alpha_{i+1}=1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &\approx Pr[\alpha_{i+1}=1 \cap (n_{i,1}=0 \cup n_{i+1,1}=N_{i+1,1}) \text{ at } t+1| \\ \alpha_{i+1}=0 \cup (\alpha_{i+1}=1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}) \text{ at } t] \\ &\times Pr[n_{i+1,2}=N_{i+1,2} \text{ at } t+1|n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= r_{1}^{d}(i,2)(1-R^{d}(i+1,2)) \end{aligned}$$

and

$$\begin{aligned} z_{3,1}^d(i,1) &= Pr[\Delta_1^d(i,2) \text{ at } t+1|\Delta_3^d(i,2) \text{ at } t] \end{aligned} (6.4.7) \\ &= Pr[\alpha_{i+1}=0\cup(n_{i,1}=1\cup n_{i,1}>0\cap n_{i+1,1}< N_{i+1,1})\cap n_{i+1,2}< N_{i+1,2} \text{ at } t+1|\\ \alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= Pr[\alpha_{i+1}=0\cup(n_{i,1}=1\cup n_{i,1}>0\cap n_{i+1,1}< N_{i+1,1}) \text{ at } t|\\ \alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &\times Pr[n_{i+1,2}< N_{i+1,2} \text{ at } t+1|\\ \alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= Pr[\alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cap n_{i+1,1}< N_{i+1,1}) \text{ at } t+1|\\ \alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1}) \text{ at } t+1|\\ \alpha_{i+1}=0\cup(\alpha_{i+1}=1\cap n_{i,1}>0\cup n_{i+1,1}< N_{i+1,1}) \text{ at } t] \\ &\times Pr[n_{i+1,1}< N_{i+1,1}) \text{ at } t+1|n_{i+1,1}< N_{i+1,1})\cap n_{i+1,2}=N_{i+1,2} \text{ at } t] \\ &= (1-r_1^d(i,2))R^d(i+1,2) \end{aligned}$$

With similar approaches, the failure mode changes for $M^{u}(i, 1)$ are:

$$z_{1,2}^u(i,2) = 0 (6.4.8)$$

$$z_{1,3}^u(i,2) = 0 (6.4.9)$$

$$z_{2,1}^{u}(i,2) = q^{d}(i,2)R^{u}(i,2)$$
(6.4.10)

$$z_{2,3}^{u}(i,2) = (1 - R^{u}(i,2))q^{d}(i,2)$$
(6.4.11)

$$z_{3,1}^{u}(i,2) = R^{u}(i,2)(1-r^{u}(i,2))$$
(6.4.12)

$$z_{3,2}^{u}(i,2) = (1 - R^{u}(i,2))r^{u}(i,2)$$
(6.4.13)

6.5 Interruption of Flow

Since there are three down states, three interruption of flow probabilities are needed to be derived. First, the probability $p_1^d(i, 2)$ is the transition from $\Upsilon^d(i, 2)$ to $\Delta_1^d(i, 2)$, which represents the failure of $M^d(i, 2)$ when $B_{i,2}$ is not empty. This transition probability is

$$p_{1}^{d}(i,2) = Pr[\Delta_{1}^{d}(i,2) \text{ at } t+1|\Upsilon^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t]$$

$$= Pr[\alpha_{i+1} = 0 \cup (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1})$$

$$\cap n_{i+1,2} < N_{i+1,2} \text{ at } t+1|$$

$$\alpha_{i+1} = 1 \cap (n_{i,2} = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i+1,2} < N_{i+1,2}$$

$$\cap n_{i,2} > 0 \text{ at } t]$$

$$(6.5.1)$$

This is equivalent to

$$p_{1}^{d}(i,2) = Pr[(\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \cup (6.5.2)$$

$$(\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1}) \cup (\alpha_{i+1} = 0 \cap n_{i,1} = 0) \cup (\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1}) \text{ at } t+1|$$

$$((\alpha_{i+1} = 1 \cap n_{i,1} = 0) \cup (\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1}))$$

$$\cap (\alpha_{i+1} = 1 \cap n_{i,2} > 0 \cup n_{i+1,2} < N_{i+1,2}) \text{ at } t]$$

Let us define the following events:

$$V = (\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1})$$

$$D_0 = (\alpha_{i+1} = 0 \cap n_{i,1} > 0 \cap n_{i+1,1} < N_{i+1,1})$$

$$B = (\alpha_{i+1} = 1 \cap n_{i+1,1} = N_{i+1,1})$$

$$S = (\alpha_{i+1} = 1 \cap n_{i,1} = 0)$$

$$D_b = (\alpha_{i+1} = 0 \cap n_{i+1,1} = N_{i+1,1})$$

$$D_s = (\alpha_{i+1} = 0 \cap n_{i,1} = 0)$$

$$W_2 = (\alpha_{i+1} = 1 \cap n_{i,2} > 0 \cap n_{i+1,2} < N_{i+1,2})$$

The event B is the set of states in which M_{i+1} is blocked for a Type 1 part, while S is the set of states in which M_{i+1} is starved for a Type 1 part. These two sets are disjoint events because of the approximation that the probability of a machine being blocked and starved is zero as we stated in (3.2.3). Then

$$p_1^d(i,2) = Pr[V \cup D_0 \cup D_s \cup D_b | (S \cap W_2) \cup (B \cap W_2)]$$
(6.5.4)

By expanding this, we have

$$p_{1}^{d}(i,2) =$$

$$\frac{Pr[S \cap W_{2}]}{Pr[B \cap W_{2}] + Pr[S \cap W_{2}]} \times$$

$$\left(Pr[V|S \cap W_{2}] + Pr[D_{0}|S \cap W_{2}] + Pr[D_{s}|S \cap W_{2}] + Pr[D_{b}|S \cap W_{2}]\right) + \frac{Pr[B \cap W_{2}]}{Pr[B \cap W_{2}] + Pr[S \cap W_{2}]} \times$$

$$\left(Pr[V|B \cap W_{2}] + Pr[D_{0}|S \cap W_{2}] + Pr[D_{s}|B \cap W_{2}] + Pr[D_{b}|B \cap W_{2}]\right)$$
(6.5.5)

The fractions in the equations can be approximated as follows

$$\frac{Pr[S \cap W_2]}{Pr[B \cap W_2] + Pr[S \cap W_2]} \approx \frac{Pr[S]}{Pr[B] + Pr[S]}$$

$$\frac{Pr[B \cap W_2]}{Pr[B \cap W_2] + Pr[S \cap W_2]} \approx \frac{Pr[B]}{Pr[B] + Pr[S]}$$

$$(6.5.6)$$

It remains for us to calculate the individual probabilities. We already have defined in (5.2.4) that M being up and starved is the same event as $M^u(i+1,1)$ being down at mode $\Delta_2^u(i+1,1)$. Also it is stated that M being up and blocked is equivalent to $M^d(i+1,1)$ being down at mode $\Delta_2^d(i+1,1)$. Therefore,

$$Pr[S] = X_2^u(i+1,1)$$
 (6.5.7)
 $Pr[B] = X_2^d(i,1)$

Now, we need to calculate conditional probabilities in (6.5.5). First, note that the following conditional probabilities are zero:

$$Pr[D_s|B \cap W_2] = 0$$

$$Pr[D_b|S \cap W_2] = 0$$
(6.5.8)

This is due to our approximation in (3.2.3).

Next, $Pr[V|S \cap W_2]$ is the probability that M_{i+1} is working on a Type 1 part at time t+1, given that it was up and starved for Type 1, but not starved nor blocked for Type 2 at time t. This probability is the same as the probability that $M^u(i+1, 1)$ is up and not blocked in time t+1 give that it was down at mode $\Delta_2^u(i, 1)$. Therefore, this conditional probability is the transition probability from $\Delta_2^u(i+1,1)$ to $\Upsilon^u(i+1,1)$, which is given by.

$$Pr[V|S \cap W_2] = Pr[\Upsilon^u(i+1,1) \cap n_{i+1,1} < N_{i,1} \text{ at } t+1] \qquad (6.5.9)$$
$$\Delta_2^u(i+1,1) \text{ at } t]$$
$$= R^u(i,1)(1-q^d(i,1))$$

In a similar manner, $P[V|B \cap W_2]$, the conditional probability that M_{i+1} is working on Type 1 part at t+1, given that M_{i+1} was blocked, but was not blocked nor starved for Type 2 at t, can be written with building block parameters such that

$$Pr[V|B \cap W_2] = Pr[\Upsilon^d(i,1) \cap n_{i,1} > 0 \text{ at } t+1|$$

$$\Delta_2^d(i,1) \text{ at } t]$$

$$= (1 - q^u(i+1,1))R^d(i+1,1)$$
(6.5.10)

This is because when $M^d(i + 1, 1)$ is repaired and $M^u(i + 1, 1)$ does not go into the idleness failure — fail while it is blocked — at the end of time step t, M_{i+1} will be no longer blocked and process a Type 1 part at time t + 1.//

Next $P[D_s|S \cap W_2]$ is the probability that M_{i+1} goes down while it is working on Type 2. This is equivalent that $M^u(i+1,1)$ is initially at the down mode $\Delta_2^u(i+1,1)$ and $n_{i+1,1} < N_{i+1,1}$, but the down mode change takes place from $\Delta_2^u(i+1,1)$ to $\Delta_3^u(i+1,1)$ at the end of time step. Therefore,

$$Pr[D_s|S \cap W_2] = Pr[\Delta_3^u(i+1,1) \cap n_{i+1,1} < N_{i+1,1} \text{ at } t+1| \quad (6.5.11)$$
$$\Delta_2^u(i+1,1) \text{ at } t]$$
$$= (1 - R^u(i,1))q^d(i,1)$$

For the similar reason, $P[D_b|B \cap W_2]$ is

$$Pr[D_b|B \cap W_2] = Pr[\Delta_3^d(i,1) \cap n_{i,1} > 0 \text{ at } t + 1|\Delta_2^d(i,1) \text{ at } t] \quad (6.5.12)$$
$$= q^u(i+1,1)(1-R^d(i+1,1))$$

Next $P[D_0|S \cap W_2]$ is the probability that M_{i+1} gets not starved for Type 1 and at the same time it goes down, while it is working on Type 2. This is equivalent that $M^u(i+1,1)$ is initially at the down mode $\Delta_2^u(i+1,1)$ and $n_{i+1,1} < N_{i+1,1}$, but the down mode change takes place from $\Delta_2^u(i+1,1)$ to $\Delta_1^u(i+1,1)$ at the end of time step. Therefore,

$$Pr[D_0|S \cap W_2] = Pr[\Delta_1^u(i+1,1) \cap n_{i+1,1} < N_{i+1,1} \text{ at } t+1| \quad (6.5.13)$$
$$\Delta_2^u(i+1,1) \text{ at } t]$$
$$= R^u(i,1)q^d(i,1)$$

For the similar reason, $P[D_0|B \cap W_2]$ is

$$Pr[D_0|B \cap W_2] = Pr[\Delta_1^d(i,1) \cap n_{i,1} > 0 \text{ at } t + 1|\Delta_2^d(i,1) \text{ at } t] \quad (6.5.14)$$
$$= q^u(i+1,1)R^d(i+1,1)$$

Putting everything together, we have

$$p_{1}^{d}(i,2) = \frac{1}{P_{s}(i,1) + P_{b}(i+1,1)} \times (6.5.15) \\ \left(P_{s}(i,1) \left(R^{u}(i,1)q^{d}(i,1) + R^{u}(i,1)(1-q^{d}(i,1)) + R^{u}(i,1)q^{d}(i,1) \right) \\ + P_{b}(i+1,1) \left(q^{u}(i+1,1)(1-R^{d}(i+1,1)) + (1-q^{u}(i+1,1))R^{d}(i+1,1) \\ + q^{u}(i+1,1)R^{d}(i+1,1) \right) \right)$$

Next, we derive the interruption of flow equation for $p_2^d(i, 2)$, the transition probability from the up state to the state in which $M^d(i+1, 2)$ is down and $B_{i+1,2}$ is full. This transition probability is,

$$p_2^d(i,2) = Pr[\Delta_2^d(i,2) \text{ at } t+1 \mid \Upsilon^d(i,2) \cap n_{i,1} > 0 \text{ at } t]$$

We first start with the derivation of this equation by applying the fact that the probability of going out of a state is equal to the probability of going into that state.

$$\begin{aligned} X_2^d(i,2) & \left((1-q^u(i+1,2))R^d(i+1,2) + q^u(i+1,2)R^d(i+1,2) \right. \\ & \left. + q^u(i+1,2)(1-R^d(i+1,2)) \right) \\ & \left. = W^d(i,2)p_2^d(i,2) + X_3^d(i+1,2)r_1^d(i,2)(1-R^d(i+1,2)) \right) \end{aligned}$$

If we simplify this equation, then

$$X_2^d(i,2) \left(R^d(i+1,2) + q^u(i+1,2)(1-R^d(i+1,2)) \right)$$

= $W^d(i,2) p_2^d(i,2) + X_3^d(i,2) r_1^d(i,2)(1-R^d(i+1,2))$

That is,

$$p_{2}^{d}(i,2) = \frac{1}{W^{d}(i,2)} \times$$

$$\left(X_{2}^{d}(i,2) \left(R^{d}(i+1,2) + q^{u}(i+1,2)(1-R^{d}(i+1,2)) \right) - X_{3}^{d}(i,2)r_{1}^{d}(i,2)(1-R^{d}(i+1,2)) \right)$$

$$(6.5.16)$$

As a next step, we derive the interruption of flow equation for $p_3^d(i, 2)$, the transition probability from the up state to the state in which $M^d(i + 1, 2)$ is down and $B_{i+1,1}$ is full and at the same time M_{i+1} gets failed or switches to Type 1 production. Note that when M_{i+1} is up and not blocked for Type 2, it is impossible for M_{i+1} to get blocked for Type 2 when it either gets failed or switched to Type 1. Therefore,

$$p_{3}^{d}(i,2) = Pr[\Delta_{3}^{d}(i,2) \text{ at } t+1 |\Upsilon^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t]$$
(6.5.17)
= 0

Similarly, the interruption of flow equations for $M^u(i,2)$ are:

$$p_{1}^{d}(i,2) = \frac{1}{P_{s}(i,1) + P_{b}(i+1,1)} \times (6.5.18) \\ \left(P_{s}(i,1) \left(R^{u}(i,1)q^{d}(i,1) + R^{u}(i,1)(1-q^{d}(i,1)) + R^{u}(i,1)q^{d}(i,1) \right) \\ + P_{b}(i+1,1) \left(q^{u}(i+1,1)(1-R^{d}(i+1,1)) + (1-q^{u}(i+1,1))R^{d}(i+1,1) \\ + q^{u}(i+1,1)R^{d}(i+1,1) \right) \right)$$

$$p_{2}^{d}(i,2) = \frac{1}{W^{d}(i,2)} \times \left(X_{2}^{d}(i,2) \left(R^{d}(i+1,2) + q^{u}(i+1,2)(1-R^{d}(i+1,2)) \right) - X_{3}^{d}(i,2)r_{1}^{d}(i,2)(1-R^{d}(i+1,2)) \right)$$

$$p_{3}^{d}(i,2) = Pr[\Delta_{3}^{d}(i,2) \text{ at } t+1 |\Upsilon^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t]$$

$$= 0$$

6.6 Idleness failure

Now we need to derive expressions for the idleness failure of the building block. The idleness failure is the transition from an up state to a down state while it is idle. The probability $q_1^d(i, 2)$ represents the probability that $M^d(i, 2)$ gets failed to $\Delta_1^d(i, 2)$ at t+1 given that it was up and starved at t. That is,

$$q_{1}^{d}(i,2) = Pr[\Delta_{1}^{d}(i,1) \text{ at } t+1 |\Upsilon^{d}(i,1) \cap n_{i,1} = 0 \text{ at } t]$$

$$= Pr[\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\}|$$

$$\alpha_{i+1} = 1 \cap (n_{i,1}(t) = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i,2} = 0]$$
(6.6.1)

That is, the idleness failure occurs when M_i gets failed while it is working on Type 3, or M_i switches a process to Type 1. Then (6.6.1) can be rewritten as

$$q_{1}^{d}(i,2) = Pr[\alpha_{i+1} = 0 \text{ at } t+1]$$

$$\alpha_{i+1} = 1 \cap (n_{i,1}(t) = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i,2} = 0 \text{ at } t]$$

$$+Pr[\{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\} \text{ at } t+1]$$

$$\alpha_{i+1} = 1 \cap (n_{i,1}(t) = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i,2} = 0 \text{ at } t]$$
(6.6.2)

The first term represents the case that M_{i+1} gets failed while it is processing a Type 3 part. This quantity is related with parameters for Type 3 since M_{i+1} is allowed to fail while it is working Type 3. Note that the probability that M_{i+1} is starved or blocked for Type 1 as well as it is starved for Type 2 is expressed with the parameters for a Type 3 building block such that $W^d(i,3) + P_s(i,2) + X_2^d(i,2)$. That is, the observer in L(i,3) may find that $M^d(i,3)$ is either working or starved or down in Mode 2. However, between these events, M_{i+1} can get failed only when it is working on Type 3. Therefore, the first term in (6.6.2) is a fraction of the failure probability, p_{i+1} , and we approximate it such that

$$Pr[\alpha_{i+1} = 0 \text{ at } t+1]$$

$$\alpha_{i+1} = 1 \cap (n_{i,1}(t) = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i,2} = 0 \text{ at } t]$$

$$= p_{i+1} \frac{W_d(i,2)}{W_d(i,2) + X_2(i,2) + P_s(i,2)}$$
(6.6.3)

The second term in (6.6.2) is the case that M_{i+1} switches a process from Type 2 to Type 3. This quantity is similar with the failure probability, $p_1^d(i, 2)$. Note that $M^d(i, 1)$ get failure to Mode 1 when M_{i+1} is down or M_{i+1} switches a production from Type 2 to Type 1. Therefore, we can derived the quantity by eliminating the terms related to the local failure in the expression of $p_1^d(i, 2)$. Then the second term of (6.6.2) is

$$Pr[\{\alpha_{i+1} = 1 \cap n_{i,1} > 0 \cup n_{i+1,1} < N_{i+1,1}\} \text{ at } t+1]$$

$$\alpha_{i+1} = 1 \cap (n_{i,1}(t) = 0 \cup n_{i+1,1} = N_{i+1,1}) \cap n_{i,2} = 0 \text{ at } t]$$

$$= \frac{1}{P_s(i,1) + P_b(i+1,1)} \left(P_s(i,1)R^u(i,1) + P_b(i+1,1)R^d(i+1,1) \right)$$
(6.6.4)

Therefore,

$$q_{1}^{d}(i,2) = p_{i+1} \frac{W^{d}(i,2)}{W^{d}(i,2) + X_{2}^{d}(i,2) + P_{s}(i,2)}$$

$$+ \frac{1}{P_{s}(i,1) + P_{b}(i+1,1)} \left(P_{s}(i,1)R^{u}(i,1) + P_{b}(i+1,1)R^{d}(i+1,1) \right)$$
(6.6.5)

Similarly,

$$q_{1}^{u}(i,2) = p_{i} \frac{W^{u}(i,2)}{W^{u}(i,2) + X_{2}^{u}(i,2) + P_{b}(i,2)}$$

$$+ \frac{1}{P_{s}(i-1,1) + P_{b}(i,1)} \left(P_{s}(i-1,1)R^{u}(i-1,1) + P_{b}(i,1)R^{d}(i,1) \right)$$
(6.6.6)

The second failure mode $X_2^d(i,2)$ and the third failure mode $X_3^d(i,2)$ represent the case that M_{i+1} is blocked for Type 2. Note that $M^d(i,2)$ cannot get starved and blocked at the same time with our assumption in (3.2.3), the idleness failures to these failure modes do not occur. That is,

$$q_2^d(i,2) = 0$$
 (6.6.7)
 $q_3^d(i,3) = 0$

6.7 Boundary Conditions

Let us consider the last building block for Type 2, L(K, 2). The only reason that the observer in the buffer $B_{K,2}$ do not find a part moving out of the buffer is that the machine, $M_{K+1,2}$, is down. Since $M_{K+1,2}$ is the last machine and only machine downstream from the buffer, no remote failure is presented in the machine $M^d(K, 2)$. Therefore the downstream machine has only one down state and no failure mode changes are presented. Also $M_{K+1,2}$ processes only only Type 1 parts, therefore no idleness failure occurs in $M^d(K, 2)$. The parameters for $M^d(K, 2)$ are

$$r^{d}(K,2) = r_{K+1,2}$$

 $p^{d}(K,2) = p_{K+1,2}$ (6.7.1)
 $q^{d}(K,2) = 0$

Similarly, the parameters for the upstream are

$$r^{d}(K,0) = r_{0,2}$$

$$p^{d}(0,2) = p_{0,2}$$

$$q^{d}(0,2) = 0$$
(6.7.2)

6.8 Summary of the Decomposition Equations for Type 2

For $i = 0 \dots K$

6.8.1 Resumption of flow

$$\begin{aligned} r_1^d(i,2) &= 1/\left(W^d(i,1) + D_s(i,1) + D_b(i+1,1) + X_1^d(i,1)\right) \times \quad (6.8.1) \\ & \left(W^d(i,1)(p_2^u(i+1,1) + p_2^d(i,1)) + D_s(i,1)(1 - R^u(i,1))r_{i+1} \right. \\ & \left. + D_b(i+1,1)r_{i+1}(1 - R^d(i+1,1))\right) \\ r_2^d(i,2) &= (1 - q^u(i+1,2))R^d(i+1,2) \quad (6.8.2) \\ r_3^d(i,2) &= r_1^d(i,2)R^d(i+1,2) \quad (6.8.3) \end{aligned}$$

$$r_{1}^{u}(i+1,2) = 1/\left(W^{u}(i+1,1) + D_{s}(i,1) + D_{b}(i+1,1) + X_{1}^{u}(i+1,1)\right) \times \left(W^{u}(i+1,1)(p_{2}^{u}(i+1,1) + p_{2}^{d}(i,1)) + D_{s}(i,1)(1 - R^{u}(i,1))r_{i+1} + D_{b}(i+1,1)r_{i+1}(1 - R^{d}(i+1,1))\right)$$

$$(6.8.4)$$

$$r_2^u(i+1,2) = (1-q^u(i+1,2))R^d(i+1,2)$$
 (6.8.5)

$$r_3^u(i+1,2) = r_1^d(i,2)R^d(i,2)$$
(6.8.6)

6.8.2 Failure mode change

$$z_{1,2}^d(i,2) = 0 (6.8.7)$$

$$z_{1,2}^d(i,2) = 0 (6.8.8)$$

$$z_{1,2}^d(i,2) = q^u(i+1,2)R^d(i+1,2)$$
(6.8.9)

$$z_{2,3}^d(i,2) = q^u(i+1,2)(1-R^d(i+1,2))$$
(6.8.10)

$$z_{3,1}^d(i,2) = (1 - r_1^d(i,2))R^d(i+1,2)$$
 (6.8.11)

$$z_{3,2}^d(i,2) = r_1^d(i,2)(1 - R^d(i+1,2))$$
(6.8.12)

$$z_{1,2}^u(i+1,2) = 0 (6.8.13)$$

$$z_{1,3}^u(i+1,2) = 0 (6.8.14)$$

$$z_{2,1}^{u}(i+1,2) = q^{d}(i,2)R^{u}(i,2)$$
(6.8.15)

$$z_{2,3}^{u}(i+1,2) = (1 - R^{u}(i,2))q^{d}(i,2)$$
(6.8.16)

$$z_{3,1}^{u}(i+1,2) = R^{u}(i+1,2)(1-r^{u}(i+1,2))$$
(6.8.17)

$$z_{3,2}^{u}(i+1,2) = (1 - R^{u}(i+1,2))r^{u}(i+1,2)$$
(6.8.18)

6.8.3 Interruption of flow

$$p_{1}^{d}(i,2) = \frac{1}{P_{s}(i,1) + P_{b}(i+1,1)} \times (6.8.19) \\ \left(P_{s}(i,1) \left(R^{u}(i,1)q^{d}(i,1) + R^{u}(i,1)(1-q^{d}(i,1)) + R^{u}(i,1)q^{d}(i,1) \right) \\ + P_{b}(i+1,1) \left(q^{u}(i+1,1)(1-R^{d}(i+1,1)) \\ + (1-q^{u}(i+1,1))R^{d}(i+1,1) + q^{u}(i+1,1)R^{d}(i+1,1) \right) \right)$$

$$p_{2}^{d}(i,2) = \frac{1}{W^{d}(i,2)} \times (6.8.20) \\ \left(X_{2}^{d}(i,2) \left(R^{d}(i+1,2) + q^{u}(i+1,2)(1-R^{d}(i+1,2)) \right) -X_{3}^{d}(i,2)r_{1}^{d}(i,2)(1-R^{d}(i+1,2)) \right) \\ p_{3}^{d}(i,2) = 0$$

$$(6.8.21)$$

$$p_{1}^{u}(i+1,2) = \frac{1}{P_{s}(i,1) + P_{b}(i+1,1)} \times (6.8.22)$$

$$\begin{pmatrix} P_{s}(i,1) \left(R^{u}(i+1,1)q^{d}(i,1) + R^{u}(i+1,1)(1-q^{d}(i,1)) + R^{u}(i+1,1)q^{d}(i,1) \right) \\ + R^{u}(i+1,1)q^{d}(i,1) \end{pmatrix}$$

$$+ P_{b}(i+1,1) \left(q^{u}(i+1,1)(1-R^{d}(i,1)) + (1-q^{u}(i+1,1))R^{d}(i,1) + q^{u}(i+1,1)R^{d}(i,1) \right) \end{pmatrix}$$

$$p_{2}^{u}(i+1,2) = \frac{1}{W^{d}(i+1,2)} \times (6.8.23) \\ \left(X_{2}^{d}(i,2) \left(R^{d}(i,2) + q^{u}(i+1,2)(1-R^{d}(i,2)) \right) - X_{3}^{d}(i,2)r_{1}^{d}(i,2)(1-R^{d}(i,2)) \right)$$

$$p_{3}^{u}(i+1,2) = 0 \qquad (6.8.24)$$

6.8.4 Idleness failure

$$q_{1}^{d}(i,2) = p_{i+1} \frac{W^{d}(i,2)}{W^{d}(i,2) + X_{2}^{d}(i,2) + P_{s}(i,2)}$$

$$+ \frac{\left(P_{s}(i,1)R^{u}(i,1) + P_{b}(i+1,1)R^{d}(i+1,1)\right)}{P_{s}(i,1) + P_{b}(i+1,1)}$$

$$q_{2}^{d}(i,2) = 0$$

$$(6.8.26)$$

$$q_3^d(i,2) = 0 (6.8.27)$$

$$q_{1}^{u}(i+1,2) = p_{i} \frac{W^{u}(i+1,2)}{W^{u}(i+1,2) + X_{2}^{u}(i+1,2) + P_{b}(i+1,2)}$$
(6.8.28)
+ $\frac{\left(P_{s}(i,1)R^{u}(i,1) + P_{b}(i+1,1)R^{d}(i+1,1)\right)}{P_{s}(i,1) + P_{b}(i+1,1)}$ (6.8.29)
 $q_{3}^{u}(i+1,2) = 0$ (6.8.30)

Chapter 7

Decomposition Analysis for Type 3

7.1 State definitions

The downstream machine $M^d(i,3)$, i = 1, ..., K - 1, represents all the Type 3 flow behavior from downstream $B_{i,3}$. The upstream machine $M^d(i,3)$ is up, when M_{i+1} is up and is either blocked or starved for Type 1, and is also starved or blocked for Type 2, but is not blocked for Type 3. That is,

$$\Upsilon^{d}(i,3) = \{ \alpha_{i+1} = 1 \cap (n_{i,1} = 0 \cup n_{i+1,1} = N_{i+1,1})$$

$$\cap (n_{i,2} = 0 \cup n_{i+1,2} = N_{i+1,2}) \cap n_{i+1,2} < N_{i+1,2} \}$$
(7.1.1)

From the state definitions defined in Chapter 6, this up state can be express as

$$\Upsilon^{d}(i,3) = \{ (\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \}$$
(7.1.2)

A down state of $M^d(i,3)$ is the complement of the up state,

$$\sim \{ (\Delta_2^d(i,2) \cup \Delta_2^u(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \}$$
(7.1.3)

In order to derive the decomposition equations, we categorized the down state into three different states. Before we define each down state, the following terms are defined first for our convenience.

- $A = \{\Delta_2^d(i,2) \cup \Delta_2^u(i+1,2)\}$
- $B = \{n_{i+1,3} < N_{i+1,3}\}$

The event A is the state that M_{i+1} is up and either stated or blocked for Type 1 or Type 2. The down state in (7.1.3) can be written as ~ $\{A \cap B\}$. This event is equivalent to

$$\begin{array}{lll} \sim (A \cap B) &=& \sim A \cup \sim B \\ \\ &=& (\sim A \cap B) \cup (A \cap \sim B) \cap (\sim A \cap \sim B) \end{array}$$

Therefore, three separate down states are defined as following:

$$\begin{aligned} \Delta_1^d(i,3) &= (\sim A \cap B) \\ &= \{(\sim \Delta_2^d(i,2) \cap \sim \Delta_2^u(i+1,2)) \cap n_{i+1,3} < N_{i+1,3}\} \\ \Delta_2^d(i,3) &= (A \cap B) \\ &= \{(\Delta_2^d(i,2) \cup \Delta_2^u(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}\} \\ \Delta_3^d(i,3) &= (\sim A \cap \sim B) \\ &= \{(\sim \Delta_2^d(i,2) \cap \sim \Delta_2^u(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}\} \end{aligned}$$
(7.1.4)

7.2 Equalities

For convenience, we define the following building block probabilities:

$$W^{u}(i,3) = Pr[\Upsilon^{u}(i,3) \cap n_{i,3} < N_{i,2}]$$

$$W^{d}(i,3) = Pr[\Upsilon^{d}(i,3) \cap n_{i,3} > 0]$$

$$X^{u}_{m}(i,3) = Pr[\Delta^{u}_{m}(i,3) \cap n_{i,3} < N_{i,2}], \quad m = 1, 2, 3 \quad (7.2.1)$$

$$X^{d}_{n}(i,3) = Pr[\Delta^{d}_{n}(i,3) \cap n_{i,3} > 0], \quad n = 1, 2, 3$$

$$P_{b}(i,3) = Pr[\Upsilon^{u}(i,3) \cap n_{i,3} = N_{i,2}]$$

$$P_{s}(i,3) = Pr[\Upsilon^{d}(i,3) \cap n_{i,3} = 0]$$

$$D_{b}(i,3) = Pr[\Delta_{1}(i,3) \cap n_{i,3} = N_{i,3}]$$

$$D_{s}(i,3) = Pr[\Delta_{1}(i,3) \cap n_{i,3} = 0]$$

$$i = 1 \dots K - 1$$

In Equation (6.1.1), the following equality can be stated.

$$\Upsilon^d(i,3) = \{ (\Delta_2^d(i,3) \cup \Delta_2^u(i+1,3)) \cap n_{i+1,3} < N_{i+1,3} \}$$

Then

$$W^{d}(i,3) = Pr[\Upsilon^{d}(i,3) \cap n_{i,3} < N_{i,3}]$$

= $Pr[(\Delta_{2}^{d}(i,3) \cup \Delta_{2}^{u}(i+1,3)) \cap n_{i+1,3} < N_{i+1,3} \cap n_{i,3} > 0]$
= $W^{u}(i+1,2)$

Next, $X_1^d(i,3)$ is the probability that the downstream machine of L(i,3) is down in Mode 1 and not starved. From the definition (6.1.3) and (7.2.1), it is

$$\begin{aligned} X_1^d(i,3) &= Pr[\Delta_1^d(i,3) \cap n_{i,3} > 0] \\ &= Pr[\Delta^u(i+1,3) \cap n_{i+1,3} < N_{i+1,3}] \\ &= X_1^u(i+1,3) \end{aligned}$$

Therefore,

$$X_1^d(i,3) = X_1^u(i+1,3) \tag{7.2.2}$$

Next, $X_2^d(i,3)$ is the probability that the downstream machine of L(i,3) is down in Mode 2 and not starved. From the definition (6.1.3), it is

$$X_2^d(i,3) = Pr[\{\alpha_{i+1} = 1 \cap \{n_{i,3} = 0 \cup n_{i+1,3} = N_{i+1,3}\} \\ \cap n_{i+1,3} = N_{i+1,3} \cap n_{i,3} > 0] \\ = Pr[\Upsilon^u(i+1,3) \cap n_{i+1,3} = N_{i+1,3}] \\ = P_b(i+1,3)$$

A similar equality can be derived for $X_2^u(i+1,1)$. Therefore,

$$X_2^d(i,3) = P_b(i+1,3)$$
 (7.2.3)

$$X_2^u(i+1,3) = P_s(i,3) (7.2.4)$$

Last,

$$X_3^d(i,3) = Pr[\alpha_{i+1} = 0 \cup \{\alpha_{i+1} = 1 \cap n_{i,3} > 0 \cup n_{i+1,3} < N_{i+1,3}\}$$
$$\cap n_{i,3} > 0 \cap n_{i+1,3} = N_{i+1,3}]$$
$$= Pr[\Delta_3^u(i+1,3) \cap n_{i+1,3} = N_{i+1,3}]$$
$$= D_b(i+1,3)$$

Again, $X_3^u(i+1,3)$ can be derived in the similar way. Therefore,

$$X_3^d(i,3) = D_b(i+1,3) (7.2.5)$$

$$X_3^u(i+1,3) = D_s(i,3) (7.2.6)$$

7.3 Resumption of flow

Since there are three down states for $M^d(i, 2)$, three separate repair probabilities need to be derived. First, we define the following quantities for the derivations of the resumption of flow equations.

$$R^{d}(i,3) = \frac{\sum_{n=1}^{3} r_{n}^{d}(i,3) X_{n}^{d}(i,3)}{\sum_{n=1}^{3} X_{n}^{d}(i,3)}$$

likewise,

$$R^{u}(i,3) = \frac{\sum_{m=1}^{3} r_{m}^{u}(i,3) X_{m}^{u}(i,3)}{\sum_{m=1}^{3} X_{m}^{d}(i,3)}$$

By the definition the repair probability from the first down state of $M^d(i,3)$ is

$$r_{1}^{d}(i,3) = Pr[\Upsilon^{d}(i,3) \text{ at } t+1|\Delta_{1}^{d}(i,3) \text{ at } t]$$

$$= Pr[(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$

$$(\sim \Delta_{2}^{d}(i,2) \cup \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t]$$
(7.3.1)

Since $M^{d}(i, 3)$ is down at t, even if it is repaired, the buffer state $n_{i+1,3} < N_{i+1,3}$ will remain as it is at t + 1. Therefore,

$$r_1^d(i,3) = Pr[(\Delta_2^d(i,2) \cup \Delta_2^u(i+1,2)) \text{ at } t+1]$$

$$(\sim \Delta_2^d(i,2) \cap \sim \Delta_2^u(i+1,2)) \text{ at } t]$$
(7.3.2)

Notice that

$$\{(\sim \Delta_2^d(i,2) \cap \sim \Delta_2^u(i+1,2))\}$$

= $\{(\Upsilon^d(i,2) \cap n_{i,2} > 0) \cup (\Delta_1^d(i,2) \cap n_{i,2} > 0) \cup (\Delta_3^d(i,2) \cap n_{i,2} > 0) \cup (\Delta_3^d(i+1,2) \cap n_{i+1,2} < N_{i+1,2})\}$

Since the assumption stated in (3.2.3),

$$\{n_{i,2} = 0 \cap n_{i+1,2} = N_{i+1,2}\} = \emptyset$$

it can be rewritten as

$$\{(\sim \Delta_2^d(i,2) \cap \sim \Delta_2^u(i+1,2))\}$$

= $(\Upsilon^d(i,2) \cap n_{i,2} > 0) \cup (\Delta_1^d(i,2) \cap n_{i,2} > 0) \cup (\Delta_3^d(i,2)) \cup (\Delta_3^d(i+1,2))$

Therefore,

$$r_{1}^{d}(i,3) = 1/\left(W^{d}(i,2) + D_{s}(i,2) + D_{b}(i+1,2) + X_{1}^{d}(i,2)\right) \times \left(W^{d}(i,2)(p_{2}^{u}(i+1,2) + p_{2}^{d}(i,2)) + D_{s}(i,2)(1 - R^{u}(i,2))R^{d}(i+1,2) + D_{b}(i+1,2)R^{u}(i,2)(1 - R^{d}(i+1,2))\right)$$

$$(7.3.3)$$

Next, the resumption of flow from the second down state is derived such that

$$\begin{aligned} r_{2}^{d}(i,3) &= Pr[\Upsilon^{d}(i,2) \text{ at } t+1|\Delta_{2}^{d}(i,2) \text{ at } t] \\ &= Pr[(\Delta_{2}^{d}(i,2)\cup\Delta^{u}(i+1,2))\cap n_{i+1,3} < N_{i+1,3} \text{ at } t+1| \\ &\quad (\Delta_{2}^{d}(i,2)\cup\Delta_{2}^{u}(i+1,2))\cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \\ &= Pr[(\Delta_{2}^{d}(i,2)\cup\Delta^{u}(i+1,2)) \text{ at } t+1| \\ &\quad (\Delta_{2}^{d}(i,2)\cup\Delta_{2}^{u}(i+1,2))\cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \\ &\times Pr[n_{i+1,3} < N_{i+1,3} \text{ at } t+1| \\ &\quad (\Delta_{2}^{d}(i,2)\cup\Delta_{2}^{u}(i+1,2))\cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \end{aligned}$$

Note that

$$Pr[\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2) \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= Pr[\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2) \text{ at } t+1|\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2) \text{ at } t]$$

$$= (1-q^{u}(i+1,3))$$

Also,

$$Pr[n_{i+1,3} < N_{i+1,3} \text{ at } t+1]$$

$$(\Delta_2^d(i,2) \cup \Delta_2^u(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= R^d(i+1,3)$$
(7.3.6)

Therefore,

$$r_2^d(i,3) = (1 - q^u(i+1,3))R^d(i+1,3)$$
(7.3.7)

Next the derivation of $r_3^d(i,3)$ is as follows,

$$\begin{aligned} r_{3}^{d}(i,3) &= Pr[\Upsilon^{d}(i,3) \text{ at } t+1|\Delta_{3}^{d}(i,3) \text{ at } t] \\ &= Pr[\Upsilon^{d}(i,3) \text{ at } t+1| \\ &\quad (\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \\ &= Pr[(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \text{ at } t+1| \\ &\quad (\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \\ &\times Pr[n_{i+1,3} < N_{i+1,3} \text{ at } t+1| \\ &\quad (\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t] \\ &= r_{1}^{d}(i,3)R^{d}(i+1,3) \end{aligned}$$

7.4 Failure Mode Changes

Again, the down state, $\Delta_1^d(i, 3)$, represents the state that either M_{i+1} is down or M_{i+1} is working on Type 1 or Type 2. From the observer in $B_{i,3}$, any state change further downstream of M_{i+1} , except the resumption of flow, will not change the state of $M^d(i, 3)$. Therefore, there in no transition from $\Delta_1^d(i, 3)$ to any other down state. Therefore,

$$z_{1,2}^d(i,3) = 0 \tag{7.4.1}$$

$$z_{1,3}^d(i,3) = 0 (7.4.2)$$

Next, once $M^d(i, 3)$ is in the down state $\Delta_2^d(i, 3) - M_{i+1}$ is up and is able to work for Type 3 due to starvation or blockage of Type 1 or Type 2 but idle due to blockage for Type 3, it can make a transition into two other down states. First, let us consider the transition to $\Delta_3^d(i, 3)$

$$z_{2,3}^{d}(i,3) = Pr[\Delta_{3}^{d}(i,3) \text{ at } t+1|\Delta_{2}^{d}(i,3) \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \text{ at } t|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$\times Pr[n_{i+1,3} = N_{i+1,3} \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= q^{u}(i+1,3)(1 - R^{d}(i+1,3))$$

With the similar approach, for $z_{2,1}^d(i,3)$,

$$z_{2,1}^{d}(i,3) = Pr[\Delta_{1}^{d}(i,3) \text{ at } t+1|\Delta_{2}^{d}(i,3) \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$\times Pr[n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= q^{u}(i+1,3)R^{d}(i+1,3)$$

$$(7.4.4)$$

For $z_{3,1}^d(i,3)$

$$z_{3,1}^{d}(i,3) = Pr[\Delta_{1}^{d}(i,3) \text{ at } t+1|\Delta_{3}^{d}(i,3) \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \text{ at } t+1|$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$\times Pr[n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3} \text{ at } t]$$

$$= (1 - r_{1}^{d}(i,3))R^{d}(i+1,3)$$

Also for $z_{3,2}^d(i,3)$

$$z_{3,2}^{d}(i,3) = Pr[\Delta_{1}^{d}(i,2)|\Delta_{2}^{d}(i,3)]$$

$$= Pr[(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}]$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}]$$

$$= Pr[(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2))|$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}]$$

$$\times Pr[n_{i+1,3} = N_{i+1,3}]$$

$$(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} = N_{i+1,3}]$$

$$= r_{1}^{d}(i,3)(1 - R^{d}(i+1,3))$$

$$(7.4.6)$$

7.5 Interruption of Flow

Since there are three down states, three interruption of flow probabilities need to be derived. First, the probability $p_1^d(i,3)$ is the transition from $\Upsilon^d(i,2)$ to $\Delta_1^d(i,3)$, which represents the failure of $M^d(i,3)$ when $B_{i,3}$ is not empty. This transition probability

$$p_{1}^{d}(i,3) = Pr[\Delta_{1}^{d}(i,3) \text{ at } t+1|\Upsilon^{d}(i,3) \cap n_{i,3} > 0 \text{ at } t]$$
(7.5.1)
$$= Pr[(\sim \Delta_{2}^{d}(i,2)) \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$
$$(\Delta_{2}^{d}(i,2) \cup \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t]$$
$$= Pr[(\sim \Delta_{2}^{d}(i,2)) \sim \Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t+1|$$
$$(\Delta_{2}^{d}(i,2) \cap n_{i+1,3} < N_{i+1,3}) \cup$$
$$(\Delta_{2}^{u}(i+1,2)) \cap n_{i+1,3} < N_{i+1,3} \text{ at } t]$$

Note that if M_i is working on a Type 3 part at time t, it cannot get failed and blocked for a Type 3 part at the same time at t + 1. Therefore,

$$p_{1}^{d}(i,3) = Pr[(\sim \Delta_{2}^{d}(i,2) \cap \sim \Delta_{2}^{u}(i+1,2)) \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2) \cup (\Delta_{2}^{u}(i+1,2) \text{ at } t]$$

$$= Pr[(\Upsilon^{d}(i,2) \cap n_{i,2} > 0) \cup (\Delta_{1}^{d}(i,2) \cap n_{i,2} > 0)$$

$$\cup \Delta_{3}^{d}(i,2) \cup \Delta_{3}^{u}(i+1,2) \text{ at } t+1|$$

$$(\Delta_{2}^{d}(i,2)) \cup (\Delta_{2}^{u}(i+1,2) \text{ at } t]$$

$$(7.5.2)$$

By applying the probability quantities defined in Section 7.2,

$$p_{1}^{d}(i,3) = \frac{1}{P_{s}(i+1,3) + P_{b}(i,3)} \times (7.5.3)$$

$$\begin{pmatrix} P_{b}(i,3)Pr[(\Upsilon^{d}(i,2) \cap n_{i,2} > 0) \cup (\Delta_{1}^{d}(i,2) \cap n_{i,2} > 0) \\ \cup \Delta_{3}^{d}(i,2) \cup \Delta_{3}^{u}(i+1,2) \text{ at } t+1 | \Delta_{2}^{d}(i,2) \text{ at } t] \\ +P_{s}(i+1,3)Pr[(\Upsilon^{d}(i,2) \cap n_{i,2} > 0) \cup (\Delta_{1}^{d}(i,2) \cap n_{i,2} > 0) \\ \cup \Delta_{3}^{d}(i,2) \cup \Delta_{3}^{u}(i+1,2) \text{ at } t+1 | \Delta_{2}^{u}(i+1,2) \text{ at } t] \end{pmatrix}$$

is

The following identities can be defined.

$$\begin{aligned} \Pr[\Upsilon^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t + 1 | \Delta_{2}^{d}(i,2) \text{ at } t] &= q^{u}(i+1,2)R^{d}(i+1,2) \\ \Pr[\Delta_{1}^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t + 1 | \Delta_{2}^{d}(i,2) \text{ at } t] &= (1-q^{u}(i+1,2))R^{d}(i+1,2) \\ \Pr[\Delta_{3}^{d}(i,2) \text{ at } t + 1 | \Delta_{2}^{d}(i,2) \text{ at } t] &= q^{u}(i+1,2)(1-R^{d}(i+1,2)) \\ \Pr[\Delta_{3}^{u}(i+1,2) \text{ at } t + 1 | \Delta_{2}^{d}(i,2) \text{ at } t] &= 0 \\ \Pr[\Upsilon^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t + 1 | \Delta_{2}^{u}(i+1,2) \text{ at } t] &= q^{d}(i,2)R^{u}(i,2) \\ \Pr[\Delta_{1}^{d}(i,2) \cap n_{i,2} > 0 \text{ at } t + 1 | \Delta_{2}^{u}(i+1,2) \text{ at } t] &= (1-q^{d}(i,2))R^{u}(i,2) \\ \Pr[\Delta_{3}^{d}(i,2) \text{ at } t + 1 | \Delta_{2}^{u}(i+1,2) \text{ at } t] &= q^{d}(i,2)(1-R^{u}(i,2)) \\ \Pr[\Delta_{3}^{u}(i+1,2) \text{ at } t + 1 | \Delta_{2}^{u}(i+1,2) \text{ at } t] &= 0 \end{aligned}$$

Then

$$p_{1}^{d}(i,3) = \frac{1}{P_{s}(i+1,2) + P_{b}(i,2)} \times (7.5.4)$$

$$\left(P_{s}(i+1,2) \left(R^{u}(i,2)q^{d}(i,2) + R^{u}(i,2)(1-q^{d}(i,2)) + (1-R^{u}(i,2))q^{d}(i,2) \right) + P_{b}(i,2) \left(q^{u}(i+1,2)(1-R^{d}(i+1,2)) + (1-Q^{u}(i+1,2))R^{d}(i+1,2) + Q^{u}(i+1,2)R^{d}(i+1,2) \right) \right)$$

Next, we derive the interruption of flow equation for $p_2^d(i,3)$, the transition probability from the up state to the state in which $M^d(i+1,3)$ is down and $B_{i+1,3}$ is full. This transition probability is,

$$p_2^d(i,3) = Pr[\Delta_2^d(i,3) \text{ at } t+1 \mid \Upsilon^d(i,3) \cap n_{i,3} > 0 \text{ at } t]$$

We first start with the derivation of this equation by applying the fact that the probability of going out of a state is equal to the probability of going into that state.

$$\begin{aligned} X_2^d(i,3) & \left((1-q^u(i+1,3))R^d(i+1,3) + q^u(i+1,3)R^d(i+1,3) \right. \\ & \left. + q^u(i+1,3)(1-R^d(i+1,3)) \right) \\ & \left. = W^d(i,3)p_2^d(i,3) + X_3^d(i+1,3)r_1^d(i,3)(1-R^d(i+1,3)) \right) \end{aligned}$$

If we simplify this equation, then

$$X_2^d(i,3) \left(R^d(i+1,3) + q^u(i+1,3)(1-R^d(i+1,3)) \right)$$

= $W^d(i,3) p_2^d(i,3) + X_3^d(i,3) r_1^d(i,3)(1-R^d(i+1,3))$

That is,

$$p_{2}^{d}(i,3) = \frac{1}{W^{d}(i,3)} \times$$

$$\left(X_{2}^{d}(i,3) \left(R^{d}(i+1,3) + q^{u}(i+1,3)(1-R^{d}(i+1,3)) \right) - X_{3}^{d}(i,3)r_{1}^{d}(i,3)(1-R^{d}(i+1,3)) \right)$$

$$(7.5.5)$$

As a next step, we derive the interruption of flow equation for $p_3^d(i,3)$, the transition probability from the up state to the state in which $M^d(i+1,3)$ is down and $B_{i+1,3}$ is full and at the same time M_{i+1} gets failed or switches to Type 1 production. Note that when M_{i+1} is up and not blocked for Type 2, it is impossible for M_{i+1} to get blocked for Type 2 when it either gets failed or switched to Type 1. Therefore,

$$p_3^d(i,3) = Pr[\Delta_3^d(i,3) \text{ at } t+1 | \Upsilon^d(i,3) \cap n_{i,3} > 0 \text{ at } t]$$
(7.5.6)
= 0

7.6 Idleness Failure

Again, the probability $q_1^d(i,3)$ represents the probability that $M^d(i,1)$ is down at t+1 given that it was up and starved at t. That is,

$$q^{d}(i,1) = Pr[\Delta_{1}^{d}(i,1) \text{ at } t+1 | \Upsilon^{d}(i,1) \cap n_{i,1} = 0 \text{ at } t]$$
(7.6.1)

Note that the idleness failure of Type 3 is different from those of Type 1 and Type 2, since Type 3 is the lowest priority type. For example, the idleness failure in Type 1 occurs when M_i gets failed while it is either blocked or starved for Type 1. Likewise, for Type 2, the idleness failure happens when M_i is either get failed or it switches back to Type 1 production, while it is blocked or starved for Type 2. Suppose M_i is blocked and starved for Type 1 and Type 2, and it works for Type 3. While it is working for a Type 3 part, it is also blocked or starved for Type 3. In this case, M_i is blocked and starved for all the part types and is idle. Although, with our assumption, the machine cannot not get failed at this condition, the idleness failure if M_i starts making any higher priority part type. We argue that the idleness failure for $M^d(i, 3)$ is as follows:

$$q_{1}^{d}(i,3) = \frac{P_{b}(i+1,2)}{P_{s}(i+1,2) + P_{b}(i,2)} R^{d}(i+1,2) + \frac{P_{s}(i,2)}{P_{s}(i+1,2) + P_{b}(i,2)} R^{u}(i,2)$$
(7.6.2)

That is, $q_1^d(i,3)$ is the weighted sum of the repair probabilities of the higher priority part. If M_i is no longer blocked or starved for Type 1 or Type 2, while M_i is idle, it will start making the higher priority part type. Similarly for $q_1^u(i,3)$ is

$$q_{1}^{u}(i,3) = \frac{P_{b}(i,2)}{P_{s}(i-1,2) + P_{b}(i,2)} R^{d}(i,2)$$

$$+ \frac{P_{s}(i-1,2)}{P_{s}(i-1,2) + P_{b}(i,2)} R^{u}(i-1,2)$$
(7.6.3)

7.7 Summary of the Decomposition Equations for Type 3

For $i = 1 \dots K$

7.7.1 Resumption of flow

$$r_{1}^{d}(i,3) = 1/\left(W^{d}(i,2) + D_{s}(i,2) + D_{b}(i+1,2) + X_{1}^{d}(i,2)\right) \times \left(W^{d}(i,2)(p_{2}^{u}(i+1,2) + p_{2}^{d}(i,2)) + D_{s}(i,2)(1 - R^{u}(i,2))R^{d}(i+1,2) + D_{b}(i+1,2)R^{u}(i,2)(1 - R^{d}(i+1,2))\right)$$

$$(7.7.1)$$

$$r_2^d(i,3) = (1 - q^u(i+1,3))R^d(i+1,3)$$
 (7.7.2)

$$r_3^d(i,3) = r_1^d(i,3)R^d(i+1,3)$$
(7.7.3)

$$r_{1}^{u}(i+1,3) = 1/\left(W^{u}(i+1,2) + D_{s}(i,2) + D_{b}(i+1,2) + X_{1}^{d}(i,2)\right) \times \left(W^{u}(i+1,2)(p_{2}^{u}(i+1,2) + p_{2}^{d}(i,2)) + D_{s}(i,2)(1 - R^{d}(i,2))r_{i+1} + D_{b}(i+1,2)r_{i+1}(1 - R^{d}(i+1,2))\right)$$

$$(7.7.4)$$

$$r_2^u(i+1,3) = (1-q^d(i,3))R^d(i,3)$$
(7.7.5)

$$r_3^u(i+1,3) = r_1^d(i,3)R^u(i,3)$$
(7.7.6)

7.7.2 Failure mode changes

$$z_{1,2}^d(i,3) = 0 (7.7.7)$$

$$z_{1,3}^d(i,3) = 0 (7.7.8)$$

$$z_{2,3}^d(i,3) = q^u(i+1,3)(1-R^d(i+1,3))$$
 (7.7.9)

$$z_{2,1}^d(i,3) = q^u(i+1,3)R^d(i+1,3)$$
 (7.7.10)

$$z_{3,1}^d(i,3) = (1 - r_1^d(i,3))R^d(i+1,3)$$
(7.7.11)

$$z_{3,2}^d(i,3) = r_1^d(i,3)(1 - R^d(i+1,3))$$
(7.7.12)

$$z_{1,2}^u(i+1,3) = 0 (7.7.13)$$

$$z_{1,3}^u(i+1,3) = 0 (7.7.14)$$

$$z_{2,3}^{u}(i+1,3) = q^{d}(i,3)(1-R^{u}(i,3))$$
(7.7.15)

$$z_{2,1}^{u}(i+1,3) = q^{d}(i+1,3)R^{u}(i,3)$$
(7.7.16)

$$z_{3,1}^{u}(i+1,3) = (1 - r_1^{u}(i+1,3))R^{u}(i,3)$$
(7.7.17)

$$z_{3,2}^{u}(i+1,3) = r_1^{u}(i+1,3)(1-R^{u}(i,3))$$
(7.7.18)

7.7.3 Interruption of flow

$$p_{1}^{d}(i,3) = \frac{1}{P_{s}(i+1,2) + P_{b}(i,2)} \times$$

$$\left(P_{s}(i+1,2) \left(R^{u}(i,2)q^{d}(i,2) + R^{u}(i,2)(1-q^{d}(i,2)) + (1-R^{u}(i,2))q^{d}(i,2) \right) + (1-R^{u}(i,2))q^{d}(i,2) \right) + P_{b}(i,2) \left(q^{u}(i+1,2)(1-R^{d}(i+1,2)) + (1-q^{u}(i+1,2))R^{d}(i+1,2) + q^{u}(i+1,2)R^{d}(i+1,2) \right) \right)$$

$$p_{2}^{d}(i,3) = \frac{1}{W^{d}(i,2)} \times (7.7.20) \\ \left(X_{2}^{d}(i,3) \left(R^{d}(i+1,3) + q^{u}(i+1,3)(1-R^{d}(i+1,3)) \right) - X_{3}^{d}(i,3)r_{1}^{d}(i,3)(1-R^{d}(i+1,3)) \right)$$

$$p_{3}^{d}(i,3) = 0$$

$$p_{1}^{u}(i+1,3) = \frac{1}{(7.7.22)} \times$$

$$(7.7.22)$$

$$p_{1}^{*}(i+1,3) = \frac{1}{P_{s}(i,2) + P_{b}(i+1,2)} \times (1.1.22)$$

$$\left(P_{s}(i,2) \left(R^{d}(i+1,2)q^{u}(i+1,2) + R^{d}(i+1,2)(1-q^{u}(i+1,2)) + (1-R^{d}(i+1,2))q^{u}(i+1,2) \right) + (1-R^{d}(i+1,2))q^{u}(i+1,2) \right) + (1-q^{d}(i,2)(1-R^{u}(i,2)) + (1-q^{d}(i,2))R^{u}(i,2) + q^{d}(i,2)R^{u}(i,2) \right) \right)$$

$$p_{2}^{u}(i+1,3) = \frac{1}{W^{u}(i+1,2)} \times (7.7.23) \left(X_{2}^{u}(i+1,3) \left(R^{u}(i,3) + q^{d}(i,3)(1-R^{u}(i,3)) \right) - X_{3}^{u}(i+1,3)r_{1}^{u}(i+1,3)(1-R^{u}(i,3)) \right)$$

$$p_{3}^{u}(i+1,3) = 0 \qquad (7.7.24)$$

 $p_3^u(i+1,3) = 0$

7.7.4 Idleness failure

$$q_1^d(i,3) = \frac{P_b(i,2)R^d(i,2) + P_s(i-1,2)R^u(i-1,2)}{P_s(i-1,2) + P_s(i,2)}$$
(7.7.25)

$$q_2^d(i,3) = 0 (7.7.26)$$

$$q_3^d(i,3) = 0 (7.7.27)$$

$$q_1^u(i+1,3) = \frac{P_b(i+1,2)R^d(i+1,2) + P_s(i,2)R^u(i,2)}{P_s(i,2) + P_b(i,2)}$$
(7.7.28)

$$q_2^u(i+1,3) = 0 (7.7.29)$$

$$q_3^u(i+1,3) = 0 (7.7.30)$$
Chapter 8

Generalization of Decomposition

Although the decomposition equations introduced in Chapter 5, 6, and 7 are constructed for the line processing three part types, the state classifications in the building blocks are designed in such a way that the equations can be generalized and applied to a production line processing more than three-part types. This section presents an idea of the general decomposition based on the equations derived in the previous chapters. The total number of part types that the general flexible line processes is denoted by Z. In this section, we let j be the index for a specific part type. The decomposition is categorized into three different cases: the highest priority (j = 1), the intermediate priority (j = 2, ..., Z), and the lowest priority part type (j = Z).

8.1 Idea of Generalization

The decomposition equations in Section 5.8, 6.8, and 7.7 are constructed so that they can be generalized for a production line processing more than three part types. Note that the machines in the building blocks for all the part types share the same number of up and down states as well as the identical structure of the Markov transitions, regardless of the number of machines in the line and the number of part types. Moreover, comparing the equations in Section 5.8, 6.8, and 7.7, considerable similarity is found in those equations within part types. Observe that those quantities are the function of the adjacent building blocks within the part types and also the function of the immediately higher or lower part type building blocks. For example, the $p_1^d(i, 2)$ in the building block, L(i, 2) is function of the parameters in L(i + 1, 2), L(i, 1), L(i, 3), L(i + 1, 1), and L(i + 1, 3). If these equations are generalized for Z-part-type line, the behavior of the lower or higher priority part type is observed by the immediately lower or higher part type building blocks. That is, for the line $L(i, j_0)$, the flow behavior of the lower priority part types $(j = j_0 + 1...Z)$ is captured by $L(i, j_0 + 1)$, while the flow behavior of the higher priority part types $(j = 1...j_0 - 1)$ is captured by $L(i, j_0 - 1)$.

8.2 General Z Part Type Decomposition Equations

8.2.1 Highest priority part type

Comparing the equations in Section 5.8, 6.8, and 7.7, it can be observed that the failure and repair probabilities for the first down state are different from those in other part types. In the first part type building block, the first down state represents only the failure of a real machine in the line. Therefore, as stated in (5.8.1), (5.8.4), (5.8.20), and (5.8.23), the machine parameters are identical to the real machine parameters. Also note that for the highest priority part, the existence of a lower priority part type is made apparent to Type 1 only through the existence of idleness failures. Note that idleness failure equations in the first part type (8.2.29) and (5.8.26) are the function of building blocks in Type 2. The decomposition equations for the highest priority part are the following:

Resumption of flow

$$r^d(i,1) = r_{i+1}$$
 (8.2.1)

$$r_2^d(i,1) = (1 - q^u(i+1,1))R^d(i+1,1))$$
 (8.2.2)

$$r_3^d(i,1) = r_{i+1}R^d(i+1,1)$$
 (8.2.3)

$$r_1^u(i+1,1) = r_{i+1} (8.2.4)$$

$$r_2^u(i+1,1) = (1-q^d(i,1))R^u(i,1)$$
 (8.2.5)

$$r_3^u(i+1,1) = R^u(i,1)r_{i+1} (8.2.6)$$

Failure mode change

$$z_{1,2}^d(i,1) = 0 (8.2.7)$$

$$z_{1,3}^d(i,1) = 0 (8.2.8)$$

$$z_{2,1}^d(i,1) = q_1^u(i+1,1)R^d(i+1,1)$$
 (8.2.9)

$$z_{2,3}^{d}(i,1) = q_1^{u}(i+1,1)((1-R^d(i+1,1)))$$

$$(8.2.10)$$

$$z_{3,1}^d(i,1) = (1-r_{i+1})R^d(i+1,1)$$
(8.2.11)

$$z_{3,2}^{a}(i,1) = r_{i+1} \left(1 - R^{a}(i+1,1) \right)$$
(8.2.12)

$$z_{1,2}^u(i+1,1) = 0 (8.2.13)$$

$$z_{1,3}^u(i+1,1) = 0 (8.2.14)$$

$$z_{2,1}^{u}(i+1,1) = q_{1}^{d}(i,1)R^{u}(i,1)$$
(8.2.15)

$$z_{2,3}^{u}(i+1,1) = \left(1 - R^{u}(i,1)\right) q_{1}^{d}(i,1)$$
(8.2.16)

$$z_{2,1}^{u}(i+1,1) = q_{1}^{d}(i,1)R^{u}(i,1)$$
(8.2.17)

$$z_{3,1}^{u}(i+1,1) = R^{u}(i,1)\left(1-r_{i+1}\right)$$

$$(8.2.18)$$

$$(3.2.18)$$

$$z_{3,2}^{u}(i+1,1) = \left(1 - R^{u}(i,1)\right)r_{i+1}$$
(8.2.19)

Interruption of flow

$$p_1^d(i,1) = p_{i+1} (8.2.20)$$

$$p_{2}^{d}(i,1) = \frac{1}{W^{d}(i,1)} \times$$

$$\left[X_{2}^{d}(i,1) \left(R^{d}(i+1,1) + q^{u}(i+1,1) \left(1 - R^{d}(i+1,1) \right) \right) \right) - X_{3}^{d}(i,1)r_{i+1}(1 - R^{d}(i+1,1)) \right]$$

$$(8.2.21)$$

$$p_3^d(i,1) = 0 (8.2.22)$$

$$p_1^u(i+1,1) = p_{i+1}$$
 (8.2.23)

$$p_{2}^{u}(i+1,1) = \frac{1}{W^{u}(i+1,1)} \times (8.2.24)$$

$$\left[X_{2}^{d}(i+1,1) \left(R^{u}(i,1) + \left(1 - R^{u}(i,1)\right) q^{d}(i,1) \right) - X_{2}^{d}(i+1,1) \left(1 - R^{u}(i,1)\right) r_{i+1} \right]$$

$$p_3^u(i+1,1) = 0 \tag{8.2.25}$$

Idleness failure

$$q_1^d(i,1) = p_{i+1} \frac{W_d(i,2)}{W_d(i,2) + X_2(i,2) + P_s(i,2)}$$
(8.2.26)
$$q_1^d(i,1) = 0$$
(8.2.27)

$$q_2^a(i,1) = 0 (8.2.27)$$

$$q^{u}(i+1,1) = p_{i+1} \frac{W(i+1,2)}{W^{u}(i+1,2) + P_{b}(i+1,2) + X_{2}(i+1,2)}$$
(8.2.29)

$$q_2^u(i+1,1) = 0 (8.2.30)$$

$$q_3^u(i+1,1) = 0 (8.2.31)$$

8.2.2 Intermediate priority part type

For the intermediate priority part type (j = 2...Z - 1), we have to account for the possibility that an observer in a intermediate part type building block will see flow into her buffer cease because a machine switches to a higher priority part. The first down states in the building block not only represents the failure of the real machine in the line, but it also represent the interruption of the flow caused by switching a production to a higher priority part type. Therefore, the failure and repair probabilities for the first down states, $\Delta_1^u(i, j)$ and $\Delta_1^d(i, j)$, are more complicated than those in the first part type building blocks. Like the first part type line, the the existence of a lower priority part type is felt through the existence of idleness failure. Note that idleness failure equations in the second part type, (8.2.59) and (8.2.59), are the function of Type 1 as well as Type 3.

The decomposition equations for intermediate priority parts are following ($i = 0 \dots K, j = 2 \dots Z - 1$).

Resumption of flow

$$r_{1}^{d}(i,j) = 1/\left(W^{d}(i,j-1) + D_{s}(i,j-1) + D_{b}(i+1,j-1) + X_{1}^{d}(i,j-1)\right) \times \left(W^{d}(i,1)(p_{2}^{u}(i+1,j-1) + p_{2}^{d}(i,j-1)) + D_{s}(i,j-1)(1 - R^{u}(i,j-1))r_{i+1}\right)$$

$$+D_b(i+1,j-1)r_{i+1}(1-R^d(i+1,j-1))\bigg)$$
(8.2.32)

$$r_2^d(i,j) = (1 - q^u(i+1,j))R^d(i+1,j)$$
(8.2.33)

$$r_3^d(i,j) = r_1^d(i,j)R^d(i+1,j)$$
(8.2.34)

$$r_{1}^{u}(i+1,j) = 1/\left(W^{u}(i+1,j-1) + D_{s}(i,j-1) + D_{b}(i+1,j-1) + X_{1}^{u}(i+1,j-1)\right) \times \left(W^{u}(i+1,j-1)(p_{2}^{u}(i+1,j-1) + p_{2}^{d}(i,j-1)) + D_{s}(i,j-1)(1-R^{u}(i,j-1))r_{i+1} + D_{b}(i+1,j-1)r_{i+1}(1-R^{d}(i+1,j-1))\right)$$

$$r_{2}^{u}(i+1,j) = (1-q^{u}(i+1,j))R^{d}(i+1,j)$$

$$(8.2.36)$$

$$r_3^u(i+1,j) = r_1^d(i,j)R^d(i,j)$$
 (8.2.37)

8.2.3 Failure mode change

$$z_{1,2}^d(i,j) = 0 (8.2.38)$$

$$z_{1,2}^d(i,j) = 0 (8.2.39)$$

$$z_{1,2}^d(i,j) = q^u(i+1,j)R^d(i+1,j)$$
 (8.2.40)

$$z_{2,3}^d(i,j) = q^u(i+1,j)(1-R^d(i+1,j))$$
 (8.2.41)

$$z_{3,1}^d(i,j) = (1 - r_1^d(i,2))R^d(i+1,j)$$
 (8.2.42)

$$z_{3,2}^d(i,j) = r_1^d(i,j)(1 - R^d(i+1,j))$$
 (8.2.43)

$$z_{1,2}^u(i+1,j) = 0 (8.2.44)$$

$$z_{1,3}^u(i+1,j) = 0 (8.2.45)$$

$$z_{2,1}^{u}(i+1,j) = q^{d}(i,j)R^{u}(i,j)$$
 (8.2.46)

$$z_{2,3}^{u}(i+1,j) = (1 - R^{u}(i,j))q^{d}(i,j)$$
 (8.2.47)

$$z_{3,1}^{u}(i+1,j) = R^{u}(i+1,j)(1-r^{u}(i+1,j))$$
(8.2.48)

$$z_{3,2}^{u}(i+1,j) = (1 - R^{u}(i+1,j))r^{u}(i+1,j)$$
(8.2.49)

8.2.4 Interruption of flow

$$p_{1}^{d}(i,j) = \frac{1}{P_{s}(i,j-1) + P_{b}(i+1,j-1)} \times (8.2.50) \\ \left(P_{s}(i,j-1) \left(R^{u}(i,j-1)q^{d}(i,j-1) + R^{u}(i,j-1)(1-q^{d}(i,j-1)) + R^{u}(i,j-1)q^{d}(i,j-1) \right) \right) \\ + R^{u}(i,j-1)q^{d}(i,j-1) \\ + P_{b}(i+1,j-1) \left(q^{u}(i+1,j-1)(1-R^{d}(i+1,j-1)) + (1-q^{u}(i+1,j-1)R^{d}(i+1,j-1)) \right) \\ + (1-q^{u}(i+1,j-1))R^{d}(i+1,j-1) + q^{u}(i+1,j-1)R^{d}(i+1,j-1) \right) \\ p_{2}^{d}(i,j) = \frac{1}{W^{d}(i,j)} \times (8.2.51)$$

$$p_{2}(i,j) = \frac{W^{d}(i,j)}{W^{d}(i,j)} \times (8.2.51)$$

$$\left(X_{2}^{d}(i,j) \left(R^{d}(i+1,j) + q^{u}(i+1,j)(1-R^{d}(i+1,j)) \right) - X_{3}^{d}(i,j)r_{1}^{d}(i,j)(1-R^{d}(i+1,j)) \right)$$

$$p_{3}^{d}(i,j) = 0 \qquad (8.2.52)$$

 $p_3^a(i,j) = 0$

$$p_{1}^{u}(i+1,j) = \frac{1}{P_{s}(i,j-1) + P_{b}(i+1,j-1)} \times (8.2.53) \\ \left(P_{s}(i,j-1) \left(R^{u}(i+1,j-1)q^{d}(i,j-1) + R^{u}(i+1,j-1)(1-q^{d}(i,j-1)) \right) + R^{u}(i+1,j-1)q^{d}(i,j-1) \right) \\ + P_{b}(i+1,j-1) \left(q^{u}(i+1,j-1)(1-R^{d}(i,j-1)) + (1-q^{u}(i+1,j-1)R^{d}(i,j-1)) \right) \right) \\ + (1-q^{u}(i+1,j-1))R^{d}(i,j-1) + q^{u}(i+1,j-1)R^{d}(i,j-1) \right) \right)$$

$$p_{2}^{u}(i+1,j) = \frac{1}{W^{d}(i+1,j)} \times (8.2.54)$$

$$\left(X_{2}^{d}(i,j) \left(R^{d}(i,j) + q^{u}(i+1,j)(1-R^{d}(i,j)) \right) - X_{3}^{d}(i,j)r_{1}^{d}(i,j)(1-R^{d}(i,j)) \right)$$

$$p_{3}^{u}(i+1,2) = 0 \qquad (8.2.55)$$

8.2.5 Idleness failure

$$q_{1}^{d}(i,j) = p_{i+1} \frac{W^{d}(i,j)}{W^{d}(i,j) + X_{2}^{d}(i,j) + P_{s}(i,j)}$$

$$+ \frac{\left(P_{s}(i,j-1)R^{u}(i,j-1) + P_{b}(i+1,j-1)R^{d}(i+1,j-1)\right)}{P_{s}(i,j-1) + P_{b}(i+1,j-1)}$$

$$q_{2}^{d}(i,j) = 0$$

$$(8.2.57)$$

$$q_{3}^{d}(i,j) = 0$$

$$(8.2.58)$$

$$q_{1}^{u}(i+1,j) = p_{i} \frac{W^{u}(i+1,j)}{W^{u}(i+1,j) + X_{2}^{u}(i+1,j) + P_{b}(i+1,j)}$$

$$+ \frac{\left(P_{s}(i,j-1)R^{u}(i,j-1) + P_{b}(i+1,j-1)R^{d}(i+1,j-1)\right)}{P_{s}(i,j-1) + P_{b}(i+1,j-1)}$$

$$q_{2}^{u}(i+1,j) = 0$$

$$(8.2.60)$$

$$q_{3}^{u}(i+1,j) = 0$$

$$(8.2.61)$$

8.2.6 Lowest priority part type

The unique feature of the lowest priority part type (j = Z) is its idleness failure. Suppose that M_i is idle due to blockage of starvation for all the part types. At this moment, the observer in Type Z believes that a machine she watches is down like the observers in the rest of the part type building blocks. Note that with our assumption, M_i is not allowed to fail while it is idle. Therefore, the Type Z observer never sees the failure of her machine due to a failure of M_i while it is idle. However, the idleness failure still can be observed in a different way. Consider the following case; at this moment when M_i is idle, M_i becomes neither starved nor blocked for some higher priority part types ($j = 1 \dots Z - 1$). Then the Type Z observer believes that the machine that has been idle fails. This failure of the machine in the building block occurs because this machine switches a production to a higher priority part type. Note that this is the only case that the idleness failure can occur. Therefore, the idleness failure for Type Z is different from the rest of the part types.

Resumption of flow

$$r_{1}^{d}(i,Z) = 1/\left(W^{d}(i,Z-1) + D_{s}(i,Z-1) + D_{b}(i+1,Z-1) + X_{1}^{d}(i,Z-1)\right) \times \left(W^{d}(i,Z-1)(p_{2}^{u}(i+1,Z-1) + p_{2}^{d}(i,Z-1)) + D_{s}(i,Z-1)(1-R^{u}(i,Z-1))R^{d}(i+1,Z-1) + D_{b}(i+1,Z-1)(1-R^{u}(i,Z-1)(1-R^{d}(i+1,Z-1)))\right)$$

$$(8.2.62)$$

$$r_2^d(i,Z) = (1 - q^u(i+1,Z))R^d(i+1,Z)$$
 (8.2.63)

$$r_3^a(i,Z) = r_1^a(i,Z)R^a(i+1,Z)$$
(8.2.64)

$$r_{1}^{u}(i+1,Z) = 1/\left(W^{u}(i+1,Z-1) + D_{s}(i,Z-1) + D_{b}(i+1,Z-1) + X_{1}^{d}(i,Z-1)\right) \times \left(W^{u}(i+1,Z-1)(p_{2}^{u}(i+1,Z-1) + p_{2}^{d}(i,Z-1)) + D_{s}(i,Z-1)(1-R^{d}(i,Z-1))r_{i+1} + D_{b}(i+1,Z-1)(1-R^{d}(i+1,Z-1))r_{i+1} + D_{b}(i+1,Z-1)r_{i+1}(1-R^{d}(i+1,Z-1))\right)$$

$$(8.2.65)$$

$$r_{2}^{u}(i+1,Z) = (1-q^{d}(i,Z))R^{d}(i,Z)$$

$$(8.2.66)$$

$$r_3^u(i+1,Z) = r_1^d(i,Z)R^u(i,Z)$$
 (8.2.67)

Failure mode changes

$$z_{1,2}^d(i,Z) = 0 (8.2.68)$$

$$z_{1,3}^d(i,Z) = 0 (8.2.69)$$

$$z_{2,3}^d(i,Z) = q^u(i+1,Z)(1-R^d(i+1,Z))$$
 (8.2.70)

$$z_{2,1}^d(i,Z) = q^u(i+1,Z)R^d(i+1,Z)$$
 (8.2.71)

$$z_{3,1}^d(i,Z) = (1 - r_1^d(i,Z))R^d(i+1,Z)$$
 (8.2.72)

$$z_{3,2}^d(i,Z) = r_1^d(i,Z)(1 - R^d(i+1,Z))$$
(8.2.73)

$$z_{1,2}^u(i+1,Z) = 0 (8.2.74)$$

$$z_{1,3}^u(i+1,Z) = 0 (8.2.75)$$

$$z_{2,3}^{u}(i+1,Z) = q^{d}(i,Z)(1-R^{u}(i,Z))$$

$$z_{2,1}^{u}(i+1,Z) = q^{d}(i+1,Z)R^{u}(i,Z)$$
(8.2.77)
(8.2.77)

$$z_{2,1}^{u}(i+1,Z) = q^{d}(i+1,Z)R^{u}(i,Z)$$
(8.2.77)

$$z_{3,1}^{u}(i+1,Z) = (1 - r_1^{u}(i+1,Z))R^{u}(i,Z)$$
(8.2.78)

$$z_{3,2}^{u}(i+1,Z) = r_1^{u}(i+1,Z)(1-R^{u}(i,Z))$$
(8.2.79)

Interruption of flow

$$p_{1}^{d}(i,Z) = \frac{1}{P_{s}(i+1,Z-1) + P_{b}(i,Z-1)} \times$$

$$\left(P_{s}(i+1,Z-1)\left(R^{u}(i,Z-1)q^{d}(i,Z-1) + R^{u}(i,Z-1)(1-q^{d}(i,Z-1))\right) + (1-R^{u}(i,Z-1))q^{d}(i,Z-1)\right) + (1-R^{u}(i,Z-1))q^{d}(i,Z-1)\right) + (1-q^{u}(i+1,Z-1)(1-R^{d}(i+1,Z-1)) + (1-q^{u}(i+1,Z-1))R^{d}(i+1,Z-1) + q^{u}(i+1,Z-1)R^{d}(i+1,Z-1)\right)\right)$$

$$p_{2}^{d}(i,Z) = \frac{1}{W^{d}(i,Z-1)} \times$$

$$\left(X_{2}^{d}(i,Z)\left(R^{d}(i+1,Z) + q^{u}(i+1,Z)(1-R^{d}(i+1,Z))\right) - X_{3}^{d}(i,Z)r_{1}^{d}(i,Z)(1-R^{d}(i+1,Z))\right)\right)$$

$$p_{3}^{d}(i,Z) = 0 \qquad (8.2.82)$$

$$p_3^d(i,Z) = 0$$

$$p_{1}^{u}(i+1,Z) = \frac{1}{P_{s}(i,Z-1) + P_{b}(i+1,Z-1)} \times$$

$$\begin{pmatrix} P_{s}(i,Z-1) \left(R^{d}(i+1,Z-1)q^{u}(i+1,Z-1) \right) \\ + R^{d}(i+1,Z-1)(1-q^{u}(i+1,Z-1)) \\ + (1-R^{d}(i+1,Z-1))q^{u}(i+1,Z-1) \right) \\ + P_{b}(i+1,Z-1) \left(q^{d}(i,Z-1)(1-R^{u}(i,Z-1)) \\ + (1-q^{d}(i,Z-1))R^{u}(i,Z-1) + q^{d}(i,Z-1)R^{u}(i,Z-1) \right) \right) \\ p_{2}^{u}(i+1,Z) = \frac{1}{W^{u}(i+1,Z-1)} \times$$

$$(8.2.84)$$

$$\left(X_{2}^{u}(i+1,Z) \left(R^{u}(i,Z) + q^{d}(i,Z)(1-R^{u}(i,Z)) \right) \right)$$

$$\begin{pmatrix} X_2^u(i+1,Z) \begin{pmatrix} R^u(i,Z) + q^u(i,Z)(1 - R^u(i,Z)) \end{pmatrix} \\ -X_3^u(i+1,Z)r_1^u(i+1,Z)(1 - R^u(i,Z)) \end{pmatrix}$$

$$p_3^u(i+1,Z) = 0$$
(8.2.85)

Idleness failure

$$q_1^d(i,Z) = \frac{P_b(i,Z-1)R^d(i,Z-1) + P_s(i-1,Z-1)R^u(i-1,Z-1)}{P_s(i-1,Z-1) + P_b(i,Z-1)}$$
(8.2.86)

$$q_2^d(i,Z) = 0 (8.2.87)$$

$$q_3^d(i,Z) = 0 (8.2.88)$$

$$q_1^u(i+1,Z) = \frac{P_b(i+1,Z-1)R^d(i+1,Z-1) + P_s(i,Z-1)R^u(i,Z-1)}{P_s(i,Z-1) + P_b(i,Z-1)}$$
(8.2.89)

$$q_2^u(i+1,Z) = 0 (8.2.90)$$

$$q_3^u(i+1,Z) = 0 (8.2.91)$$

Chapter 9

Algorithm and Numerical Behavior

Once the decomposition equations are constructed, an algorithm that will solve the equations needs to be developed. This chapter presents an algorithm to solve the decomposition equations. Then it provides numerical data for determining the accuracy of the algorithm. Although no mathematical proof of convergence is presented in this thesis, Numerical experiments show that the developed algorithm converges well for lines with reasonable line parameters. Also, it always converges to the same values, regardless of the starting point.

9.1 Algorithm

This section presents an algorithm for solving the decomposition equations derived in Chapter 8. The algorithm presented in the thesis is based on the DDX algorithm which won first developed for a single-part-type production line introduced in [6]. The new algorithm is distinguished from the previous one since it consists of two separate iterative loops. The inner loop sweeps down the line within a part type, using parameter values from previous iterations within a part type to calculate parameters at the current iteration, and then sweeps up the line in the reverse direction. The outer loop iteration sweeps up the line between part types, using parameter values from previous iteration between part types. The inner loop termination criterion is considered met when Conservation of Flow holds among all two-machine lines to within a certain tolerance. This method exploits the recursive nature of the interruption and resumption of flow equations.

Step 0: Initialization

Initialize upstream parameters for i = 2 to NumMachines for j = 1 to NumParts $p_1^u(i,j) = p_2^u(i,j) = p_i$ $r_1^u(i,j) = r_2^u(i,j) = r_3^u(i,j) = r_i$ $q^u(i,j) = p_i$ $z_{2,1}^u(i,j) = z_{2,3}^u(i,j) = z_{3,2}^u(i,j) = z_{3,1}^u(i,j) = p_i$ end;

end;

Initialize downstream parameters

for i = 1 to NumMachines-1

for j = 1 to NumParts

$$p_1^d(i,j) = P_2^d(i,j) = p_i$$

 $r_1^d(i,j) = r_2^d(i,j) = r_3^d(i,j) = r_i$
 $q^d(i,j) = p_i$
 $z_{2,1}^d(i,j) = z_{2,3}^d(i,j) = z_{3,2}^d(i,j) = z_{3,1}^d(i,j) = p_i$

end;

end;

Initialize boundary conditions

for j = 1 to NumParts

$$p^{u}(1, j) = p_{0,j}$$

$$r^{u}(1, j) = r_{0,j}$$

$$q^{u}(1, j) = 0$$

$$p^{d}(NumMachines + 1, j) = p_{NumMachines + 1,j}$$

$$r^{d}(NumMachines + 1, j) = r_{NumMachines + 1,j}$$

$$q^{d}(NumMachines + 1, j) = 0;$$

end;

While (Step C.1 criterion is not met) do Step 1 through Step J

Outer loop iteration

While (Step C.2 criterion is not met) do Step 1.1 and Step 1.2 Inner loop iteration

Step 1.1: Upstream Sweep for Type 1

for i = 1 to NumMachines Evaluate Two Machine Line L(i - 1, 1)Calculate $p_1^u(i, 1)$ and $p_1^u(i, 1)$ Calculate $r_1^u(i, 1), r_2^u(i, 1)$ and $r_3^u(i, 1)$ Calculate $z_{2,1}^u(i, 1), z_{2,3}^u(i, 1), z_{3,1}^u(i, 1)$ and $z_{3,2}^u(i, 1)$

end

Step 1.2: Downstream Sweep for Type 1

for i = NumMachines-1 to 1 Evaluate Two Machine Line L(i + 1, 1)Calculate $p_1^d(i, 1)$ and $p_1^u(i, 1)$ Calculate $r_1^d(i, 1), r_2^u(i, j)$ and $r_3^u(i, 1)$ Calculate $z_{2,1}^d(i, 1), z_{2,3}^u(i, 1), z_{3,1}^u(i, 1)$ and $z_{3,2}^u(i, 1)$

 \mathbf{end}

end

for j=2 to NumParts-1

While (Step 5 criterion is not met) do Step j.1 and Step j.2

Step j.1: Upstream Sweep for Type j

for i = 1 to NumMachines

Evaluate Two Machine Line L(i-1, j)Calculate $p_1^u(i, j)$ and $p_1^u(i, j)$ Calculate $r_1^u(i, j), r_2^u(i, j)$ and $r_3^u(i, j)$ Calculate $z_{2,1}^u(i, j), z_{2,3}^u(i, j), z_{3,1}^u(i, j)$ and $z_{3,2}^u(i, j)$ Calculate q(i, j - 1)end Step j.2: Downstream Sweep for Type jfor i = NumMachines-1 to 1 Evaluate Two Machine Line L(i + 1, j)Calculate $p_1^d(i, j)$ and $p_1^u(i, j)$ Calculate $r_1^d(i, j), r_2^u(i, j)$ and $r_3^u(i, j)$ Calculate $z_{2,1}^d(i, j), z_{2,3}^u(i, j), z_{3,1}^u(i, j)$ and $z_{3,2}^u(i, j)$ Calculate q(i, j - 1)end;

Step C.1: Evaluate Inner Loop Stopping Criterion

Terminate the inner loop when the maximum value of

$$||E(i,j) - E(0,j)||$$

for i = 1, ... NumMachines is less than some pre-specified ϵ for each part type j.

Step C.2: Evaluate Outer Loop Stopping Criterion

Terminate the algorithm when the maximum value of

$$||E^{(m-1)}(i,j) - E^{(m)}(i,j)||$$

where i = 1, ..., NumMachines and j = 1, ..., NumParts and (m) is the outer loop iteration repetition number, is less than some pre-specified δ .

9.2 Algorithm Behavior

It is observed that the algorithm seems to converge most reliably when the mean time to fail (MTTF) and mean time to repair (MTTR) of all the machines in the line, including supply, processing, and demand machines are of the same order of magnitude as each other, and the MTTR one order of magnitude less than the MTTF. It is found that the algorithm is does not converge reliably when the MTTF and MTTR are radically different from one machine to the next, and if the algorithm does converge in these cases, it is usually not accurate. In physical terms, this corresponds to cases where adjacent machines fail and are repaired at radically different rates. It is found that, if the algorithm converges, it takes about five to ten iterations for the inner loop while it take less then three iterations for the our loop.

9.3 Randomly Generated Cases

It is hard to provide a mathematical proof of convergence of the algorithm, we follow the test procedure of Burman [3] to verify its reliability and accuracy. In the test, multiple randomly generate cases, where the parameters of the random systems are within certain pre-set tolerances, are evaluated with the algorithm. The results from the algorithm are compared with those from discrete-event simulations. This section presents the accuracy of the algorithm with respect to the simulation results for production rate and average buffer levels.

The randomly generated cases have machines that have similar, though not identical, characteristics. The machines are allowed to have different isolated efficiencies, but we do not generate lines where one machine is an extreme bottleneck. Note that the isolated production rate or isolated machine efficiency of an unreliable machine is:

$$e_i = \frac{r_i}{r_i + p_i}$$

In general, the isolated efficiencies of the supply and processing machines vary from

0.8 and 1. For the demand machines, repair probabilities are generated such that they are of the same order of magnitude as those generated for the supply and processing machines, but the failure probabilities of the demand machines are higher; they are of the same order of magnitude as the repair probabilities. This ensures that the demand rate for each part type is below the capacity of the line. This is because a system in which the demand machine for the highest priority has an isolated efficiency similar to that of the other machines in the line tends to be uninteresting, as the line spends all of its time producing type one products, and none producing lower priority parts. For example, a two-part-type processing line with eight processing machines behaves like a single-part transfer line with ten machines (where the additional machines are the supply machine upstream, and the demand machine downstream). Moreover, it is ensured that the combined rate for the demand machines for all the types is less than the capacity of the line. The reason for this is that if the line does have the capacity to meet demand for all the part types, then the estimation process is trivial; production rate will equal demand, and all intermediate buffers will be nearly full. Therefore, we ensure that

$$e_{d,j} < \min(e_i) < \sum_j e_{d,j}$$

where $e_{d,j}$ is the isolated demand rate for Type j.

9.4 Computation of Approximation Error

Each simulation consisted of 10 independent simulations runs of 1,500,000 time periods each, where the first 500,000 time periods were discarded to ensure data was only collected on a system in steady state. The length of the simulation runs was chosen so that the transient period did not affect the results. The performance measures simulated were the production rate for each part type, and the average inventory level for each type in all buffers. Approximate 95% confidence intervals (see [2, 9]) were calculated for all performance measures. For production rates, the percent error of the approximated production rate from the simulated production rate is evaluated in the following manner.

$$\% \text{ Error} = 100 \times \frac{E_{decomp} - E_{sim}}{E_{sim}}.$$
 (9.4.1)

This metric is standard in the literature, and provides easy recognition of whether the approximation is under- or over-estimating the simulated production rate.

For average buffer levels, the percent error of the approximated average buffer level from that of the simulated buffer level is calculated using a metric that took the difference between the approximated average buffer level (\overline{N}_{decomp}) and the simulated average buffer level (\overline{N}_{sim}), divided the difference by the buffer size (N), and multiplied the quotient by 100. In other words,

% Error =
$$100 \times \frac{\overline{N}_{decomp} - \overline{N}_{sim}}{N}$$
. (9.4.2)

This measurement is standard in the literature cited as well.

9.5 Numerical Results for Two-Part-Type Lines

Production lines with five processing machines processing two part types were test with the algorithm. For the test, 300 random lines are generated. The first 100 random lines are lines where demand for Type 1 and Type 2 are roughly the same. The second 100 cases are of the line where the demand for part Type 1 exceeds that of Type 2 by up to 30%. The remaining 100 cases are of the line where the demand for part Type 2 exceeds that of part Type 1 by up to 30%. Buffers size vary from 5 to 20. For the machines modeling the supply process, and the processing machines, the repair probabilities were generated from a tight triangular distribution with mean 0.1 and a range of values from 0.08 to 0.12. The failure probabilities (the p parameters) were generated from a tight triangular distribution with mean 0.01



Figure 9.5.1: The errors in the decomposition approximation for Type 1 production rates

and a range of values from 0.05 to 0.015. Using these values gives an average isolated efficiency of approximately 0.91, taking values ranging from 0.8 to 1.0. The repair and failure probabilities for the demand machines were generated from similar triangular distributions, and are summarized in Table 9.6. There, \bar{e} refers to the average isolated production rate of the supply machines and processing machines, and \bar{e}_{d1} and \bar{e}_{d2} are the average demand rates for part types one and two, respectively.

	Case 1-100	Case 101-200	Case 201-300
\overline{e}	0.91	0.91	0.91
\overline{e}_{d1}	0.55	0.55	0.4
\overline{e}_{d2}	0.55	0.4	0.55

=

Table 9.5.1: Machine parameters for the randomly generated two-part type processing lines.

The percent errors calculated for all 300 cases are shown in Figure 9.5.1 and 9.5.2. The average absolute errors for Type 1 is -0.52% while Type 2 is 2.2%. The



Figure 9.5.2: The errors in the decomposition approximation for Type 2 production rates

average errors for buffer levels are 7.3% and 8.2% for Type 1 and Type 2, respectively. As shown in the figures, the algorithm tends to slightly underestimate the Type 1 production rate, while overestimating for Type 2 parts.

9.6 Numerical Results for Three-Part Type Lines

Production lines with five processing machines processing three-part types is tested with the algorithm. For the test, 400 random lines are generated. The first 100 random lines are lines where demand for part Type 1, Type 2, and Type 3 are roughly the same. The second 100 cases are of the line where the demand for Type 2 and Type 3 are roughly the same while the demand for part Type 1 exceeds those of Type 2 and Type 3 by up to 30%. For the next 100 cases, the demand for Type 1 and Type 3 are roughly the same while the demand for part Type 2 exceeds those of Type 1 and Type 3 by up to 30%. The remaining 100 cases are of the line where the demand for Type 1 and Type 2 are roughly the same the demand for part Type

	Percent Error
Production rate for Type 1	-0.52
Production rate for Type 2	2.2
Average Buffer Level for Type 1	7.3
Average Buffer Level for Type 2	8.2

Table 9.5.2: Absolute percent errors for production rates and average buffer levels for the two-part type processing machines.

3 exceeds that of part Type 1 by up to 30%. Buffers size vary from 5 to 20 for all the cases. For the machines modeling the supply process, and the processing machines, the repair probabilities were generated from a tight triangular distribution ranging of values from 0.08 to 0.12. The failure probabilities (the p parameters) were generated from a tight triangular distribution with mean 0.01 and a range of values from 0.05 to 0.015. Using these values gives an average isolated efficiency of approximately 0.91, taking values ranging from 0.8 to 1.0. The repair and failure probabilities for the demand machines were generated from similar triangular distributions, and are summarized in Table 9.6.

	Case 1-100	Case 101-200	Case 201-300	Case 301-400
\overline{e}	0.91	0.91	0.91	0.91
\overline{e}_{d1}	0.4	0.4	0.3	0.3
\overline{e}_{d2}	0.4	0.3	0.4	0.3
\overline{e}_{d3}	0.4	0.3	0.3	0.4

Table 9.6.1: Machine parameters for the randomly generated three-part type processing lines.

The percent errors calculated for all 200 cases are shown in Figure 9.6.1, 9.6.2 and 9.6.3. The average absolute errors for Type 1 is -3.7%, Type 2 is 1.7%, and Type 3 is -8.7% The average errors for buffer levels are 8.5%, 7.4%, and 11.2%, for Type 1, Type 2, and Type 3 respectively. As shown in the figures, the algorithm tends to slightly underestimate the Type 1 and Type 3 production rate, while overestimating for Type 2 parts. It is interesting to note that the Type 2 results are the most accurate. More



Figure 9.6.1: The errors in the decomposition approximation for Type 1 production rates $% \left({{{\bf{r}}_{\rm{T}}}} \right)$



Figure 9.6.2: The errors in the decomposition approximation for Type 2 production rates



Figure 9.6.3: The errors in the decomposition approximation for Type 2 production rates

	Percent Error
Production rate for Type 1	-3.7
Production rate for Type 2	1.7
Production rate for Type 3	-8.7
Average Buffer Level for Type 1	8.5
Average Buffer Level for Type 2	7.4
Average Buffer Level for Type 3	11.2

Table 9.6.2: Absolute percent errors for production rates and average buffer levels for the three-part type processing machines.

research needs to be conducted to investigate this behavior.

Chapter 10

System Behavior

In order to understand the system behavior of the multiple-part-type line, system sensitivity analysis were performed with a line processing three different part types. This line consists of one processing machine, three demand machines, and three supply machines. All buffers are finite and machines are unreliable. Although the line has only one processing machine, it is long enough to capture the interesting dynamic behavior of the system presented in a multiple-part-type line. The processing machine in the line is denoted by M_1 , and the supply and demand machines are denoted by $B_{0,j}$ and $B_{1,j}$. The line processes three different part types (Z = 3) and Type 1 (j = 1) is the highest priority part type, while Type 3 (j = 3) is the lowest priority part type. With our priority rule, the processing machine M_1 works on a Type 1 part when it finds a part in the Type 1 upstream buffer $B_{0,1}$ and finds space in the Type 1 downstream buffer, $B_{1,1}$. If not, it switches to the next priority part type, Type 2, it switches to the lowest priority part type, Type 3. The throughput of Type j is denoted by E_j .

10.1 Changing Demand

We first investigate the system behavior of the line with a changing demand rate. We assume that the supply rates are high enough that the processing machine is rarely starved for input material. By choosing the high supply rates, we can concentrate on the effect of varying demand rates. Moreover, if we consider the processing machine as a flexible production line processing multiple products, it is more realistic to assume that demand rates have higher uncertainty compared to supply rates.

Let us consider the following hypothetical example. The machine in the line M_1 represents a production line processing three different sedans — a luxury sedan, a full size sedan, and an economy size sedan. The line is fully automated and flexible so that it can switch production from one model to another at a tiny fraction of the total production cost. The line is operating according to a strict priority rule. Since the luxury model is sold with the highest profit margin, the highest priority is given to the luxury model. If there is no demand for the luxury model, the line produces the full size model, which generates the second highest profit margin. The production of the economy sedan is performed when there is no demand for both the luxury and full size models. In this hypothetical case, the luxury, full size, and economy models are considered Type 1, Type 2, and Type 3, respectively.

Let us consider the case where the demand for Type 1 varies as shown in Figure 10.1.1. The corresponding system parameters are shown in Table 10.1.

Machines		Buffers	
$e_{0,1}$	0.91	$B_{0,1}$	15
$e_{0,2}$	0.91	$B_{0,2}$	15
$e_{0,3}$	0.91	$B_{0,3}$	15
$e_{1,1}$	Varying	$B_{1,1}$	15
$e_{1,2}$	0.5	$B_{1,2}$	15
$e_{1,3}$	0.5	$B_{1,3}$	15
e_1	0.833		

Table 10.1.1: Machine and buffer parameters for Figure 10.1.1

The supply machine $(M_{0,j})$ parameters are chosen in such a way that each supply machine has an isolated production rate of 0.91. The isolated production rate of the processing machine M_1 is 0.833. The isolated production rates of the demand



Figure 10.1.1: Throughput vs Demand for Type 1

machines for Type 2 and Type 3 are 0.5 and 0.5, respectively. The demand rate for Type 1 varies from 0.08 to 0.85. The sizes of the buffers are identical.

Changing rate for Type 1

In Figure 10.1.1, lines represent the throughput of a part type in response to the changing demand rate for Type 1. As shown in the figure, the production rate for Type 1 increases linearly at an almost a 45 degree angle. This is because Type 1 is the highest priority part and the processing machine always tries to meet the demand for Type 1 whenever possible. Therefore, as the demand for Type 1 increases, the processing machine spends more time to meet the demand for Type 1. However, as the Type 1 demand rate increases, the throughput rates for Type 1 and Type 2 decrease. Since the processing machines needs to spend more time on processing Type 1, it spends less time on processing Type 2 and Type 3. Note that the throughput of Type 2 is always higher than that of Type 3 due to our priority rule.

Machines		Buffers	
$e_{0,1}$	0.91	$B_{0,1}$	15
$e_{0,2}$	0.91	$B_{0,2}$	15
$e_{0,3}$	0.91	$B_{0,3}$	15
$e_{1,1}$	0.14	$B_{1,1}$	15
$e_{1,2}$	varying	$B_{1,2}$	15
$e_{1,3}$	0.33	$B_{1,3}$	15
e_1	0.83		

Table 10.1.2: Machine and buffer parameters for Figure 10.1.2

Changing rate for Type 2

Figure 10.1.2 illustrates the system behavior as the Type 2 demand rate varies. The system parameters are shown in Table 10.1. The demand rates for Type 1 and Type 3 are fixed at 0.14 and 0.33, respectively. In the figure, the throughput rate for Type 2 increases linearly as the demand rate for Type 2 increases. However, the throughput rate for Type 1 remains constant. Since Type 1 has priority over Type 2, the increasing demand rate for Type 2 does not affect the production of Type 1 parts. However, if the machine spends more time on processing Type 2, it will spend less time on processing Type 3. This is why the throughput rate for Type 3 drops as the demand rate for Type 2 increases.

Machines		Buffers	
$e_{0,1}$	0.91	$B_{0,1}$	15
$e_{0,2}$	0.91	$B_{0,2}$	15
$e_{0,3}$	0.91	$B_{0,3}$	15
$e_{1,1}$	0.25	$B_{1,1}$	15
$e_{1,2}$	0.25	$B_{1,2}$	15
$e_{1,3}$	Varying	$B_{1,3}$	15
e_1	0.833		

Table 10.1.3: Machine and buffer parameters for Figure 10.1.3



Figure 10.1.2: Throughput vs Demand for Type 2

Changing rate for Type 3

The next case is the system behavior due to the variation of the Type 3 demand rate shown in Figure 10.1.3. In this case, the demand rates for Type 1 and Type 2 are fixed, while the demand rate for Type 3 varies. In the figure, the throughput rate for Type 3 increases in response to the increasing demand rate for Type 3. However, the throughput rates for Type 1 and Type 2 remain constant. This is because Type 1 and Type 2 are higher priority part types, so they are not affected by the demand rate of the lower priority part. Note that the demand rates for Type 1 and Type 2 are 0.25. However, their throughput rates, E_1 and E_2 , are less than this demand rates. This is because the size of the buffer is finite and not large enough to absorb the random behavior of the line. Also note that, due to the priority rule, the Type 1 throughput is slightly higher than the Type 2 throughput rate.

Another interesting behavior is observed in Figure 10.1.3. The throughput rate of Type 3 increases almost linearly for a while, but the rate of the increase drops slowly and gets saturated as the demand rate further increases. Note that the isolated



Figure 10.1.3: Throughput vs Demand for Type 3

production rate of the processing machine, e_1 , is 0.833. Let us call this isolated production rate *line capacity*. When the demand rate for Type 3 is low, the total demand rate is less than the line capacity. That is,

$$e_{0,1} + e_{0,2} + e_{0,3} < e_1$$

In this case, the processing machine is under-utilized, and there is room to accommodate the increasing demand rate for the lowest priority part type. However, as the demand rate further increases, the processing machine gets to the line capacity and the throughput rate becomes saturated to the limit.

Machines		Buffers	
$e_{0,1}$	0.75	$B_{0,1}$	20
$e_{0,2}$	0.75	$B_{0,2}$	20
$e_{0,3}$	0.75	$B_{0,3}$	20
$e_{1,1}$	0.28	$B_{1,1}$	20
$e_{1,2}$	0.28	$B_{1,2}$	20
$e_{1,3}$	0.28	$B_{1,3}$	20
e_1	Varying		

Table 10.2.1: Machine and buffer parameters for Figure 10.2.1

10.2 Changing Processing Machine Capacity and Buffer Size

Changing machine capacity

This section investigates the system behavior due to the varying line capacity. The demand rates are fixed as shown in Table 10.2. The rate of the processing machine e_1 varies from 0.4 to 98. Figure 10.2.1 illustrates the system behavior resulting from changing e_1 . Note that since a machine cannot produce more than its capacity, the combined throughput rate must be less then the line capacity, That is,

$$E_1 + E_2 + E_3 < e_1$$

As the line capacity increases, the throughput rate for each part type also increases. Note that once the line capacity, e_1 , becomes larger than the Type 1 demand rate,

$$e_1 > e_{1,1}$$

the increase in the line capacity does not influence the Type 1 throughput rate much, because the processing machine already has a sufficient rate to meet the highest priority part type. Likewise, once the line capacity is larger than the combined Type



Figure 10.2.1: Throughput vs Capacity of M_1

1 and Type 2 demand rates,

$$e_1 > e_{1,1} + e_{1,2}$$

the increase in the line capacity does not impact the both throughput rate. The processing machine has sufficient capacity to meet the demand for both part types. In the case shown in Figure 10.2.1, e_1 varies from 0.4 to 0.98. The smallest value of e_1 in the figure, which is 0.4, is less than $e_{1,1} + e_{1,2}$. At this capacity, the Type 1 demand has almost met already and therefore, as the further increase in the capacity does not influence E_1 , but the increase does influence E_2 . Once both E_1 and E_1 come close to their corresponding demand rates, the Type 3 throughput rate begins to increase faster. Once all the demand rates are met, the throughput rates do not increase and they get saturated at their corresponding demand rates.

Changing buffer size

Figure 10.2.2 shows the throughput rate for each part type in response to the variation of the size of buffer $B_{1,2}$. The size of the buffer changes from 5 to 15. As shown in

Machines		Buffers	
e _{0,1}	0.99	$B_{0,1}$	5
$e_{0,2}$	0.99	$B_{0,2}$	5
$e_{0,3}$	0.99	$B_{0,3}$	5
$e_{1,1}$	0.25	$B_{1,1}$	10
$e_{1,2}$	0.33	$B_{1,2}$	Varying
$e_{1,3}$	0.33	$B_{1,3}$	10
e_1	0.91		

Table 10.2.2: Machine and buffer parameters for Figure 10.2.2

Table 10.2, the total demand rate is close to the rate of the processing machine. That is,

$$e_{1,1} + e_{1,2} + e_{1,3} \approx e_1$$

In general, the size of a buffer is the most influential when the upstream and downstream machines have similar isolated processing rates. In the figure, as $N_{1,2}$ increases, E_2 increases while E_3 decreases. When the size of the buffer $B_{1,2}$ gets bigger, M_1 becomes less blocked for Type 2 and therefore spends more time processing Type 2, Consequently, it spends less time on Type 3 and E_3 decreases. However, Type 1 is not affected since it is a higher priority part type. Note that the Type 2 throughput rate is saturated at 0.33 because the processing machine does not process parts more than the demand rate for Type 2.



Figure 10.2.2: Throughput vs Size of

Chapter 11

Extension: Re-entrant Line

In this chapter we present a decomposition for a re-entrant production line. In developing equations for the building blocks for the re-entrant production line, we modify the decomposition model that has been created for the multiple-part-type line. Semiconductor device and liquid crystal display (LCD) fabrication processes are characterized as re-entrant processes, in which a similar sequence of processing step is always repeated several times.

11.1 Re-entrant Production Line

Figure 11.1.1 represents a re-entrant production line with a single re-entrant loop. In this example, a supply machine, M_0 and a demand machine, $M_{K+1,1}$ are placed at the beginning and end of the line. Once parts enter the line through the supply machine, they are first processed by machines from M_1 to M_K . During these processing steps, parts are stored in buffers $B_{i,2}$, $i = 0 \dots K$. Then they are processed by the *reentrant machine*, $M_{K+1,2}$. The function of this machine is to send parts back to



Figure 11.1.1: Re-entrant production line model
machine M_1 so that the parts can go through the same processing steps again. This re-entrant machine can be either an actual machine or an imaginary machine that logically creates the re-entrant loop.

11.2 Decomposition Equations

Note that in Figure 11.1.1 the configuration of the re-entrant line is similar to that of the production line processing multiple part types. The only difference is that the Type j flow is sent back as Type j - 1 through $M_{K+1,j}$ and $M_{0,j-1}$. Therefore, all the decomposition equations developed for the multiple part type line can be applied to the re-entrant line with modifications. Suppose that in the two-part type production line, the parameters for the demand machine for Type 2, $M_{k+1,2}$, are given such that the machine imitates the flow behavior of Type 1 part in the line. Also, at the same time, the parameters for the supply machine for Type 1, $M_{0,1}$, are assigned such that the machine imitates the flow behavior of Type 2 parts in the line. Then we can require that the flow rate for Type 1 is equal to that of Type 2.

$$E_1 = E_2$$
 (11.2.1)

If we apply the above approach to the decomposition method for the re-entrant line, all the decomposition equations are the same as those constructed in two-part type line except the decomposition equations for L(k, 2) and L(0, 1).

Interruption of flow

For the interruption of flow for $M^{u}(0,1)$, we use the balance equation:

$$\sum_{i=1}^{3} Ps_i(K,2)r_i^u(K,2) = W^d(K-1,2)p_*^d$$
(11.2.2)

,where p_*^d is the probability that $M_{K+1,2}$ becomes starved due to any machine failure upstream of $B_{K,2}$. Then

$$p_1^u(0,j) = p_*^d + p_{K+1,j+1}$$

= $\frac{1}{W^d(1,j+1)} \sum_{i=1}^3 Ps_i(K,j+1)r_i^u(K,j+1) + p_{K+1,2}$ (11.2.3)

Similarly, for $M^d(2,2)$

$$\sum_{i=1}^{3} Pb_i(0,j)r_i^d(0,j) = W^u(1,j)p_*^u$$
(11.2.4)

$$p_{1}^{d}(K, j+1) = p_{*}^{u} + p_{K+1,j+1}$$

= $\frac{1}{W^{u}(1,j)} \sum_{i=1}^{3} Pb_{i}(0,j)r_{i}^{d}(0,j) + p_{K+1,j+1}$ (11.2.5)

Resumption of flow

Flow rate idle time is used for the derivations of the resumption of flow equations.

$$E = e_{K+1,j+1} \left(1 - \widetilde{Ps}(K,j+1) - \widetilde{Pb}(0,j) \right)$$
(11.2.6)

where $e_{K+1,j+1} = \frac{r_{K+1,j+1}}{r_{K+1,j+1}+p_{K+1,j+1}}$, $\widetilde{Ps} = \sum_{i=1}^{3} Ps_i$ and $\widetilde{Pb} = \sum_{i=1}^{3} Pb_i$. Also we know that

mow that

$$E^{u}(0,j) = e^{u}(0,j) \left(1 - \widetilde{Pb}(0,j)\right)$$
$$E^{d}(K,j+1) = e^{d}(K,j+1) \left(1 - \widetilde{Ps}(K,j+1)\right)$$

These can be written

$$\widetilde{Ps}(K, j+1) = 1 - \frac{E^d(K, j+1)}{e^d(K, j+1)}$$

$$\widetilde{Pb}(0,j) = 1 - \frac{E^u(0,j)}{e^u(0,j)}$$

Then (11.2.6) becomes

$$E = e_{K+1,j+1} \left(\frac{E^d(K,j+1)}{e^d(K,j+1)} + \frac{E^u(0,j)}{e^u(0,j)} - 1 \right)$$

or since $E = E^d(K, j + 1) = E^u(1, j)$,

$$1 = e_{K+1,j+1} \left(\frac{1}{e^d(K,j+1)} + \frac{1}{e^u(0,j)} - \frac{1}{E} \right)$$
(11.2.7)

We know that

$$\frac{1}{e^u(0,j)} = \frac{p^u(0,j) + r^u(0,j)}{r^u(0,j)}$$
$$\frac{1}{e^d(K,j+1)} = \frac{p^d(K,2) + r^d(K,j+1)}{r^d(K,j+1)}$$

Then (11.2.7) becomes

$$1 = e_{K+1,j+1} \left(\frac{p^u(0,j) + r^u(0,j)}{r^u(0,j)} + \frac{p^d(K,j+1) + r^d(K,j+1)}{r^d(K,j+1)} - \frac{1}{E} \right)$$

$$1 = e_{K+1,j+1} \left(\frac{p^u(0,j)}{r^u(0,j)} + \frac{p^d(K,j+1)}{r^d(K,j+1)} - \frac{1}{E} + 2 \right)$$

That is,

$$\frac{1}{e_{K+1,j+1}} + \frac{1}{E} - 2 = \frac{p^u(0,j)}{r^u(0,j)} + \frac{p^d(K,j+1)}{r^d(K,j+1)}$$
(11.2.8)

Two equation are introduced

$$I^{u}(0,1) = \frac{p^{u}(0,j)}{r^{u}(0,j)} \quad \text{and} \quad I^{d}(K,j+1) = \frac{p^{d}(K,j+1)}{r^{d}(K,j+1)} \quad (11.2.9)$$

Then we can rewrite (11.2.8) such that,

$$I^{u}(0,j) = \frac{1}{E^{d}(K,j+1)} + \frac{1}{e_{K+1,j+1}} - I^{d}(K,j+1) - 2 \quad (11.2.10)$$
$$I^{d}(K,j+1) = \frac{1}{E^{u}(0,j)} + \frac{1}{e_{K+1,j+1}} - I^{u}(0,j) - 2$$



Figure 11.3.1: Simple re-entrance production line model

11.3 Numerical Results

In order to verify the analytic equations derived in the previous section, we compare the numerical results of a small system with four machines and four buffers with simulations. The small system is shown in Figure 11.3.1. Two separate cases are presented. For both cases, machine parameters of M_4 vary, while the rest of machine parameters remain constant. We examine the response of the production rate of the system to the varying parameters and compare the results with simulations.

11.3.1 Case1: Varying p_4 and r_4 (e_4 constant)

The system parameters are shown in Table 11.3.1. For this case, we increase the failure rate of M_4 from 0.3 to 0.52. At the same time we vary the repair rate of M_4 to keep the isolated production rate of M_4 constant at 0.48. The rest of the parameters are unchanged. The results of this case are shown in Figure 11.3.2. In the figure, the

Machine	Parameter	Value	Isolated Prod Rate
M_1	r_1	0.48	0.48
	p_1	0.52	
M_2	r_2	0.1	0.9091/2
	p_2	0.11	= 0.4545
M_3	r_3	0.48	0.48
	p_3	0.52	
M_4	r_3	varying	0.48
	p_3	$0.3 \sim 0.52$	

Table 11.3.1: Failure and repair parameters. $(N_1 = N_2 = N_3 = N_4 = 20)$

straight line represents the the production rate of the analytical result and the stars and circles represent the upper and lower bounds of 95% confidence intervals evaluated from simulation runs. As shown in the figure, the production rate of the system is little bit below 0.45. This result matches our expectations, because although the parameters of M_4 are changed, the isolated production rate of the machine remains the same. Also the bottleneck machine of the system is M_2 and therefore the parameter change of the non-bottleneck machine M_4 has little influence the production rate of the system.

From the figure, we can see that the analytical results are within the upper and lower bounds of the 95% confidence intervals. We calculated the percent error of the production rate from the simulated production rate in the following manner.

$$\% Error = 100 \times \frac{E_{analytical} - E_{sim}}{E_{sim}}$$

The result is shown in Figure 11.3.3. As shown in the figure, most of errors are within 1.5% and the maximum error is about 2.5%.

11.3.2 Case 2: Varying p_4 with constant r_4

The system parameters of the second case are shown in Table 11.3.2. In this case, we vary p_4 from 0.1 to 0.8. However, unlike the first case, we fix the value r_4 , therefore,



Figure 11.3.2: Production rate vs. p_4 (e_4 fixed)



Figure 11.3.3: Percent of Error vs. p_4



Figure 11.3.4: Average buffer level vs. p_4

the isolated production rate of M_4 decreases as p_4 increases. The result is of the case is shown in Figure 11.3.5. The production rate of the system is unchanged until p_4 is bigger than 0.58. This is because when p_4 is than 0.58, the bottleneck machine is M_2 and any parameter changes of the non-bottleneck machine do not influence the system production rate. However, if p_4 is bigger than 0.58 the bottleneck machine becomes M_4 and the production rate decreases as the bottleneck machine deceases its capacity. Again, the analytical results are also within the range of the confidence intervals evaluated from the simulation runs.

Figure 11.3.6 shows the percent of error of the production rate of case 2. As shown in the figure, all the errors remain within 3%. Notice that the analytical result tends to over-estimate the production rate when M_2 is bottleneck, while it under-estimate it when M_4 is bottleneck. This behavior should be investigated.

Machine	Parameter	Value	Isolated Prod Rate
M_1	r_1	0.48	0.48
	p_1	0.52	
M_2	r_2	0.1	0.9091/2
	p_2	0.11	= 0.4545
M_3	r_3	0.48	0.48
	p_3	0.52	
M_4	r_3	0.48	varying
	p_3	$0.1 \sim 0.8$	

Table 11.3.2: Failure and repair parameters. $(N_1 = N_2 = N_3 = N_4 = 20)$



Figure 11.3.5: Production rate vs. p_4 (r_4 fixed)



Figure 11.3.6: Percent of Error vs. p_4



Figure 11.3.7: Average buffer level vs. p_4

Chapter 12

Conclusion

In this thesis, we present a model and analysis of a flexible production line with unreliable machines and finite buffers, processing different part types. In the line, the machines operate according to a static priority rule, working on the highest priority part whenever possible, and on lower priority parts only when unable to process those with higher priorities due to either blockage or starvation. We construct decomposition equations to analyze the line and then develop a solution algorithm to solve the decomposition equations. The algorithm converges reliably with reasonable line parameters. Simulations confirmed that the results generated with the decomposition were accurate. System analysis with various parameter shows interesting behaviors of the line.

In the thesis, we also introduce decomposition equations for a re-entrant flow line. We modify the decomposition equations for the multiple-part type production line for the re-entrant system. For verification, the results from the analytical model are compared with results from simulations runs. From the verification, we found that the analytical results were well matched with our intuitions and with results from the simulation.

Based on the decomposition equations, an optimal buffer allocation algorithm for the multiple-part-type line can be developed. Also, the size of buffer can be used to control the flow of each product type, and therefore, this work can be modified to develop a control policy for multiple products. We leave these topics for future research.

Appendix A

Two-Machine One-Buffer Line Transition Equations

A.1 Internal States

The transition equations for the internal states are defined by

$$\begin{split} P(n, \Delta_{j}^{u}, \Delta_{l}^{d}) &= P(n, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \qquad (A.1.1) \\ &+ \sum_{g=1}^{L} P(n, \Delta_{j}^{u}, \Delta_{g}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} P(n, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ P(n, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})p_{l}^{d} \\ &+ \sum_{f=1}^{J} p(n, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}p_{l}^{d} \\ &+ P(n, \Upsilon^{u}, \Delta_{l}^{d})p_{j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} p(n, \Upsilon^{u}, \Delta_{g}^{d})p_{j}^{u}z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} \sum_{g=1}^{L} P(n, \Delta_{f}^{u}, \Delta_{g}^{d})z_{f,j}^{u}z_{g,l}^{d} \\ &+ P(n, \Upsilon^{u}, \Upsilon^{d})p_{j}^{u}p_{l}^{d} \end{split}$$

$$P(n, \Delta_{j}^{u}, \Upsilon^{d}) = \sum_{l=1}^{L} P(n+1, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})r_{l}^{d}$$
(A.1.2)
+
$$\sum_{f=1}^{J} \sum_{l=1}^{L} P(n+1, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}r_{g}^{d}$$
+
$$P(n+1, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - P^{d})$$
+
$$\sum_{f=1}^{L} P(n+1, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}(1 - P^{d})$$
+
$$\sum_{l=1}^{L} P(n+1, \Upsilon^{u}\Delta_{l}^{d})p_{j}^{u}r_{l}^{d}$$
+
$$P(n+1, \Upsilon^{u}\Upsilon^{d})p_{j}^{u}(1 - P^{d})$$

$$P(n, \Upsilon^{u}, \Delta_{l}^{d}) = \sum_{j=1}^{J} P(n-1, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d})$$
(A.1.3)
$$= \sum_{j=1}^{J} \sum_{g=1}^{L} P(n-1, \Delta_{j}^{u}, \Delta_{g}^{d}) r_{j}^{u} z_{g,l}^{d}$$
$$+ \sum_{j=1}^{J} P(n-1, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} p_{l}^{d}$$
$$+ P(n-1, \Upsilon^{u}, \Delta_{l}^{d}) (1 - P^{u}) (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d})$$
$$+ \sum_{g=1}^{L} P(n-1, \Upsilon^{u} \Delta_{g}^{d}) (1 - P^{u}) z_{g,l}^{d}$$
$$+ P(n-1, \Upsilon^{u} \Upsilon^{d}) (1 - P^{u}) p_{l}^{d}$$

$$P(n, \Upsilon^{u}, \Upsilon^{d}) = \sum_{j=1}^{J} \sum_{l=1}^{L} P(n, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} r_{l}^{d}$$
(A.1.4)
+
$$\sum_{j=1}^{J} P(n, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} (1 - P^{d})$$

+
$$\sum_{l=1}^{L} P(n, \Upsilon^{u} \Delta_{l}^{d}) (1 - P^{u}) r_{l}^{d}$$

+
$$P(n, \Upsilon^{u} \Upsilon^{d}) (1 - P^{u}) (1 - P^{d})$$

A.2 Upper Boundary States

The transition equations for the Upper boundary states are the following:

$$\begin{split} P(0, \Delta_{j}^{u}, \Delta_{l}^{d}) &= P(0, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{f=1}^{J} P(0, \Delta_{f}^{u}, \Delta_{l}^{d}) z_{f,j}^{u} (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(0, \Delta_{j}^{u}, \Delta_{g}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u}) z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} \sum_{g=1}^{L} P(0, \Delta_{f}^{u}, \Delta_{g}^{d}) z_{f,j}^{u} z_{g,l}^{d} \\ &+ P(0, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u}) q_{l}^{d} \\ &+ \sum_{f=1}^{J} P(0, \Delta_{f}^{u}, \Upsilon^{d}) z_{f,j}^{u} q_{l}^{d} \end{split}$$

$$P(0, \Delta_{j}^{u}, \Upsilon^{d}) = \sum_{l=1}^{L} P(0, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})r_{l}^{d}$$

$$+ \sum_{l=1}^{L} P(1, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})r_{l}^{d}$$

$$+ \sum_{f=1}^{L} \sum_{l=1}^{L} P(0, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}r_{l}^{d}$$

$$+ \sum_{f=1}^{L} \sum_{l=1}^{L} P(1, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}r_{l}^{d}$$

$$+ P(0, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - Q^{d})$$

$$+ \sum_{f=1}^{L} P(0, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}(1 - Q^{d})$$

$$+ P(1, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - P^{d})$$

$$+ \sum_{f=1}^{L} P(1, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}(1 - P^{d})$$

$$+ \sum_{l=1}^{L} P(1, \Upsilon^{u}, \Delta_{l}^{d})p_{j}^{u}r_{l}^{d}$$

$$+ P(1, \Upsilon^{u}\Upsilon^{d})p_{j}^{u}(1 - P_{l}^{d})$$

$$\begin{split} P(1, \Delta_{j}^{u}, \Delta_{l}^{d}) &= P(1, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \qquad (A.2.3) \\ &+ \sum_{g=1}^{L} P(1, \Delta_{j}^{u}, \Delta_{g}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} P(1, \Delta_{f}^{u}, \Delta_{g}^{d})z_{f,j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{f=1}^{J} \sum_{g=1}^{L} P(1, \Delta_{f}^{u}, \Delta_{g}^{d})z_{f,j}^{u}z_{g,l}^{d} \\ &+ P(1, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{d})p_{l}^{d} \\ &+ \sum_{f=1}^{J} P(1, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}p_{l}^{d} \\ &+ P(1, \Upsilon^{u}, \Delta_{l}^{d})p_{j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(1, \Upsilon^{u}\Delta_{g}^{d})p_{j}^{y}z_{g,l}^{d} \\ &+ P(1, \Upsilon^{u}, \Upsilon^{d})p_{j}^{u}p_{l}^{d} \end{split}$$

$$P(1, \Delta_{j}^{u}, \Upsilon^{d}) = \sum_{l=1}^{L} P(2, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})r_{l}^{d}$$

$$+ \sum_{f=1}^{J} P(2, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}r_{l}^{d}$$

$$+ P(2, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - P^{d})$$

$$+ \sum_{f=1}^{J} P(2, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}(1 - P^{d})$$

$$+ \sum_{l=1}^{L} P(2, \Upsilon^{u}\Delta_{l}^{d})p_{j}^{u}r_{l}^{d}$$

$$+ P(2, \Upsilon^{u}, \Upsilon^{d})p_{j}^{u}(1 - P^{d})$$

$$P(1, \Upsilon^{u}, \Delta_{l}^{d}) = \sum_{j=1}^{J} P(0, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d})$$

$$+ \sum_{j=1}^{J} \sum_{g=1}^{L} P(0, \Delta_{j}^{u}, \Delta_{g}^{d}) r_{j}^{u} z_{g,l}^{d}$$

$$+ \sum_{j=1}^{J} P(0, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} q_{l}^{d}$$
(A.2.5)

$$P(1, \Upsilon^{u}, \Upsilon^{d}) = \sum_{j=1}^{J} \sum_{l=1}^{L} P(0, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} r_{l}^{d}$$

$$+ \sum_{j=1}^{J} \sum_{l=1}^{L} P(1, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} r_{l}^{d}$$

$$+ \sum_{j=1}^{J} P(0, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} (1 - Q^{d})$$

$$+ \sum_{j=1}^{J} P(1, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} (1 - P^{d})$$

$$+ \sum_{l=1}^{L} P(1, \Upsilon^{u}, \Delta_{l}^{d}) (1 - P^{u}) r_{l}^{d}$$

$$+ P(1, \Upsilon^{u}, \Upsilon^{d}) (1 - P^{u}) (1 - P^{d})$$

$$P(2, \Upsilon^{u}, \Delta_{l}^{d}) = \sum_{j=1}^{J} P(1, \Delta_{j}^{u}, \Delta_{l}^{d}) r_{j}^{u} (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d})$$

$$+ \sum_{j=1}^{J} \sum_{g=1}^{L} P(1, \Delta_{j}^{u}, \Delta_{g}^{d}) r_{j}^{u} z_{g,l}^{d}$$

$$+ \sum_{j=1}^{J} P(1, \Delta_{j}^{u}, \Upsilon^{d}) r_{j}^{u} p_{l}^{d}$$

$$+ P(1, \Upsilon^{u}, \Delta_{l}^{d}) (1 - P^{u}) (1 - r_{l}^{d} - \widetilde{Z}_{l}^{d})$$

$$+ \sum_{g=1}^{L} P(1, \Upsilon^{u}, \Delta_{g}^{d}) (1 - P^{u}) z_{g,l}^{d}$$

$$+ P(1, \Upsilon^{u}, \Upsilon^{d}) (1 - P^{u}) p_{l}^{d}$$
(A.2.7)

A.3 Lower Boundary States

The transition equations for the lower boundary states are the followings:

$$P(N-2, \Delta_{j}^{u}, \Upsilon_{l}^{d}) = \sum_{l=1}^{L} P(N-1, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{d})r_{l}^{d}$$
(A.3.1)
+
$$\sum_{f=1}^{J} \sum_{l=1}^{L} P(N-1, \Delta_{f}^{u}, \Delta_{l}^{d})z_{f,j}^{u}r_{l}^{d}$$
+
$$P(N-1, \Delta_{j}^{u}, \Upsilon^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - P^{d})$$
+
$$\sum_{f=1}^{J} P(N-1, \Delta_{f}^{u}, \Upsilon^{d})z_{f,j}^{u}(1 - P^{d})$$
+
$$\sum_{l=1}^{L} P(N-1, \Upsilon^{u}\Delta_{l}^{d})p_{j}^{u}r_{l}^{d}$$
+
$$P(N-1, \Upsilon^{u}, \Upsilon^{d})p_{j}^{u}(1 - P^{d})$$

$$\begin{split} P(N-1,\Delta_{j}^{u},\Delta_{l}^{d}) &= P(N-1,\Delta_{j}^{u},\Delta_{l}^{d})(1-r_{j}^{u}-\widetilde{Z}_{j}^{u})(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) & (A.3.2) \\ &+ \sum_{f=1}^{J} P(N-1,\Delta_{f}^{u},\Delta_{l}^{d})z_{f,j}^{u}(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(N-1,\Delta_{j}^{u},\Delta_{g}^{d})(1-r_{j}^{u}-\widetilde{Z}_{j}^{u})z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} \sum_{g=1}^{L} P(N-1,\Delta_{f}^{u},\Delta_{g}^{d})z_{f,j}^{u}z_{g,l}^{d} \\ &+ P(N-1,\Delta_{j}^{u},\Upsilon^{d})(1-r_{j}^{u}-\widetilde{Z}_{j}^{u})p_{l}^{d} \\ &+ \sum_{f=1}^{J} P(N-1,\Delta_{f}^{u},\Upsilon^{d})z_{f,j}^{u}p_{l}^{d} \\ &+ P(N-1,\Upsilon^{u},\Delta_{l}^{d})p_{j}^{u}(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(N-1,\Upsilon^{u},\Delta_{g}^{d})p_{j}^{u}z_{g,l}^{d} \\ &+ P(N-1,\Upsilon^{u},\Upsilon^{d})p_{j}^{u}p_{l}^{d} \end{split}$$

$$P(N-1,\Delta_{j}^{u},\Upsilon^{d}) = \sum_{l=l}^{L} P(N,\Delta_{j}^{u},\Delta_{l}^{d})(1-r_{j}^{u}-\widetilde{Z}_{j}^{u})r_{l}^{d}$$
(A.3.3)
$$=\sum_{f=l}^{J} P(N,\Delta_{f}^{u},\Delta_{l}^{d})z_{f,j}^{u}r_{l}^{d}$$
$$+\sum_{l=l}^{L} P(N,\Upsilon^{u},\Delta_{l}^{d})q_{j}^{u}r_{l}^{d}$$

$$P(N-1,\Upsilon^{u},\Delta_{l}^{d}) = \sum_{j=1}^{J} P(N-2,\Delta_{j}^{u},\Delta_{l}^{d})r_{j}^{u}(1-r_{l}^{d}-\widetilde{Z}_{l}^{d})$$
(A.3.4)
+
$$\sum_{j=1}^{J}\sum_{g=1}^{L} P(N-2,\Delta_{j}^{u},\Delta_{g}^{d})r_{j}^{u}z_{g,l}^{d}$$

+
$$\sum_{j=1}^{J} P(N-2,\Lambda_{j}^{u},\Upsilon^{d})r_{j}^{u}p_{l}^{d}$$

+
$$P(N-2,\Upsilon^{u},\Delta_{l}^{d})(1-P^{u})(1-r_{l}^{d}-\widetilde{Z}_{l}^{d})$$

+
$$\sum_{g=1}^{L} P(N-2,\Upsilon^{u},\Delta_{g}^{d})(1-P^{u})z_{g,l}^{d}$$

+
$$P(N-2,\Upsilon^{u},\Upsilon^{d})(1-P^{u})p_{l}^{d}$$

$$P(N-1,\Upsilon^{u},\Upsilon^{d}) = \sum_{j=1}^{J} \sum_{l=1}^{L} P(N-1,\Delta_{j}^{u},\Delta_{l}^{d})r_{j}^{u}r_{l}^{d}$$
(A.3.5)
+
$$\sum_{j=1}^{J} \sum_{l=1}^{L} P(N,\Delta_{j}^{u},\Delta_{l}^{d})r_{j}^{u}r_{l}^{d}$$
+
$$\sum_{j=1}^{J} P(N-1,\Delta_{j}^{u},\Upsilon^{d})r_{j}^{u}(1-P^{d})$$
+
$$\sum_{l=1}^{L} P(N-1,\Upsilon^{u},\Delta_{l}^{d})(1-P^{u})r_{l}^{d}$$
+
$$\sum_{l=1}^{L} P(N,\Upsilon^{u},\Delta_{l}^{d})(1-Q^{u})r_{l}^{d}$$
+
$$P(N-1,\Upsilon^{u},\Upsilon^{d})(1-P^{u})(1-P^{d})$$

$$\begin{split} P(N, \Delta_{j}^{u}, \Delta_{l}^{d}) &+ P(N, \Delta_{j}^{u}, \Delta_{l}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u})(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{f=1}^{J} P(N, \Delta_{f}^{u}, \Delta_{l}^{d}) z_{f,j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(N, \Delta_{j}^{u}, \Delta_{g}^{d})(1 - r_{j}^{u} - \widetilde{Z}_{j}^{u}) z_{g,l}^{d} \\ &+ \sum_{f=1}^{J} \sum_{g=1}^{L} P(N, \Delta_{f}^{u}, \Delta_{g}^{d}) z_{f,j}^{u} z_{g,l}^{d} \\ &+ P(N, \Upsilon^{u}, \Delta_{l}^{d}) q_{j}^{u}(1 - r_{l}^{d} - \widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(N, \Upsilon^{u}, \Delta_{g}^{d}) q_{j}^{u} z_{g,l}^{d} \end{split}$$

$$\begin{split} P(N,\Upsilon^{u},\Delta_{l}^{d}) &= \sum_{j=1}^{J} P(N-1,\Delta_{j}^{u},\Delta_{l}^{d})r_{j}^{u}(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) \\ &+ \sum_{j=1}^{J} P(N,\Delta_{j}^{u},\Delta_{l}^{d})r_{j}^{u}(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) \\ &+ \sum_{j=1}^{J} \sum_{g=1}^{L} P(N-1,\Delta_{j}^{u},\Delta_{g}^{d})r_{j}^{u}z_{g,l}^{d} \\ &+ \sum_{j=1}^{J} \sum_{g=1}^{L} P(N,\Delta_{j}^{u},\Delta_{g}^{d})r_{j}^{u}z_{g,l}^{d} \\ &+ \sum_{j=1}^{J} P(N-1,\Delta_{j}^{u},\Upsilon^{d})r_{j}^{u}p_{l}^{d} \\ &+ P(N-1,\Upsilon^{u},\Delta_{l}^{d})(1-P^{u})(1-r_{l}^{d}-\widetilde{Z}_{l}^{d}) \\ &+ \sum_{g=1}^{L} P(N,\Upsilon^{u},\Delta_{g}^{d})(1-Q^{u})z_{g,l}^{d} \\ &+ P(N-1,\Upsilon^{u},\Lambda_{g}^{d})(1-Q^{u})z_{g,l}^{d} \\ &+ P(N-1,\Upsilon^{u},\Lambda_{g}^{d})(1-Q^{u})p_{l}^{d} \end{split}$$

Bibliography

- Mostafa H. Ammar and Stanley B. Gershwin. Equivalence relations in queuing models of fork/join networks. *Performance Evaluation Journal*, 10:233–245, 1989. Special Issue on Queueing Networks with Finite Capacities.
- [2] Donald W. Berry and Bernard W. Lindgren. Statistics, Theory and Methods. Brooks/Cole Publishing Company, Pacific Grove, CA, 1990.
- [3] M. H. Burman. New Results in Flow Line Analysis. PhD thesis, Massachusetts Institute of Technology, Operations Research Center, June 1995.
- [4] J. A. Buzacott and L. E. Hanifin. Models of automatic transfer lines with inventory banks — a review and comparison. *AIIE Transactions*, 10(2):197–207, 1978.
- [5] John A. Buzacott and J. George Shanthikumar. Stochastic Models of Manufacturing Systems. Prentice Hall, Englewood Cliffs, NJ, 1993.
- [6] Yves Dallery, R. David, and X.-L. Xie. An efficient algorithm for analysis of transfer lines with unreliable machines and finite buffers. *IIE Transactions*, 20(3):280–283, 1988.
- [7] S. B. Gershwin. An efficient decomposition method for the approximate evaluation of tandem queues with finite storage space and blocking. *Operations Research*, 35(2):291–305, 1987.

- [8] S. B. Gershwin. Manufacturing Systems Engineering. Prentice-Hall, Englewood Cliffs, NJ, 1994. For corrections, see http://web.mit.edu/manuf-sys/www/gershwin.errata.html.
- [9] Averill M. Law and W. David Kelton. Simulation Modeling and Analysis. McGraw-Hill, New York, NY, second edition, 1991.
- [10] N. Maggio, A. Matta, S. B. Gershwin, and T. Tolio. An approximate analytical method for evaluating the performance of closed loop flow systems with unreliable machines and finite buffers — Part I: Small loops. Unpublished manuscript, 2003.
- [11] Micheline Maynard. Ford moving toward to flexible manufacturing. New York Times, April 14, 2006.
- [12] J. E. Nemec. Diffusion and Decomposition Approximations of Stochastic Models of Multiclass Processing Networks. Mit operations research center ph. d. thesis, Massachusetts Institute of Technology, February 1999.
- [13] T. Tolio and Stanley B. Gershwin. Deterministic two-machine lines with multiple failure modes. Technical report, Politicnico di Milano — Dipartimento di Meccanica, 1996.